

Peatland initiation through time and space Cindy Quik

### Propositions

- Incomplete descriptions of sample taphonomy are the main limitation for reconstructing peat initiation. (this thesis)
- Current peat remnants offer sufficient options to validate peatland reconstructions. (this thesis)
- 3. It is outdated that FAIR datasets are not credited for the PhD degree.
- 4. Open review is the best peer-review strategy to enhance scientific quality.
- 5. It is inconceivable that bags of garden compost are certified organic when they contain peat-based products.
- 6. Compulsory use of carbon credits in research projects is needed to reduce environmental impact of international travel for scientific conferences.

Propositions belonging to the PhD thesis, entitled

Peatland initiation through time and space

Cindy Quik Wageningen, 5 April 2023

# Peatland initiation through time and space

Cindy Quik

#### Thesis committee

#### Promotor

Prof. Dr Jakob Wallinga Professor of Soil Geography and Landscape Wageningen University & Research

#### **Co-promotors**

Dr Roy van Beek Associate professor, Soil Geography and Landscape and Cultural Geography Wageningen University & Research

Dr Ype van der Velde Associate professor, Faculty of Science, Earth and Climate Vrije Universiteit Amsterdam

#### Other members

Prof. Dr Gerlinde de Deyn, Wageningen University & Research Dr Minna Väliranta, University of Helsinki, Finland Prof. Dr Michael Dee, University of Groningen Dr Kim Cohen, University of Utrecht

This research was conducted under the auspices of the C.T. de Wit Graduate School for Production Ecology and Resource Conservation (PE&RC).

## Peatland initiation through time and space

Cindy Quik

Thesis

submitted in fulfilment of the requirements for the degree of doctor at Wageningen University by the authority of the Rector Magnificus Prof. Dr A.P.J. Mol, in the presence of the Thesis Committee appointed by the Academic Board to be defended in public on Wednesday 5 April 2023 at 1:30 p.m. in the Omnia Auditorium.

Cindy Quik Peatland initiation through time and space, 197 pages.

PhD thesis, Wageningen University, Wageningen, the Netherlands (2023) With references, with summary in English

ISBN 978-94-6447-520-3 DOI 10.18174/582788

## **Table of Contents**

1	Gen	General introduction 1					
	1.1	Wetla	nds and peatlands				
	1.2	From	dryland to wetland to peatland				
		1.2.1	Å transition in time and space				
		1.2.2	Reconstructing peat initiation and lateral expansion 3				
		1.2.3	The case of peat remnants				
	1.3	Proble	m statement and aim				
	1.4	Appro	ach and thesis outline				
	1.5	Study	area				
	1.6	Forma	I embedding of this PhD thesis				
2	Usir	ig lega	cy data to reconstruct the past?				
	Res	cue, rig	our and reuse in peatland geochronology 15				
	2.1	Introd	uction $\ldots$ $\ldots$ $\ldots$ $\ldots$ $17$				
	2.2	Backg	round				
		2.2.1	Legacy data in geoscience				
		2.2.2	Processes of peat formation				
		2.2.3	Legacy data in peatland research				
		2.2.4	Radiocarbon dating				
		2.2.5	Case study selection and aims				
		2.2.6	Case study area				
	2.3	3 Approach and methods					
		2.3.1	Workflow, database set-up and quality assessment 26				
		2.3.2	Application of the workflow to a case study				
	2.4	Result	s				
		2.4.1	Data rescue for case study region				
		2.4.2	Quality assessment				
		2.4.3	Meta-analysis				
	2.5	Discus	sion				
		2.5.1	Case study				
		2.5.2	Evaluation of approach				
		2.5.3	Implications and recommendations				
	2.6	Conclu	ısion				

3	Dat	Dating basal peat:				
	The	e geochronology of peat initiation revisited	57			
	3.1	Introduction	59			
	3.2	Methods	65			
		3.2.1 Selection of study area and overview of methods	65			
		3.2.2 Study Area	66			
		3.2.3 Site selection and stratigraphy	66			
		3.2.4 Collection of cores	71			
		3.2.5 Core processing and subsampling	72			
		3.2.6 (Bio)stratigraphical analyses	75			
		3.2.7 Defining basal peat (determining $\mathbf{M}_{d}$ value)	75			
		3.2.8 Radiocarbon dating	75			
		3.2.9 Calibration and age-depth modelling	77			
	3.3	Results	80			
		3.3.1 The organic matter gradient of the mineral-to-peat transi-				
		tion defining basal peat $(M_d)$	80			
		3.3.2 Plant macrofossils (PM) and testate amoehae (TA)	80			
		3.3.3 Dating results	83			
	34	Discussion	91			
	0.1	3.4.1 Mineral-to-neat transition and basal neat (defining $M_{J}$ )	91			
		3.4.2 Poloeoenvironment	01			
		3.4.3 Time span of the peat initiation process	02			
		3.4.4 Are assemblare of carbon fractions	05			
		3.4.5 Recommendations for dating next initiation	08			
	3 5	Conclusions	102			
	5.5	2.5.1 Dating post initiation	102			
		2.5.2 Delegeonvironment	102			
			102			
Δ	Fad	ed landscape:				
-	Unr	avelling neat initiation and lateral expansion				
	at o	one of NW-Europe's largest hog remnants	105			
	4 1	Introduction	107			
	4.2	Study area	100			
	7.2	A 2.1 Selection and description of study area	100			
		4.2.2 Peatland development and decline in the (wider) study area	112			
	13	Methods	116			
	4.5	A 2.1 Methods Dart 1: Collection of field data and radiocarbon	110			
		4.5.1 Methods Falt 1. Collection of held data and fadiocarbon	116			
		4.2.2. Methode Devt 2: Deconstructing post initiation are contially	110			
	4 4	4.3.2 Methods Part 2: Reconstructing peat initiation age spatially.	119			
	4.4	Results	124			
		4.4.1 Results Part 1: Collection of field data and radiocarbon	104			
		dating evidence	124			
		4.4.2 Results Part 2: Reconstructing peat initiation age spatially	124			
	4.5	Discussion	133			
		4.5.1 Peat initiation and lateral expansion at Fochteloerveen	133			
		4.5.2 Evaluation of approach	135			
	4.6	Conclusion	138			

5	Gene	eral syn	thesis	141		
	5.1	Introdu	lction	142		
	5.2	Method	dological advances	142		
		5.2.1	Reuse of existing radiocarbon dates in peatland			
			geochronology	142		
		5.2.2	Defining and dating basal peat	143		
		5.2.3	Reconstructions based on peat remnants	144		
		5.2.4	Relevance and integration of methodological tools	145		
	5.3	es in understanding initiation and lateral development of				
		peatlands in the Northwest European mainland				
		5.3.1	Palaeoenvironment in the Fochteloërveen area during peat			
			initiation	147		
		5.3.2	Peat initiation and lateral expansion at Fochteloërveen and			
			the northern Dutch coversand landscape	149		
		5.3.3	Process of peat initiation and steering factors for peat growth	150 ו		
	5.4	Conclus	sions	157		
	5.5	Recom	mendations for future research	159		
		5.5.1	Recommendations for methodological developments	159		
		5.5.2	Recommendations for peatland research in the Northwest			
			European mainland	161		
	5.6	Implica	tions	163		
		5.6.1	Peatlands and ecosystem services	163		
		5.6.2	Nature conservation and restoration	164		
		5.6.3	Cultural history and heritage management	165		
		5.6.4	Peatlands in environmental education and science commu-			
			nication	166		
Ro	foron	CO5		160		
Ne	leren	CES		105		
Su	mmai	r <b>y</b>		184		
Tra	Training and education statement					
Lis	List of publications					
Ac	Acknowledgements					
Fir	Financial support					

# Chapter 1

General introduction

## 1.1 Wetlands and peatlands

Wetlands are a distinctive landscape type, and can be characterized as areas where the mineral substrate is water-saturated or inundated. Consequently the substrate is poorly aerated, and the period that the water table is near or above the surface is of such duration that the prevalent vegetation and other organisms are adapted to life in water-saturated soil conditions (Charman, 2002c: 4; Joosten and Clarke, 2002: 24; Rydin and Jeglum, 2013c: 2-3). A minimum depth of 30 cm of peat is required to classify a wetland area as a peatland (Charman, 2002c: 4; Joosten and Clarke, 2002: 24; Rydin and Jeglum, 2013c: 4). Peat is organic material that accumulated as a result of a positive production-decay balance. It consists of the (partially) undecomposed remains of plant and animal tissues, which accumulated under more or less water-saturated conditions (Rydin and Jeglum, 2013c: 4). Peatlands comprise 50 - 70% of all global wetlands, making them the most prevalent wetland type (Joosten and Clarke, 2002: 6). As peatland terminology is not fully refined and standardized (Joosten and Clarke, 2002: 9), and several terms are used interchangeably (Rydin and Jeglum, 2013c: 13). I have included a glossary with definitions as used in this thesis (Textbox 1.1).

Peatlands exist in a diversity of conditions. As a result, they are present on all continents, covering a wide latitudinal range and occurring at altitudes varying from sea level to alpine zones (Joosten and Clarke, 2002: 6; United Nations Environment Programme, 2021). Since the 1800s (and regionally earlier), the extent of peatlands has decreased, with an estimated reduction in their global surface area of 10 to 20% (Joosten and Clarke, 2002: 7). For Europe, the loss in peatland area is much greater. Joosten and Clarke (2002: 32) estimated that over 50% of the European peatland area was lost in the period between 1952 and 1992. For the Netherlands, only 1% of the former peatland area is left today (Joosten and Couwenberg, 2001: 407).

Peatlands form natural archives of past environmental changes. Through the preservation of in-situ plant remains, pollen, spores, and animals, and capture of air- and waterborne pollen, spores and other particles, peat deposits contain a record of changes over time that provides information not only about peatland development itself but also about the paleoenvironmental conditions in which these developments took place (Rydin and Jeglum, 2013b: 107). As such, the remains embedded in peat deposits are called the peat archives (Godwin, 1981). In addition, peat deposits contain cultural heritage with a degree of preservation that is unparalleled in dryland environments, including bog bodies, wooden trackways, and a range of other finds (e.g. Glob, 1969; Casparie, 1987; Coles and Lawson, 1987; Van der Sanden, 1996).

The information contained in the peat archives is of unprecedented relevance. Key areas of research include (1) understanding of peatland functioning and development (Tuittila *et al.*, 2013; Hilbert *et al.*, 2017), (2) interpretations of allogenic drivers of change such as climate and sea level rise (Morris *et al.*, 2018), (3) consequences for ecosystem services provided by peatlands (Parish *et al.*, 2008b; Kimmel and Mander, 2010; Rotherham, 2020), such as carbon dynamics (e.g. Yu *et al.*, 2011; Ratcliffe *et al.*, 2021), and (4) to contextualize long-term humanlandscape interactions (e.g. Gearey and Everett, 2021). The tremendous losses in peatland surface area as discussed above, related losses in peat thickness, and ongoing threats to their existence in Europe and elsewhere (e.g. Bragazza *et al.*, 2006; Swindles *et al.*, 2019) pose the urgency to accumulate the knowledge that is contained in the peat archives, and the need to protect the areas that are left. In addition, understanding the past is principal for navigating the future, for which dynamic scenarios need to be envisioned and management must be guided towards resilience and adaptivity to change (Gillson *et al.*, 2021).

The peat archives are generally studied intensively to answer questions stemming from various research fields. However, one of the biggest changes recorded in the peat archives, the transition to peat growth itself, is underexposed in scientific research. Recent research indicated that relatively dry forests (with high biomass, high water use, and a thin organic soil) and peat-forming wetlands (with few and/or small trees, relatively low biomass, low water use, and a thick organic soil) may form alternative stable states. Each of these stable states holds a degree of resilience to changes in climate or hydrology, but when thresholds are crossed a persistent shift between these states may occur (Van der Velde *et al.*, 2021). The transition from dryland to wetland (and to peatland) represents a huge landscape change, with major impacts for landscape functioning, ecosystem services and human-landscape interactions.

## 1.2 From dryland to wetland to peatland

#### 1.2.1 A transition in time and space

During the transition from dryland to wetland and to peatland, peat initiation may occur through (a combination of) three processes, which are briefly outlined in Textbox 1.2. In the course of time, the peat layer accumulates vertically and the peatland may reach a point where the surface grows above the groundwater level. Subsequent isolation from groundwater and full dependence on precipitation leads to ombrotrophication, a process known as the fen-bog transition (Charman, 2002a; Rydin and Jeglum, 2013d). In addition to vertical accumulation of peat, the peatland may expand laterally. Poor drainage adjacent to the peatland may lead to paludification of surrounding soils. This is an autogenic process (Charman, 2002a), but the extent and rate of lateral expansion are influenced by allogenic factors such as climate and topography of the mineral soil (Korhola, 1994; Loisel *et al.*, 2013).

#### 1.2.2 Reconstructing peat initiation and lateral expansion

Reconstructing the period, pace and pattern of peat initiation and lateral expansion requires dating the bottom of peat deposits overlying mineral sediment, often called the *basal peat*. Because of the organic composition of peat, radiocarbon (<sup>14</sup>C) dating is the best suited method to connect peat stratigraphies to an absolute time scale. Radiocarbon dating is based on the radioactive decay of <sup>14</sup>C. This isotope is produced in the upper atmosphere as a result of cosmic radiation. It is subsequently oxidized to <sup>14</sup>CO<sub>2</sub>, which becomes incorporated in the tissues

Definitions	
Wetland <sup>1</sup>	<ul> <li>An area where:</li> <li>The water table is close to or above the surface;</li> <li>As a result, the substrate is poorly aerated;</li> <li>The period that the water table is near or above the surface is of such duration that the prevalent vegetation and other organisms are adapted for life in water-saturated soil conditions.</li> </ul>
Peat <sup>2</sup>	A substance that consists of the (partially) undecomposed remains of plant and animal tissues, which accumulated under more or less water-saturated conditions. In this thesis material with an organic matter percentage of 40 or more is considered as peat.
Peatland <sup>3</sup>	An area where 30 cm of peat or more naturally accumulated.
Mire <sup>4</sup>	A wetland dominated by active peat-forming vegetation.
Fen or minerotrophic peatland <sup>5</sup>	Mire or peatland that in its present state receives water that has been in contact with mineral soil (groundwater, surface runoff). As a result, fens are often characterized by more alkaline conditions and higher nutrient availability (in comparison to bogs).
Bog or ombrotrophic peatland <sup>5</sup>	Mire or peatland that in its present state receives water and nutrients solely from precipitation. As a result, bogs are often characterized by more acidic conditions and low nutrient availability (in comparison to fens).
Minerogenous or geogenous versus ombrogenous <sup>5</sup>	Referring to the trophic status (as in minerotrophic and ombrotrophic) during peat initiation (i.e., not the present state).

#### Textbox 1.1: Definitions and classification as used in this thesis

#### Classification

Six basic hydromorphological mire types  $^{6}$  can be distinguished based on the shape of the peatland and its mineral substrate, and the hydrological situation of the peatland:

- Raised mire (bog)
- Blanket mire (bog)
- Basin mire (fen)
- Valley mire (fen)
- Floodplain mire (fen)
- Sloping mire (fen)

<sup>1</sup> Charman, 2002c: 4; Joosten and Clarke, 2002: 24; Rydin and Jeglum, 2013c: 2-3

- <sup>2</sup> Rydin and Jeglum, 2013c: 4
- <sup>3</sup> Charman, 2002c: 4; Joosten and Clarke, 2002: 24; Rydin and Jeglum, 2013c: 4
- <sup>4</sup> Joosten and Clarke, 2002: 24; Rydin and Jeglum, 2013c: 4–5
- <sup>5</sup> Charman, 2002c: 6
- <sup>6</sup> Charman, 2002c: 7

Textbox 1.2:	Processes of peat initiation
--------------	------------------------------

Terrestrialisation <sup>1</sup>	Terrestrialisation (also called infilling) refers to the process where peat develops in or at the edge of water bodies. Terrestrialisation is characterized by gyttja deposits at the base, which require a water depth of at least 0.5 m to form <sup>2</sup> .
Paludification <sup>1</sup>	Paludification refers to peat formation on previously unsaturated min- eral substrate, and thus reflects drowning of the landscape.
Primary mire formation <sup>1</sup>	Primary mire formation involves peat growth on newly exposed water- logged substrate (e.g. after land uplift from sea). Here, peat growth starts directly on the fresh parent material.
<sup>1</sup> Charman, 2002a: Rydi	n and Jeglum, 2013d

<sup>2</sup> Bos, 2010

of living organisms through photosynthesis and uptake via the food chain. Upon death of the organism, the radioactive decay of <sup>14</sup>C enables to derive its age (i.e., timing of death) (Bayliss *et al.*, 2004; Walker, 2005; Ramsey, 2008b).

The methods that are used to study peat initiation and lateral expansion (mostly applied in boreal settings), can be roughly classified in three approaches. The first group includes studies where transects of basal peat dates are obtained and lateral expansion rates are inferred based on the distance between dating points (e.g. Almquist-Jacobson and Foster, 1995; Turunen et al., 2002b; Anderson et al., 2003; Turunen and Turunen, 2003; Peregon et al., 2009; Robichaud and Bégin, 2009; Weckström et al., 2010; Loisel et al., 2013; Zhao et al., 2014). In this group, the spatio-temporal pattern of lateral development is not visualised. This is in contrast to the second group of studies, where the spatial distribution of the obtained basal dates is manually converted to lines of equal age, so-called isochrones. These isochrones visualise the pattern of lateral development, and the rate of lateral expansion can be deduced from the distance between isochrones (Foster et al., 1988; Korhola, 1994, 1996; Mäkilä, 1997; Bauer et al., 2003; Mäkilä and Moisanen, 2007; Edvardsson et al., 2014). A third group includes studies that make use of numerical modelling based on hydrological and ecohydrological feedbacks. In these models, vertical peat growth (age-depth) for a peat column is simulated (e.g. the Holocene Peat Model, Frolking et al., 2010; the DigiBog model, Baird et al., 2012 and Morris et al., 2012; the coupled DigiBog-STREAM model, Swinnen et al., 2021). Unfortunately, numerical models that include lateral expansion of peatlands are so far unavailable (see e.g. the discussion on peat models by Baird et al., 2012).

To obtain dating information for reconstructions, one can either look back in the scientific record to build on and integrate existing information, or obtain new data from the field. Data that have been collected in the past are often referred to as 'legacy' data (Griffin, 2015; Smith *et al.*, 2015). Data rescue is an area of increasing influence in geosciences as many researchers realise both the scientific potential of reusing legacy data, and the growing threat of data loss (Wyborn

*et al.*, 2015). Data loss is especially alarming in regions and landscapes where natural archives are degrading or at risk, such as peatlands.

When legacy data are synthesized through meta-analyses, new insights may develop that require a bird's-eye view to be discovered (see for example Tolonen and Turunen, 1996; Ruppel *et al.*, 2013. Meta-analyses cross the boundaries of time and place that form the limits of case studies, and are particularly useful when information can no longer be obtained from the field, or when long-term records are needed to investigate how phenomena change through time. However, this requires adequate data access and retrieval, transformation of data to current digital formats, and ways to evaluate data quality and effects of changing research methods to ensure robust meta-analyses. However, no overviews exist of factors that need to be taken into consideration for reuse of geochronological information in peat research, and standardized workflows or designs for quality assessments of peat dates are lacking.

When obtaining new information from the field for reconstructing peat initiation and lateral expansion, accurately dating *basal peat* is key. However, so far a variety of criteria has been used to define *basal peat*, which subsequently led to diverse approaches to date peat initiation. The accuracy of dates in representing the event of interest, i.e. the timing of peat initiation, may be questioned if the (possibly site-specific) definition of the *basal peat* remains implicit. To the best of my knowledge, both the mineral-to-peat transition (i.e., the stratigraphical range reflecting the timespan of the peat initiation process) and the layer called *basal peat* (i.e., the stratigraphical layer that is defined as the bottom of a peat deposit), are not universally defined.

Appropriate sampling and sample pre-treatment for radiocarbon dating are the following principal steps to obtain an accurate age of the basal peat layer. During the past decades multiple concerns were raised (and partly addressed) with the radiocarbon dating of peat (e.g. Törnqvist et al., 1992, 1998; Shore et al., 1995; Nilsson et al., 2001; Brock et al., 2011; Van der Plicht et al., 2019). To overcome the majority of concerns with radiocarbon dating of peat samples, AMS dating of terrestrial plant macrofossils is generally advised (e.g. Piotrowska et al., 2011). Unfortunately, this is not always possible for mineral-to-peat transitions, where the organic material may be highly amorphous and plant macrofossil content is generally low. If plant macrofossils cannot be obtained, one has little recourse but to use bulk dating. This can be problematic as bulk samples consist of a mixture of organic fractions and potentially a mixture of ages, which may yield apparent ages that are either too old or too young for the peat layer of interest (e.g. Törnqvist et al., 1992). Through chemical pre-treatment, a bulk sample can be split up in fulvic, humic and humin fractions based on their solubility (Brock et al., 2011; Van der Plicht et al., 2019). Due to its mobility, the fulvic fraction is generally not used for dating as it may result in significantly younger ages than the humic or humin fraction from the same level (Shore *et al.*, 1995). There is no clear consensus in literature on whether humic or humin dates are most representative for dating peat layers (see e.g. the contrasting findings by Hammond et al., 1991; Cook et al., 1998; Waller et al., 2006; Brock et al., 2011; Van der Plicht et al., 2019). In addition, virtually no studies focus on basal peat layers while investigating the ages of these organic fractions (with the exception of Brock *et al.*, 2011). It therefore remains unclear which carbon fractions of basal peat layers (which might be slightly different in organic carbon composition and especially in carbon content due to admixture of the mineral substrate compared to peat samples from higher positions in peat profiles) are most representative for the time period of peat initiation.

### 1.2.3 The case of peat remnants

So far, the palaeogeographical development of the former extensive peat landscapes of the Northwest European mainland has received mixed scientific attention. For the Netherlands for example, current palaeogeographic reconstructions (Zagwijn, 1986; Westerhoff et al., 2003; Vos, 2015b; Vos et al., 2020) were created with a strong focus on the development of the delta (e.g. Berendsen and Stouthamer, 2000, 2001) and coastal area (e.g. Hijma, 2009; Cohen et al., 2014; Pierik et al., 2017). In contrast, the spatio-temporal development of non-coastal and non-alluvial peatlands in the coversand landscape (see Section 1.5) remains uncertain, probably as a result of their large-scale disappearance following reclamation activities in the past few centuries (Gerding, 1995) and consequent limited amount of data for these areas (Spek, 2004; Van Beek, 2009; Vos, 2015a). Consequently, steering factors for peat initiation and lateral development in these areas are not well understood. Whereas peat growth in the coastal area is strongly related to Holocene sea level rise (e.g. Törnqvist and Hijma, 2012), the influence of (the combination of) sea level rise, climate change, permeability of the mineral substrate, vegetation changes and human influence on peat growth in the coversand landscape is more debated (e.g. Casparie and Streefkerk, 1992; Van Geel et al., 1996; Everts et al., 2002; Jansen and Grootjans, 2019).

Due to the poor preservation of peat landscapes in the Northwest European mainland, alternative approaches to reconstruct peat initiation and lateral expansion are needed compared to regions with intact peat cover. The tremendous losses of peat landscapes in Europe and in the Netherlands in particular (see Section 1.1) call for reuse of existing data in studies on the formation, dynamics and palaeoenvironmental characteristics of these landscapes, as legacy data may contain information that can no longer be obtained from the field. Currently, existing information on the age of these former peat landscapes is not yet fully synthesized. The most complete overview is contained in the work of Vos (2015a) and Vos *et al.* (2020), but still lacks a significant part of the dating evidence on peatlands in the Dutch coversand landscape that has been collected since the start of the radiocarbon dating facility at the University of Groningen in the 1950s.

The large-scale peat losses in Northwest Europe hamper new field studies and pose the need for adapted field strategies. In areas where the peatland area is significantly reduced, the placing of transects of basal dates (see section 1.2.2) is questionable as the orientation of peat remnants within the former extensive peat landscape is often unknown. Moreover, basal peat layers might be damaged or removed as a result of peat-cutting (and locally ongoing excavation). Consequently, an adapted strategy is needed to collect field data from peat remnants. In addition, alternative methods of analysis are needed to reconstruct peat initiation and lateral expansion where uncertainty is quantified (i.e., in contrast to manual

derivation of isochrones from basal age data). This is especially relevant when reconstructing a former landscape of which large parts are lost.

## 1.3 Problem statement and aim

Based on section 1.2 I define two key research deficits, which can be subdivided in several elements:

- I. Methodological developments are needed to constrain the spatio-temporal development of peatlands more accurately.
  - No overviews exist of factors that need to be taken into consideration for reuse of radiocarbon dates in geochronological peat research, and standardized workflows or designs for quality assessments of peat dates are lacking;
  - b. The lack of a universally applicable and quantitative definition for basal peat, combined with multiple concerns that have been raised previously regarding the radiocarbon dating of peat, may result in apparent ages that are either too old or too young for the timing of peat initiation;
  - c. Adapted strategies are needed to obtain field data and to model peat initiation and lateral expansion based on peat remnants.
- II. Understanding of the timing, pace and pattern of the initiation and lateral development of peatlands in the coversand landscape of the Northwest European mainland is limited, and responsible steering factors are not well understood.
  - a. Existing knowledge on the age of these peatlands is not synthesized in full;
  - b. The development of these peatlands is not well constrained in time and space;
  - c. The identification and relative influence of steering factors for peat growth in these areas is debated.

In this thesis I aim to reconstruct peat initiation and lateral expansion in the coversand landscape of the Northwest European mainland, and to develop the required methodological tools, which can be applied irrespective of the case study region.

To reach the thesis aim and contribute to resolving the above-mentioned research deficits, I defined six objectives. These objectives are addressed in Chapters 2 - 4, where each chapter focuses on elements of research deficits (I) and (II):

- Chapter 2 (I) To develop a workflow for reuse of legacy radiocarbon data in peatland studies, including rigorous quality assessment, and to propose ways for tailoring the workflow to specific research questions and case studies.
- Chapter 2 (II) To test and evaluate the proposed approach by applying it to the (former) peatlands in the coversand landscape of the northern Netherlands (see section 1.5).
- Chapter 3 (I) To formulate recommendations for dating basal peat. Issues that are specifically addressed include:
  - Peat initiation is a process of a certain timespan rather than an event;
  - Basal peat needs to be clearly defined;
  - Selection of dating samples is typically challenging due to potential poor preservation of plant macrofossils in basal peat;
  - The representativity of humic and humin dates for the age of basal peat is questionable.
- Chapter 3 (II) To obtain initial dating evidence for the period of peat development at one of the largest bog remnants of the Northwest European mainland, the Fochteloërveen (see section 1.5).
- Chapter 4 (I) To find explanatory variables within a digital soil mapping approach that enable reconstruction of the pattern of peat initiation and lateral expansion within (and potentially beyond) peat remnants.
- Chapter 4 (II) To reconstruct peat initiation ages and the pattern of lateral expansion for the Fochteloërveen peat remnant.

## **1.4** Approach and thesis outline

A schematic overview of the thesis is shown in fig. 1.1. Each of the six objectives (section 1.3) are aligned to one of the elements of the two identified research deficits. Chapters 2 - 4 each address two of these objectives.

In investigations of peatlands as a whole, where they are considered as landforms, peat landform units can be distinguished based on three scale levels: the microtope, mesotope and macrotope (lvanov, 1981; Charman, 2002d). The microtope is characterized by a uniform vegetation and physical environment. The mesotope refers to individual mire massifs that originated from a single nucleus. The macrotope is formed through the coalescence of multiple mire mesotopes. Each chapter is connected to a certain peat landform scale. The period after the Last Glacial Maximum to the present (Late Pleniglacial and Holocene) forms the temporal scope of the thesis.

In Chapter 5, I integrate the methodological developments that are presented in this thesis, and synthesize the knowledge that was gained on the initiation and lateral development of peatlands in the coversand landscape of the Northwest European mainland. Based on the latter I elaborate on drivers for peat formation and formulate a hypothesis on potential steering factors. Subsequently I discuss directions for future research and implications of the thesis results.

Chapters 2 - 4 are published open access in peer-reviewed scientific journals. The three datasets that underlie these publications are published under a CC BY license in the 4TU.ResearchData repository, where they are freely available for download (references and DOI links are included in the respective thesis chapters).

## 1.5 Study area

In this thesis I focus on mires and peatlands that formed on non-coastal (located above mean sea level) and non-alluvial topographic plains in the coversand landscape of the temperate Northwest European mainland. These plateaus are intersected by deep river valleys and locally dotted with pingo remnants (De Gans, 1976). In these various geomorphological settings, several hydromorphological mire types (Textbox 1.1) occur, including raised mires (bogs), basin mires (fens), and valley mires (fens).

During the Weichselian (OIS 4-2), deposition of aeolian coversands over Northwest Europe led to the formation of the European Sand Belt (Koster, 1988, 2005). In the eastern half and southern part of the Netherlands, these coversands occur at the surface and are located above sea level (fig. 1.2). The northern part of the Dutch coversand area was selected as case study region for this thesis. The Fochteloërveen peat remnant is located in this region and protected as a Natura 2000 area (fig. 1.2 and fig. 1.3, Provincie Drenthe, 2016; Douwes and Straathof, 2019). The Fochteloërveen does not fit within a single definition as it probably formed through coalescence of multiple smaller mires that developed on a topographic plain (see Chapter 4). However, the resultant composite peatland can best be described as a plateau raised bog, i.e. a peatland that in its current state only receives water and nutrients from precipitation, and with a generally flat and horizontal surface (Moen, 1985; Charman, 2002c: 8). More information on the northern coversand landscape and Fochteloërveen peat remnant is provided in subsequent chapters.

## **1.6 Formal embedding of this PhD thesis**

This PhD-study is part of the NWO-Vidi project 'Home Turf. An integrated approach to the long-term development, cultural connections and heritage management of Dutch raised bogs', led by Dr Roy van Beek. The project has an

interdisciplinary design and consists of several interlinked project elements: landscape archaeology (by Roy van Beek), historical geography (by Maurice Paulissen), ore geology (by Aukjen Nauta), and palaeogeography (this thesis).

In addition to the objectives listed in section 1.3, I have contributed to interdisciplinary studies embedded in the *Home Turf* Project, with the aim to improve understanding of prehistoric and early historic human-landscape interactions in landscapes with plateau raised bogs and peatlands in general. Results from these research efforts are not included in this thesis but available in the following scientific papers:

- Van Beek R, Candel JHJ, Quik C, Bos JAA, Gouw-Bouman MTIJ, Makaske B, Maas GJ. 2019. The landscape setting of bog bodies: Interdisciplinary research into the site location of Yde Girl, The Netherlands. *The Holocene* 29(7): 1206–1222, https://doi.org/10.1177/0959683619838048.
- Van Beek R, Quik C, Bergerbrant S, Huisman F, Kama P. 2022. Bogs, bones and bodies. The deposition of human remains in European mires (9000 BC – AD 1900). Antiquity 1–21, https://doi.org/10.15184/aqy.2022.163.
- Van Beek R, Quik C, Van der Linden M. Drowning landscapes revisited. Correlating peatland expansion, human habitation trends and vegetation dynamics in the North European Plain. In preparation.



Figure 1.1: Schematic overview of the thesis.



**Figure 1.2:** Location of the case study region and case study peat remnant, indicated on the palaeogeographical map of the Netherlands showing the situation for 9,000 BCE (10,950 cal y BP) (Vos et al., 2020; RCE, 2022). Only map units related to the coversand landscape are indicated here. Elevation is in metres relative to Dutch Ordnance Datum (O.D. [in Dutch: NAP], roughly equal to mean sea level). Extent of the Fochteloërveen Natura 2000 area is indicated (Ministerie van Economische Zaken - Directie Natuur & Biodiversiteit, 2018).



**Figure 1.3:** Picture compilation of Fochteloërveen (upper, right and lower picture by Cindy Quik, 2019; left picture by Roy van Beek, 2020).

## Chapter 2

Using legacy data to reconstruct the past? Rescue, rigour and reuse in peatland geochronology

Cindy Quik, Ype van der Velde, Tom Harkema, Hans van der Plicht, Jim Quik, Roy van Beek & Jakob Wallinga (2021)

Earth Surface Processes and Landforms, 46(13), 2607–2631: doi.org/10.1002/esp.5196

> Dataset at 4TU. Centre for Research Data: doi.org/10.4121/14406347

## Abstract

There is a growing interest for rescue and reuse of data from past studies (so-called legacy data). Data loss is alarming, especially where natural archives are under threat, such as peat deposits. Here we develop a workflow for reuse of legacy radiocarbon dates in peatland studies, including a rigorous quality assessment that can be tailored to specific research questions and study regions. A penalty is assigned to each date based on criteria that consider taphonomic quality (i.e. sample provenance) and dating quality (i.e. sample material and method used). The weights of quality criteria may be adjusted based on the research focus, and resulting confidence levels may be used in further analyses to ensure robustness of conclusions. We apply the proposed approach to a case study of a (former) peat landscape in the Netherlands, aiming to reconstruct the timing of peat initiation spatially. Our search yielded 313 radiocarbon dates from the 1950s to 2019. Based on the quality assessment the dates -of highly diverse quality- were assigned to four confidence levels. Results indicate that peat initiation for the study area first peaked in the Late Glacial ( $\sim$ 14,000 cal y BP), dropped during the Boreal (~9,500 cal y BP) and showed a second peak in the Subboreal (~4,500 cal y BP). We tentatively conclude that the earliest peak was mostly driven by climate (Bølling-Allerød interstadial), whereas the second was probably the result of Holocene sea level rise and related groundwater level rise in combination with climatic conditions (hypsithermal). Our study highlights the potential of legacy data for palaeogeographic reconstructions, as it is cost-efficient and provides access to information no longer available in the field. However, data retrieval may be challenging and reuse of data requires that basic information on location, elevation, stratigraphy, sample and laboratory analysis are documented irrespective of the original research aims.

## 2.1 Introduction

Data rescue in the geosciences is a field of rapidly growing interest (Wyborn *et al.*, 2015). Data that has been collected in the past is often referred to as 'legacy' data (Griffin, 2015; Smith *et al.*, 2015). Many researchers are realizing both the scientific potential of reusing data from past studies, and the increasing threat of data loss, particularly concerning data from the pre-digital era. Data loss is alarming, particularly in landscapes where natural archives are degrading or at risk, such as peatlands. Peatlands are under ongoing threat of excavation, drainage, pollution and climate change (e.g. Bragazza *et al.*, 2006; Swindles *et al.*, 2019).

The long-term archives of past environments contained in peat deposits are in some regions largely lost, as is evident from the relatively minute remnants of the once extensive peatlands of northwest Europe (Casparie, 1972; Vos, 2015a). Consequently, studies on the formation, dynamics and palaeoenvironmental characteristics of these landscapes could greatly benefit from data rescue, as legacy data may contain information that can no longer be obtained in the field. Additionally, access to field sites may be difficult due to strict nature conservation regulations in protected peat remnants. Furthermore, limited understanding of how representative these remnants are for the former intact landscape makes field-based studies challenging. Hence data rescue potentially offers a starting point for peatland research, may provide new insights through meta-analyses (e.g. Tolonen and Turunen, 1996; Ruppel *et al.*, 2013), and identify knowledge deficits to address with future research. However, data reuse is often challenging due to changing research methods, limited information on data quality, and difficulties regarding data access and retrieval.

Here we aim (1) to develop a workflow for reuse of legacy radiocarbon data in peatland studies, including rigorous quality assessment, and propose ways to tailor the workflow to specific research questions and case studies; (2) to test and evaluate the proposed approach by applying it to a case study of a (former) peat landscape in the northern Netherlands, for which we build a comprehensive dataset of legacy radiocarbon dates.

## 2.2 Background

In this section we briefly discuss the use of legacy data in geoscience, introduce processes of peat formation, review use of (legacy) radiocarbon dates in peatland research, and provide a short introduction to radiocarbon dating. In the last paragraph the case study is introduced.

## 2.2.1 Legacy data in geoscience

In the geosciences, legacy data may play a role to analyse landforms or processes of the past or that change through time, and to reinvestigate previous work (cf. Smith *et al.*, 2015). The distinction between 'new' and 'legacy' data is somewhat artificial, and partly the result of many practical issues such as unknown data storage locations, lack of accessibility, physically degrading storage media or unreadable data formats, and unwritten information on records that disappeared from the scientific community when researchers retired or passed away (Griffin, 2015; Wyborn *et al.*, 2015). Other important factors causing the artificial separation of legacy data include the continuous change in research methods, the technological advances to refine and develop new equipment, and ever-increasing computational power.

Data that were passed on by previous generations of scientists may potentially be used for purposes that are diverting from the original research objective for which the data collection was designed (Wyborn *et al.*, 2015). Meta-analyses based on legacy data may yield insights that require a birds-eye view on the subject matter, crossing boundaries of time and place that limit many case studies. This particularly applies when information is no longer available in the field, or when long-term records are needed to describe and quantify how systems changed through time. However, this requires adequate data access and retrieval, transformation of data to current digital formats, and ways to evaluate data quality and effects of changing research methods to ensure robust meta-analyses. To quote Griffin (2015): "[...] it is up to our community to remove [...] the artificial barriers that presently prevent the access that research requires simultaneously to *all* of its data."

## 2.2.2 Processes of peat formation

Peatlands form distinctive ecosystems on the verge from land and water. Their initiation is primarily dependent on the decay rate of biomass (and resulting production-decay balance), which is predominantly influenced by moisture level (Charman, 2002e). Factors that may influence moisture status and consequently peat growth potential include climate (e.g. Weckström *et al.*, 2010), changes in hydrological base level (such as sea level rise, e.g. Berendsen *et al.*, 2007) or regional groundwater changes (e.g. Van Asselen *et al.*, 2017), landforms and surface topography (e.g. Almquist-Jacobson and Foster, 1995; Mäkilä, 1997; Loisel *et al.*, 2013), impermeable deposits or resistant layers in the soil profile (e.g. Breuning-Madsen *et al.*, 2018; Van der Meij *et al.*, 2018), and anthropogenic influence (e.g. Moore, 1975, 1993). Some of these factors such as climate may act at larger spatial scales, whereas others, for instance impermeable layers, could also have

more local effects.

Given favourable boundary conditions, peat initiation may occur through (a combination of) terrestrialisation (also known as infilling), paludification, and primary mire formation (Charman, 2002e; Rydin and Jeglum, 2013d). Terrestrialisation refers to the process where peat forms in or at the edge of existing water bodies. Paludification does not include a true aquatic phase, instead peat develops directly on previously dry mineral substrate, following changes in moisture status that led to waterlogging. Primary mire formation refers to peat formation on newly exposed land (as opposed to paludification, where previous vegetation was present) that has been waterlogged since initial exposure, for instance after deglaciation or land uplift from sea.

Over time, peatlands grow vertically and may reach a point where their surface rises above groundwater level. Isolation from groundwater and resulting strong dependence on rainwater leads to ombrotrophication (Charman, 2002e; Rydin and Jeglum, 2013d). These fen-bog transitions may occur at various timing (Väliranta *et al.*, 2017).

In addition to vertical growth, peatlands may expand laterally to cover larger areas. Poor drainage adjacent to the peatland may cause paludification of surrounding soils. This is referred to as an autogenic process (Charman, 2002e), but the degree to which this happens and the rate of lateral spread are dependent on allogenic factors such as climate and topography (e.g. Korhola, 1994).

Reconstructing the period of peat initiation requires dating the peat base (also referred to as basal peat, do note that this definition of basal peat is much broader than the 'Basisveen Bed' as known in Dutch stratigraphy (TNO – Geological Survey of the Netherlands, 2021b)). Peat initiation and subsequent lateral expansion are often not easily distinguished as both require basal peat dates for reconstruction. Lateral expansion can only be deduced from a series of basal dates (e.g. Mäkilä, 1997; Mäkilä and Moisanen, 2007; Chapman *et al.*, 2013), which in fact is also needed to determine which date indicates the age and location of peat initiation.

For more elaborate information on peatlands, peat accumulation and peatland ecology we refer to e.g. Frenzel (1983), Charman (2002e), Wieder and Vitt (2006), Mitsch *et al.* (2009) or Rydin and Jeglum (2013a). For spatial distribution of peatlands see e.g. Joosten *et al.* (2017b); Tanneberger *et al.* (2017), or IMCG (2021).

### 2.2.3 Legacy data in peatland research

Meta-analyses of composite datasets often provide new supraregional insights and may point to knowledge gaps that need to be addressed by future research. For

instance, in an extensive study by Tolonen and Turunen (1996) on carbon accumulation in Finnish mires over 1000 dated peat cores were analysed, combining material from over 30 publications of various regions in Finland. Their analyses enhanced understanding on potential effects of climate warming for different mire types in Fennoscandia. Various studies include legacy dates of basal peat layers to enhance understanding of Holocene sea level rise (e.g. Berendsen *et al.*, 2007; Meijles *et al.*, 2018; Hijma and Cohen, 2019) or to increase insights in peat compaction and land subsidence (e.g. Koster, 2017). Ruppel *et al.* (2013) studied trends in peatland initiation in North America and northern Europe, through analyses of 1400 retrieved basal peat dates. Their results not only provided insights in spatiotemporal trends in peat initiation but also indicated a lack of (retrieved) data for the Northwest-European Plain. Future studies –including the research presented in this paper– may complement this image and further develop our understanding of peatlands through space and time, the influence of autogenic processes and feedbacks, and allogenic causes for changes in peatland dynamics.

### 2.2.4 Radiocarbon dating

For environmental reconstructions based on peat archives, radiocarbon  $(^{14}C)$  dating is the preferred method to connect stratigraphies to an absolute time scale. We provide a concise explanation of the radiocarbon method, as well as a summary of the development of its measurement techniques (fig. 2.1). This is relevant when using data obtained with techniques subject to methodological changes.

Radiocarbon dating is based on the radioactive decay of  ${}^{14}C$ . This isotope is produced in the upper atmosphere by cosmic radiation. It oxidizes to  ${}^{14}CO_2$  which is incorporated in living organisms through photosynthesis and the food chain. Upon death of the organism, the radioactive decay of  ${}^{14}C$  enables to derive its age (i.e. timing of death).

Although the principle is relatively simple, complications do exist. First, changes in cosmic ray flux and geomagnetic field strength cause variations in the production rate of <sup>14</sup>C through time (De Vries, 1958). This requires <sup>14</sup>C dates to be calibrated in order to express them in calendar years. This is primarily done by <sup>14</sup>C dating of tree rings, which are dated absolutely by dendrochronology. Second, isotopic fractionation (mass dependent effects) during photosynthesis leads to depletion of the heavy isotopes <sup>13</sup>C and <sup>14</sup>C in the plants, the latter causing age aberrations for which measurements need to be corrected. Third, the half-life of <sup>14</sup>C is 5,730±40 years, where originally a value of 5,568 years was used by Libby *et al.* (1949) who developed the <sup>14</sup>C method.

In its early days,  ${}^{14}C$  dates were reported in BP (Before Present, defined as 1950), using the natural  ${}^{14}C$  content as a reference. It soon became clear that



**Figure 2.1:** Timeline of developments in radiocarbon dating (right). Changes at the radiocarbon laboratory in Groningen (left) are relevant for the case study.

this is problematic because of the complications mentioned above. These are solved by the Radiocarbon Convention, which defines the  $^{14}$ C timescale (Stuiver and Polach, 1977; Van der Plicht and Hogg, 2006):

- i) The <sup>14</sup>C radioactivity is measured relative to that of a modern reference material, i.e. Oxalic Acid with a radioactivity of 0.226 Bq/g C;
- ii) From this measured radioactivity the "radiocarbon date" is calculated using a half-life of 5,568 years;
- iii) Radiocarbon dates are corrected for fractionation using the stable isotope  ${}^{13}$ C (to a reference value  $\delta^{13}$ C = -25‰, see below);
- iv) Radiocarbon dates are expressed in the unit BP.

The original half-life value was chosen to keep the meaning of earlier reported dates unchanged. The chosen value for the  $\delta^{13}$ C reference value is that of charcoal, wood and plants (including peat). The convention means that BP should not be taken literally: <sup>14</sup>C years differ from calendar years, and present is not today (or 1950). Calibration transfers <sup>14</sup>C dates into calendar dates. These are expressed in cal BP, which is defined as calendar years before 1950 CE. The calibration curves are regularly updated (fig. 2.1).

In radiocarbon practice, the  $\delta^{13}$ C and %C values are indicators for sample integrity (Mook and Streurman, 1983). When these values are not within the accepted range, the organic sample material is usually degraded, or there is contamination. They are therefore an integral part of <sup>14</sup>C dating, also for legacy data. <sup>13</sup>C is a stable isotope and thus its concentration is time-independent. It can therefore be used as a measure of fractionation of the photosynthesis process. Since  $\delta^{14}$ C = 2  $\delta^{13}$ C, we then also know the fractionation effect for <sup>14</sup>C, and thus the age deviation caused by this process. In addition, fractionation effects during laboratory procedures are taken into account automatically. The  $\delta^{13}$ C is defined as  $\delta^{13}$ C = [<sup>13</sup>R<sub>sample</sub> - <sup>13</sup>R<sub>reference</sub>]/[<sup>13</sup>R<sub>reference</sub>] (x1000‰), where <sup>13</sup>R=[<sup>13</sup>CO<sub>2</sub>]/[<sup>12</sup>CO<sub>2</sub>]. The reference is a belemnite known as PDB, with a well-known <sup>13</sup>C/<sup>12</sup>C isotope ratio (Mook, 2006, and references therein).

In the early days of radiocarbon (the 1950's, i.e. before the Convention),  $\delta^{13}C$  was not measured, and fractionation correction was not applied. Significance of  $\delta^{13}C$  is dependent on the type of photosynthesis used by plants, known as the C<sub>3</sub> and C<sub>4</sub> pathways. For C<sub>3</sub> plants, the  $\delta^{13}C$  value is around -25‰, not very different from the reference value so that fractionation corrections are small, within the measurement uncertainty and negligible. Therefore our peat dates that were not corrected for isotopic fractionation (i.e. measured before the Convention) are still useful. For completeness we note that for C<sub>4</sub> plants, the  $\delta^{13}C$  value is around -10‰ which leads to large fractionation corrections; here the difference with the reference value ( $\delta^{13}C$ =-25‰) is around 15‰, which corresponds to an effect of 240 BP for <sup>14</sup>C and cannot be neglected. Thus, for regions containing C<sub>4</sub> plants it will be necessary to correct previously uncorrected dates for the fractionation effect.

The %C refers to the organic C content of the sample after the pre-treatment (ideally ABA, the Acid-Base-Acid method) designed to isolate the pure datable fraction. This is different from the organic content of the original peat, such as measured with loss-on-ignition, where the weight loss is measured of dried untreated material before and after combustion at high temperature (e.g. Chambers *et al.*, 2011; Kennedy and Woods, 2013). The lower the carbon content of the <sup>14</sup>C sample, the larger the effect of contamination (i.e. all carbon that was not related to the sample when it was alive) will be (Lanting and Van der Plicht, 1994).

Initially, radiocarbon concentrations were measured by radiometry. This method requires large quantities (typically 1 gram) of carbon (Cook and Van der Plicht, 2013), meaning that only bulk samples could be dated. In the 1970s, Accelerator Mass Spectrometry (AMS) was developed for direct measurement of <sup>14</sup>C concentrations in a sample. This method is much more efficient, enabling dating samples of typically 1 milligram (Tuniz *et al.*, 1998). The most recent development is that of mini-AMS systems (MICADAS), based on the same technology but much smaller machines. Radiometry is still applied at some laboratories. Radiocarbon laboratory codes (available at www.radiocarbon.org) provide unique identifiers for dates and immediately provide information where the date was measured, often also on the measurement method (conventional or AMS).

For more information on radiocarbon see e.g. Bayliss *et al.* (2004); Walker (2005); Ramsey (2008b) and refer to e.g. Taylor (2000); Olsson (2009), and Wood (2015) for its development.

#### 2.2.5 Case study selection and aims

Palaeogeographic maps are often built though integration of (legacy) data from various sources (Pierik and Cohen, 2020). Current palaeogeographic reconstructions of the Netherlands (Zagwijn, 1986; Westerhoff et al., 2003; Vos, 2015a) were created with strong focus on the development of river deltas (e.g. Berendsen and Stouthamer, 2000, 2001) and coastal area (e.g. Hijma, 2009; Cohen et al., 2014; Pierik et al., 2017). In contrast, reconstructions of inland peatlands remain uncertain due to limited data on these areas (Spek, 2004; Van Beek, 2009; Vos, 2015a). Increased understanding of their spatiotemporal dynamics is needed to refine representation of these landscapes on the palaeogeographic map series, for development and validation of peat growth models (e.g. Kleinen et al., 2012), and related quantification of their role in past, present and future carbon cycles (e.g. Yu et al., 2011; Erkens et al., 2016). Furthermore, insight in peatland palaeogeography is key to understand (pre)historic human habitation (Van Beek, 2015; Van Beek et al., 2015) and to contextualise exceptional archaeological finds from wetlands (e.g. Chapman et al., 2019; Van Beek et al., 2019). Given the limited understanding on the development of Dutch inland peatlands, we selected a part of the coversand landscape in the Netherlands (fig. 2.2) as case study region, focusing on the northern area (approximately  $4,700 \text{ km}^2$ ). Our research questions within the case study are:

- i) Where and when was peat present at the surface in the study area?
- ii) In what way does the period of peat initiation differ between landforms and related elevation?

iii) What can be inferred about processes responsible for (inland) peat initiation (and lateral expansion) based on exploratory data analyses?

## 2.2.6 Case study area

The (northern) coversand landscape (fig. 2.2) is characterised by diverse landforms, enabling peat growth in various geomorphological settings, and is representative for larger parts of the Northwest European Plain considering its surface and shallow subsurface deposits (more information below). The study area contains the coversand landscape stretching from its northern limits down to the rivers Reest and Schoonebeekerdiep as southern borders (fig. 2.2b and 2.2c). Parts of the region belong to a national park, a UNESCO Global Geopark and several Natura 2000 reserves.

During part of the Saalian (MIS 6), the northern Netherlands was covered by a continental ice sheet, leading to deposition of glacial till (Rappol, 1987; Van den Berg and Beets, 1987; Rappol et al., 1989; TNO – Geological Survey of the Netherlands, 2021b). The central part of the study area is known as the Drenthe Plateau or till plateau (Ter Wee, 1972; Bosch, 1990). Meltwater scoured deep valleys east and south of the Drenthe Plateau, the Hunze valley (Bosch, 1990) and palaeo-Vecht valley (Ter Wee, 1966; Bosch, 1990; Kuijer and Rosing, 1994) respectively. The area east of the Hunze valley is also known as the Hunze Plain (Groenendijk, 1997). Here, fluvio(peri)glacial sands were deposited during the Saalian (in this part of the study area glacial till is only sporadically found; Bosch, 1990). In the Weichselian, the Drenthe Plateau became dissected by incising rivers, consequently glacial till is largely absent in river valleys (Klijnstra, 1979). During the coldest phase of the Weichselian coversand was deposited with thicknesses varying from 0.5 – 2 m (Ter Wee, 1979; TNO – Geological Survey of the Netherlands, 2021c). This deposit is present at the surface in the northeastern, eastern and south-eastern parts of the Netherlands (fig. 2.2c) and the larger European Sand Belt (Koster, 1988, 2005).

Peat deposits in the study area formed both on the low- and high-lying plains (e.g. Casparie, 1972, 1993), in river valleys (e.g. Candel *et al.*, 2017) and in fossil pingos (e.g. De Gans and Sohl, 1981). Based on historical data peat thickness on the plains appears to have reached at least 7 m at some sites (Fochteloërveen, Douwes and Straathof, 2019), maximally 7 m in the largest pingos (Stokersdobbe, Paris *et al.*, 1979), and locally at least 7 m in river valleys that were deeply incised prior to the Holocene (Drentsche Aa river, Candel *et al.*, 2017).

In the northern Netherlands large-scale peatland reclamations took place from the eleventh and twelfth century onwards. These were initiated by monasteries



Figure 2.2: Location and (palaeo)environmental characteristics of the case study region. (a) Location of the Netherlands in Europe. (b) Digital Elevation Model (DEM) of the northern coversand landscape. Elevation is in metres relative to Dutch Ordinance Datum (O.D., roughly mean sea level). Within the study area the present day positions of some of the largest rivers are shown to indicate main drainage directions. The two biggest peat remnants (nature reserves, Fochteloërveen and Bargerveen) are highlighted in yellow. Dataframe coordinates are in metres (Dutch RD-new [Rijksdriehoeksstelsel] projection). Position of the study area in the Netherlands is indicated in (c-e). (c) Extent of coversand in the Netherlands. (d) Reconstructed palaeogeography for 2,500 cal y BP, indicating assumed former extent of Dutch peatlands. For peatlands in the coversand region, this view is less certain (see text). Legend was simplified in (c) and (d), full details can be found in Vos and Vries (2013) and Vos et al. (2020). In (e) current distribution of peat soils (i.e. containing >20% organic matter) is shown. Sources: DEM of the Netherlands (AHN2: horizontal resolution 5 m, vertical resolution 0.2 m) from Van Heerd et al. (2000), AHN (2018); coversand extent and Dutch palaeogeography from Vos and Vries (2013) and Vos et al. (2020); rivers in the study area from Ministerie van Verkeer en Waterstaat (2007) Ministerie van Verkeer en Waterstaat (2007); Dutch soil map from Alterra (2014); two largest peat remnants (Natura 2000 areas) from Ministerie van Economische Zaken - Directie Natuur & Biodiversiteit (2018).

and local landlords, originally for agricultural purposes (Van Beek *et al.*, 2015). From the late sixteenth century onwards commercial-scale peat-cutting for fuel became dominant (Gerding, 1995). As a result only small remnants of the former peat landscapes remain (fig. 2.2d and fig. 2.2e).

## 2.3 Approach and methods

We propose a workflow for data rescue in geochronological peatland research (fig. 2.3, section 2.3.1), which involves a rigorous quality assessment of legacy data. The rationale of this procedure is threefold:

- i) To assist in systematically recording properties of legacy dates, using quantitative information and uniform qualitative categories where possible;
- ii) To enable evaluation of data on various quality aspects, either determined by technical aspects of the date, properties of the date related to its landscape position, or both (section 2.3.1.2);
- iii) To assign a penalty score to each date based on case-specific weights for quality aspects (section 2.3.1.3). This enables taking data quality into account in subsequent meta-analyses, to test for sensitivity using subgroups of data with different quality levels, and to safeguard robustness of conclusions.

To evaluate the power of the proposed methods, we apply the workflow to a case study, for which we have formulated three research questions on spatiotemporal peatland dynamics (section 2.2.5). For answering these questions, we tailor the proposed workflow as explained in section 2.3.2. Based on the process of data rescue and meta-analysis in the case study we identify research deficits to address during future studies and evaluate the value of data rescue in geochronological peatland research.

## 2.3.1 Workflow, database set-up and quality assessment

### 2.3.1.1 Overview of the workflow

The workflow for data rescue and reuse (fig. 2.3) consists of:

- i) A database set-up that can be tailored as required by the study scope (table 2.1);
- ii) A complementary set of quality criteria with flexible weights to suit specific research questions (table 2.2, see section 2.3.1.3 for more explanation on the use of weights);
iii) A script for automated quality assessment of the recorded legacy data using the weights defined in point (ii), to make the approach suitable for evaluating large legacy datasets.

Based on the literature discussed in section 2.2.4, we propose quality criteria that consider technical aspects of radiocarbon dating and sample selection, while other criteria are concerned with landscape position and taphonomy of the dated material (section 2.3.1.2). A penalty is assigned for traits that are considered negative (table 2.2). To allow automation of the quality screening process, quantitative information is used in the database where possible. Additionally, discrete and Boolean categories were defined that can be used to standardize qualitative descriptions. The quality assessment was scripted in Python, to automatically assign a penalty score to each date.



**Figure 2.3:** Proposed workflow for data rescue, quality assessment and meta-analysis in geochronological (peatland) research. See legend in upper right for explanation of colours.

	Table 2.1: Proposed databa	ise structure. For all categories 'NR' indicates not retrieved/reported.
GENERAL	LabCode	Laboratory code as assigned by the lab that conducted the analysis.
	Add LabCode	Any additional lab codes that were assigned to the date.
	DatingTechnique	Either 'beta-counting' (i.e. radiometry) or 'AMS'. As the first AMS measurements were performed in 1977 (see fig. 2.1) all samples analysed before 1977 will automatically be categorised under 'Beta-counting'.
	YearOfDating	Year when the sample was dated by the laboratory. In case a period is given, the earliest year is recorded.
	CopyrightLicense	License under which the data was published (if applicable).
	ConsultedLab <sup>1</sup>	Either 'yes' or 'no' to indicate whether the lab was consulted for additional details on the sample.
	Reference	Original publication(s) that mentioned the date.
	ReliabATOrigPub	Any comments about whether the date was reliable according to its original publication.
	Notes	Any other comments (e.g. whether a lithological or lithogenetic cross-section is available, short description of the definition for basal peat as used by the original publication that mentioned the date).
AGE	14CMean	Reported mean age in conventional <sup>14</sup> C years (BP).
	14CSD	Reported standard deviation of the mean age in $^{14} ext{C}$ years.
	Reservoir Corrected 14 CM ean	If applicable and reported, the mean age in $^{14}$ C years (BP) after applying a reservoir correction.
	CalMean	Reported mean calibrated age in years BCE (Before Common Era) $/$ CE (Common Era).
	CaISD	Reported standard deviation of the mean age in calendar years.
	UnknownMean	Reported mean age, unknown whether the data represent uncalibrated or calibrated ages.
	UnknownSD	Reported standard deviation of the mean age, unknown whether the data represent un- calibrated or calibrated ages.
	Delta13	$^{13}\delta$ value used to correct the dating result for isotopic fractionation.
		Continued on next page

	Ta	ble 2.1 – continued from previous page
	Delta13source	Either 'measured', 'estimated' to indicate whether the reported $^{13}\delta$ value was quantified in the laboratory or that a standard estimate was applied.
	CarbonContent	Measured carbon content (%C) of the dated sample material.
LOCATION	LocationDescription	Description of the location where the sample was taken.
	X <sup>2</sup>	Column listing all X-coordinates in the Dutch RD-new projection, either reported or trans- formed from reported longitude.
	Υ 2	Column listing all Y-coordinates in the Dutch RD-new projection, either reported or trans- formed from reported latitude.
	XYuncertainty	Reported uncertainty of measured X and Y coordinates.
	Lat	Column listing latitude, in case location is only reported as geographical coordinates.
	Lon	Column listing longitude, in case location is only reported as geographical coordinates.
	LLuncertainty	Reported uncertainty of measured latitude and longitude, in case location is only reported as geographical coordinates.
	XYcategory <sup>3</sup>	Type of coordinate data; either 'recorded', 'field' or 'placename'.
ELEVATION	DepthBelowSurface	Reported depth of the sample (in case only one number is given) below the surface at the time of sampling, given in cm.
	DepthBelowSurfaceFrom	Reported depth of the upper boundary of the sample below the surface at the time of sampling, given in cm.
	DepthBelowSurfaceTo	Reported depth of the lower boundary of the sample below the surface at the time of sampling, given in cm.
	DepthNAP	Reported depth of the sample (in case only one number is given) in m NAP (Dutch Ordnance Datum, approximately mean sea level).
	DepthFromNAP	Reported depth of the upper boundary of the sample in m NAP (Dutch Ordnance Datum, approximately mean sea level).
	DepthToNAP	Reported depth of the lower boundary of the sample in m NAP (Dutch Ordnance Datum, approximately mean sea level).
		Continued on next page

29

	Ta	ble 2.1 – continued from previous page
LANDFORM AND STRATIG- RAPHY	Landform	Either 'Valley', 'Pingo' 'Depression', 'Peatland (unspecified)', 'Ridge', 'Mound', 'Channel fill', 'Bog (hollow)', or 'Bog (hummock)'.
	Stratigraphy	Either 'Upperlimit', 'Lowerlimit', or 'Within' to indicate whether the sample was taken at the top of the peat layer, the base of the peat layer, or somewhere within the peat layer, respectively.
SAMPLE IN- FORMATION	SampleThickness <sup>4</sup>	Sample thickness in cm.
	SampleDetails	Any information on the sample material, species, etc.
	SampleContext	Either 'Environmental', 'Artefact' or 'Anthropogenic' (the latter in case of sediment sam- pled from anthropogenically disturbed contexts).
	SampleMaterial	Peat / Organic / Gyttja / Wood / Charred / Carbonate / <i>Gliede</i> (Dutch term used for amorphous organics in peatlands) / Dopplerite.
	MeasuredFraction	Details on measured fraction as provided by the referenced paper for the date.
	SampleType	Either 'bulk' or 'macro' to distinguish between bulk samples and samples consisting of plant macrofossils.
	SpeciesType <sup>5</sup>	In case of a 'bulk' SampleType this automatically becomes 'undefined'. In case of a 'macro' SampleType this column lists either 'terrestrial', 'aquatic', 'both', or 'NR'. Terrestrial and aquatic distinguish between plant species that might be affected by a reservoir effect.
	Aboveground	In case of a 'bulk' SampleType this automatically becomes 'undefined'. In case of a 'macro' SampleType either 'yes', 'no' or 'NR' must be selected to distinguish between seeds/leaves/stems or roots respectively.
	Pretreatment	Either 'ABA' or 'onlyA' to indicate whether the ABA protocol was used or only the first A, respectively.
		Continued on next page

1 In case lab data differed from data provided by publications, the lab was followed. For one entry in the case study this led to a major change in the age (GrN-5460, the datelist in which this date is mentioned contains an error).	2 In case no coordinates were provided by the original publications but the field of sampling was indicated, coordinates were retrieved based on georeferenced historical maps where the old field boundaries were indicated. If only a nearby village was mentioned, its central coordinates were registered.	3 These categories may after the quality assessment be replaced by a case-study dependent value, to be used as uncertainty range in further analyses (e.g. 'field' might be replaced by an uncertainty value of 100 m).	4 In case no sample thickness was retrieved, a default sample thickness can be defined after the quality assessment to use in further analyses.	5 To determine whether terrestrial or aquatic plants were used for dating the species was looked up in the Dutch online encyclopaedia of species (Soortenbank.nl, 2020). Samples that consisted of <i>Sphagnum</i> mosses that lacked further information on terrestrial or aquatic growth habit were listed as 'Undefined'.
---	--	---	---	--

	Subscore dating penalty Subscore taphonomic Total penalty score (Q) $(Q_d)$ penalty $(Q_T)$	Disqualified	are re- A sample was dated $Q_{d1}$ rr calen- but the age was not	was re-	$Q_{d3}$	. Location vas not $Q_T 1$ retrièved.	QT2	was col- QT3	of sam-	not re- QT5	$Q_{d4}$	$Q_{d5}$	of a bulk Q <sub>d6</sub>	d or not Qd7	
enany.		+2	Unknown whether data a ported in radiocarbon or dar years.	No information on $^{13}\delta$ v trieved.	Not retrieved.	Reported at village level.	Not retrieved.	Landform where sample w lected was not retrieved.	Stratigraphical position of ple was not retrieved.	Sample thickness was n trieved.	Not retrieved.	Undefined (i.e. in case of sample) or not retrieved.	Undefined (i.e. in case of sample) or not retrieved.	No pre-treatment applied retrieved.	
d callin ungew gi	Criteria	+1	Only calibrated data are re- ported, or SD in radiocarbon years is missing.	$13\delta$ value is reported but was estimated.		Reported at field level (3- number RD-coordinates; degrees and minutes for geo- graphic coordinates).	Reported in m. relative to (for- mer) surface.				Bulk sample (AMS or conven- tional dating).	Sample consisted of material from aquatic species, or both terrestrial and aquatic.	No, (also) belowground sample material (i.e. including roots).	Gentle (A only).	
ופופט איז ווועונואואוו		0	Mean and standard deviation are reported in radiocarbon years.	$13~\delta$ value is reported based on measurements.	Carbon content was reported.	Reported at site level (6-number RD-coordinates; degrees, minutes and seconds for geographic coordinates).	Reported in m. O.D.	Landform was described or clear from context.	Stratigraphical position of sample was clear from text and/or geological cross sections and allows teal cross sections between top/middle/bottom of peat layer.	Sample thickness was reported.	Macrofossil (AMS dating).	Sample consisted of material from terrestrial species.	Yes, aboveground sample mate- rial (no roots).	Robust (ABA).	
rairu	(Case-specific) weight		10	ъ	2	o	œ	9	2	2	2	ى ا	2	ы	
	Property		Mean and SD	Delta13	Carbon content	×,×	Elevation	Landform	Stratigraphy	SampleThickness	SampleType	SpeciesType	Aboveground	Pre-treatment	
ppilcable. Jul	Category		AGE			LOCATION	ELEVATION	LANDFORM AND STRATIGRAPHY		SAMPLE DETAILS					

32

### 2.3.1.2 Quality criteria

### 2.3.1.2.1 Definition of quality

For constructing the quality assessment one has to decide what quality means, and for which properties it must apply. According to the Cambridge Dictionary (2020), quality means 'how good or bad something is' or 'a characteristic or feature of [...] something'. Both definitions are used in our quality assessment (see below).

When considering radiocarbon dates of peat layers, each date's quality may be assessed for its dating quality  $(Q_d)$  and taphonomic quality  $(Q_T)$ . Dating quality refers to technical aspects of the radiocarbon date, i.e. sample characteristics and the way it was processed in the lab. Taphonomy, a term originating from palaeontology, is the science on how materials (or fossils) become embedded in their surroundings (e.g. Martin, 1999). The taphonomic quality therefore refers to characteristics of where the sample came from, e.g. its location and stratigraphical position. The degree to which a radiocarbon date represents the event of interest is determined by its dating and taphonomic quality. Both  $Q_d$  and  $Q_T$  are determined by the approach and methods that were followed by the researchers from the original study the date was obtained from. Fig. 2.4 provides a visualisation of the effects of methodology on the resulting  $Q_d$  and  $Q_T$ . As dating approaches were tailored to answer a particular research question (with a certain required level of certainty),  $Q_d$  and  $Q_T$  may diverge for radiocarbon dates originating from different studies.

In textbook examples where a bullseye is used to illustrate accuracy and precision, these concepts usually apply to a set of replicate measurements. Note that in fig. 2.4 and the explanation above,  $Q_d$  and  $Q_T$  apply to the accuracy of a single measurement (i.e. the degree to which a date represents the true age of the event of interest), and that precision (i.e. degree to which replicate measurements lead to the same result) is not indicated in fig. 2.4. The possibility to replicate a date is however fully dependent on the information contained in  $Q_d$  and  $Q_T$ , therefore a high penalty score for  $Q_d$  and  $Q_T$  will most likely result in low precision (e.g. if location is poorly known, attempting a replicate measurement cannot be performed with a high precision).

To ensure accuracy and robustness of conclusions derived from meta-analyses, quality assessment may provide insight in sources of error and allows to expand data analyses based on subsets of data with increasing uncertainty. To make the quality assessment flexible to answer a variety of research questions, we have created an adaptable, twofold approach. First, each date is evaluated for aspects that are considered negative (i.e. in line with the abovementioned definition 'how good or bad something is') for which penalties are assigned (for comparable approaches see e.g. Small *et al.*, 2017). For instance, a bulk sample is considered less reliable



**Figure 2.4:** Effect of methodology (dating and taphonomic quality) on representation of the true age of the event of interest (bull). The distance to the bull indicates how robust a date is, i.e. the degree to which the date corresponds with the true age of the event of interest. Note that multiple black dots (i.e. potential dating results) were drawn for the purpose of illustrating the effect of dating quality and taphonomic quality, whereas in reality they apply to a single measurement. In case of the lower left for example, the approach ensures a sample is collected from the right position (e.g. location, elevation, stratigraphical level), and strict methods are applied with regard to sample selection and laboratory procedures. With low dating and/or taphonomic quality, dates will deviate more from the true age of the event of interest. Our approach aims at attributing penalties to those dates in the quality assessment, as a way to characterise their trustworthiness and usefulness to answer a specific research question.

than a plant macrofossil sample (Törnqvist *et al.*, 1992, 1998). Second, each date is assessed on the availability of information that allows to make informed choices with regard to data-analysis (i.e. 'a characteristic or feature of something'). In case of missing information or a low level of detail, a penalty is assigned. In this case the focus is not on the implication of the property (for instance, the location itself is not judged), but on knowledge *about* the property (do we know location well or not). Depending on the information that is available (and the resulting penalty score), data may be filtered prior to data-analysis (for example first including only sites with well-known location, then analysing the sites with uncertain location as well). This allows a purposeful assignment of dates to various analyses.

### 2.3.1.2.2 Design of the quality criteria

Age

This category contains criteria for three properties: *Mean and SD*, *Delta13*, and *Carbon content* (table 2.2). The property *Mean and SD* distinguishes the way in which the age is retrieved. Radiocarbon measurements are reported in BP, which require calibration to calendar years. For (re)calibration the original date in BP with its standard deviation is required. The property *Delta13* measures deviations caused by isotopic fractionation and differentiates whether  $\delta^{13}$ C was measured, estimated in the original work or unknown (i.e. not reported). Carbon content refers to the %C of the <sup>14</sup>C sample after laboratory pre-treatments and is either measured or unknown.

The quality assessment does not distinguish between samples that were dated with radiometry or AMS. Both measure the  $^{14}C/^{12}C$  ratios using the same reference sample and background materials; it is the sample size that makes the difference (Lanting and Van der Plicht, 1994). The sample size is included in the quality assessment as a property under *Sample Details* through *SampleType*.

### Location

This category contains the combined property X, Y. The criteria are concerned with the level of detail regarding the location of the dated material and distinguish between recorded coordinates, field level or place names. If relevant for the research questions to be answered, these categories may be replaced by a case-study dependent value, to be used as uncertainty range in further spatial analyses (e.g. 'field' could be replaced by an uncertainty value of 100 m).

### Elevation

This category contains one property of the same name, and relates to the level of detail regarding the elevation of the dated material. This can either be known

relative to O.D., relative to the (former) land surface, or not retrieved.

### Landform and Stratigraphy

This category contains *Landform* and *Stratigraphy* as two properties, each distinguish whether these properties were clear from the context of the date as provided by its source. If clear, filtering after the quality assessment allows for the selection of samples e.g. from specific stratigraphic positions, such as basal peat layers. Note that only if information on landform and stratigraphy was retrieved with the date, the information was registered and available for further analyses. We did not deduce landform from sample location (for some studies attempting legacy data analyses this might be an interesting option, depending on the level of detail of retrieved coordinates).

### Sample Details

For this category five properties were included: *SampleThickness*, *SampleType*, *SpeciesType*, *Aboveground* and *Pre-treatment*. *SampleThickness* distinguishes whether thickness was reported or not. If thickness has consequences for the research questions to be answered, filtering after the quality assessment allows selection of samples of certain thickness ranges.

Based on the recommendation to date short-lived, aboveground plant macrofossils of terrestrial species (e.g. Törnqvist *et al.*, 1992; Piotrowska *et al.*, 2011) we have formulated the criteria for *SampleType*, *SpeciesType*, and *Aboveground*. *SampleType* differentiates macrofossil samples (dated with AMS) and bulk samples (mostly conventional dating) and implicitly contains information about sample size (mentioned above under Age). *SpeciesType* is concerned with the habitat of the organism(s) that were sampled, either terrestrial species, aquatic species, both, or undefined (i.e. in case of a bulk sample). The property *Aboveground* refers to whether only aboveground plant remains were present in the sample material (no roots) or that belowground tissues were also included (presence of roots leads to incorporation of younger carbon, e.g. Törnqvist *et al.* (1992). For bulk samples this automatically becomes undefined.

*Pre-treatment* distinguishes the preparatory protocols applied in the radiocarbon laboratory prior to measurement. This can either be robust pre-treatment (ABA, Acid-Base-Acid), a gentle one (A only), or none. The question to opt for A or ABA is closely connected to the %C parameter. Contaminations (such as mobile humic acids) are most adequately removed by a robust pre-treatment. However, when the amount of sample material is limited a gentle (or no) pretreatment may be applied to ensure preservation of sufficient material for dating.

### 2.3.1.3 Weights in the quality assessment and interpretation of penalties

For each data entry, the taphonomic quality  $Q_T$  and dating quality  $Q_d$  are calculated using the quality criteria and (case-specific) weights listed in table 2.2. In case a specific criterion is irrelevant for the research questions to be answered, it can be assigned a weight of zero and will then no longer be considered. Depending on the case study and research aims weights may be adapted to tailor the quality assessment.

The total penalty score Q results from the sum of  $Q_T$  and  $Q_d$ . Q is normalized to 1, i.e. the minimum value is 0 (no penalties, reflecting highest quality) and the maximum possible value is 1 (poorest quality). Due to this normalization the maximum values of  $Q_T$  and  $Q_d$  are always below 1 and do not need to be equal, as they depend on the chosen weights. For instance, in our case study (section 2.3.2) the weights listed in table 2.2 are used which results in maximum values of  $Q_T$  and  $Q_d$  of 0.464 and 0.536 respectively. The normalized  $Q_T$ ,  $Q_d$ and Q values may be used as such, or may be converted to four confidence levels based on user-determined cut-off values for  $Q_T$  and  $Q_d$ , defined by  $Q_{T,lim}$  and  $Q_{d,lim}$  (table 2.3).

Table 2.3: Definition of the four confidence level.
---

Confidence level	Definition		
Low $Q_T$ , low $Q_d$ (best)	$0 \le Q_T < Q_{T,lim}$	AND	$0 \le Q_d < Q_{d,lim}$
High $Q_T$ , low $Q_d$	$Q_{T,lim} \le Q_T \le Q_{T,max}$	AND	$0 \le Q_d < Q_{d,lim}$
Low $Q_T$ , high $Q_d$	$0 \le Q_T < Q_{T,lim}$	AND	$Q_{d,lim} \le Q_d \le Q_{d,max}$
High $Q_T$ , high $Q_d$ (worst)	$Q_{T,lim} \le Q_T \le Q_{T,max}$	AND	$Q_{d,lim} \leq Q_d \leq Q_{d,max}$

### 2.3.2 Application of the workflow to a case study

### 2.3.2.1 Case study: Data rescue and quality assessment

The data search scope was determined by the spatial definition of the study area presented above (fig. 2.2). All acquired dates were recorded irrespective of their measured radiocarbon age (no restrictions in time period were applied during the search phase). Data originate from 1955 to 2019 and stem from a wide variety of environmental and archaeological studies, including scientific literature, books and reports from contract-based archaeology. We used the database set-up of table 2.1 and recorded dates from peat layers (i.e. excluding dated archaeological artefacts originating from peat layers).

The majority of retrieved dates were performed by the radiocarbon facility of Groningen University (Centre for Isotope Research and its predecessors). The history of this laboratory is shown in fig. 2.1. Developments are also reflected

in labcodes, moving from GRO and GrN (conventional measurement) to GrA (AMS) and GrM (AMS-MICADAS). Data registration changed along with these transitions, resulting in three large archives that evolved from hardcopy to semidigital and now fully digital (Van der Plicht, 1992; Van der Plicht and Streurman, 2018). Consequently, data retrieval required both digital querying and hardcopy searching.

### 2.3.2.2 Case study: Meta-analysis

To answer the case study research questions (section 2.2.5), the quality assessment was adapted by choosing appropriate weights for the criteria. Subsequent metaanalysis of the resulting assessed dataset included three main elements: largescale trends in peat initiation, trends for different landforms (and elevations), and a comparison between peat initiation trends with sea level and climate. More details are provided below.

We chose the criteria weights listed in table 2.2. In this way, the penalty contribution of each criterion is ordered based on the qualities we consider most important to answer the case study research questions. For these questions age and location are crucial, followed by elevation, stratigraphy and landform. To prevent qualities from becoming irrelevant, we kept the difference in weight between criteria relatively small. Based on the penalty scores each date was assigned to one out of four confidence levels based on the definitions listed in table 2.3, where  $Q_{T,lim}$  and  $Q_{d,lim}$  were set at 50% of their respective maximum values.

After completing the quality assessment filtering was applied based on (1) confidence level, (2) *Stratigraphy* (to select only basal peat dates), (3) *SampleMaterial* (to distinguish peat initiation processes, explained below), and (4) *Landform* (to derive landform-specific age trends). For analyses on the relationship between age and elevation, we calculated elevation relative to m O.D. for samples that were only retrieved with depth to the (former) surface. To this end, we derived the surface elevation from the DEM and subtracted the sample depth. For basal dates, elevation is not affected by compaction effects. Dates from within the peat or the top might be affected by compaction effects. However, as we only used these data for a general overview of the elevation range from which samples were retrieved, they were not corrected for compaction.

All ages were (re-)calibrated in OxCal (version 4.4, Ramsey, 1995, 2009) using IntCal20 (Reimer *et al.*, 2020). To analyse trends of peat initiation dates of basal peat layers (i.e. entries registered with stratigraphy 'lowerlimit') were selected and summarised using kernel density estimation (KDE) with the KDE\_Model function in OxCal (Ramsey, 2017). To test model outcome for sensitivity to previously assessed data quality, the data subsets from the four confidence levels (fig. 2.5)



**Figure 2.5:** Overview of the quality assessment of the case study dataset (n = 313), showing resulting  $Q_d$  and  $Q_T$  values for each data point (note that some points overlap). Limits of the confidence levels are defined in table 2.3, with  $Q_{d,lim}$  and  $Q_{T,lim}$  set at 50% of their respective maximum normalized values. The coloured quadrants indicate the four confidence levels that were used in subsequent data analyses.

**Table 2.4:** Chronostratigraphy as used in this paper. The Pleniglacial started before 55,000 cal years BP, i.e. the upper limit of IntCal20 (which matches the measurement limit of  $^{14}C$ , ca. 50,000 BP).

Period	From (cal years BP)	To (cal years BP)
Subatlantic	0	2,400
Subboreal	2,400	5,660
Atlantic	5,660	9,220
Boreal	9,220	10,640
Preboreal	10,640	11,560
Late Glacial	11,560	14,650
Pleniglacial	14,650	>55,000

were added to the model in separate runs and outcomes compared.

To derive spatiotemporal insights on peat initiation, data were plotted in GIS (ESRI's ArcMap, version 10.6) using the chronostratigraphy shown in table 2.4. To assign dates to the listed periods the  $\mu$  value of the calibration was used for simplicity (i.e. instead of the  $2\sigma$  age range). Similarly,  $\mu$  was used to construct age-elevation plots.

To determine which peat initiation process (terrestrialisation, primary mire formation or paludification) was responsible for peat formation at a specific site, the

Peat initiation process	SampleMaterial filter	Green confidence level, n:	All confidence levels, n:		
Terrestrialisation	Gyttja	5	6		
Either primary mire forma- tion or paludification (indis- tinguishable)	<i>Gliede</i> (Dutch term used for amorphous organics in peatlands) Peat Wood	43	64		
Paludification	Charred	2	4		

 Table 2.5: Classification of SampleMaterial to derive peat initiation process.

sediment underlying basal peat often provides indications (Ruppel *et al.*, 2013). Typically, peat from terrestrialisation is underlain by lake sediments such as gyttja. Primary mire formation starts on inorganic sediment where fresh parent material is exposed, whereas paludification occurs on inorganic sediment where soils have formed through time, sometimes with litter layers of past vegetations. Unfortunately, information on soil horizons underlying peat deposits is limited for our case study data. To determine the prevalence of these three processes in the study area, we therefore assigned basal peat dates to each initiation process based on registered *SampleMaterial* (table 2.5).

# 2.4 Results

## 2.4.1 Data rescue for case study region

We compiled a dataset consisting of 313 legacy radiocarbon dates. The majority of retrieved dates indicates peat layers of Holocene age (85%), but also Late Glacial and Pleniglacial ages are represented (fig. 2.6a/c). Ages of the Subboreal (37%) and Atlantic (29%) periods are by far most frequent (fig. 2.6b), followed by the Late Glacial (14%) and Subatlantic (10%). Comparison of the reconstructed extent of peatlands (on the current palaeogeographical map series) and the spatial distribution of legacy data points shows that several large areas are underrepresented in the dataset (fig. 2.6a). Precision regarding the locations where the dated samples were collected appeared to be mixed (fig. 2.6e), with most sites only known at field level (52%; error range in order of 100 m). For only 21% the location was retrieved based on registered coordinates (the most detailed location description), while for 27% only the place name of the nearest village was retrieved (error range in order of 1 km).

# 2.4.2 Quality assessment

Based on the values for  $Q_d$  and  $Q_T$ , every date was subsequently assigned to one of four confidence levels (table 2.3, fig. 2.5). For green dates both  $Q_d$  and  $Q_T$ were fairly low, meaning that sufficient information is available regarding dating aspects and taphonomic characteristics. On the opposite side, red dates have high penalty scores for  $Q_d$  and  $Q_T$  indicating that information for these dates is very limited. Orange dates have sufficient information regarding taphonomy but lack detail regarding dating aspects, and vice versa for purple dates.

fig. 2.6b and fig. 2.7 provide an overview of the years when dates were performed, geographic focus through time, and relationship with assessed quality. In the initial stages of radiocarbon dating several studies applied the method for dating peat layers in the study area (fig. 2.7a). During the 1960s and 1970s numbers dropped, followed by a revival during the 1980s when several detailed peat studies were conducted. It appears that in the 1990s less peat dates were performed, however some large studies were published that were (partly) initiated in the 1980s (e.g. Groenendijk, 1997; Van Geel *et al.*, 1998). This relates to certain geographic foci (fig. 2.6b), e.g. the eastern part of the province of Groningen (Groenendijk, 1997), and the Bargerveen (Dupont, 1986) and Fochteloërveen (Van Geel *et al.*, 1998) peat remnants. The majority of retrieved dates were performed in the 2000s, with a main focus on the northern and north-western parts of the study area. Data quality does not show a strong trend over time (fig. 2.7b), indicating that year of dating is not necessarily indicative for quality. However, samples from the 2010s received on average the lowest penalty for taphonomic quality  $(Q_T)$ .

### 2.4.3 Meta-analysis

### 2.4.3.1 Large-scale trends of peat initiation

To deduce spatiotemporal trends in peat initiation, we focused analyses on dates from basal peat layers only (n = 74, see 'lower limit' in fig. 2.6d). The estimated



**Figure 2.6:** Overview of age and location of case study legacy data points (n = 313). (a) Locations of data points binned based on chronostratigraphy, using definitions listed in table 2.4. Uncertainty of locations (see text) not shown for legibility. Do note that several data points overlap (i.e. multiple samples collected at (nearly) the same location). Basal peat date means stratigraphical position is 'lower limit'. Background map shows the reconstructed palaeogeography of the Netherlands for 2,500 cal y BP (also see fig. 2.2c). (b) Location of data points binned per decade when the date was performed. See legend in fig. 2.2b for other map elements. (c) Histogram of calibrated radiocarbon dates. (d) Histogram of chronostratigraphy. (e) Histogram showing precision classes for retrieved locations.



**Figure 2.7:** Overview of years when dates were performed (i.e. between 1950s and 2020). (a) Histogram of data points per 5-year period. (b) Assessed quality  $(Q, Q_d, Q_T)$  averaged per decade.

distribution of these ages is shown in fig. 2.8a to fig. 2.8c, based on green dates with applied filter for aboveground remains of terrestrial macrofossils (n = 12), green dates without filtering applied (n = 50), and dates from all confidence levels combined (n = 74) respectively. The distribution in fig. 2.8a shows a clear bimodal distribution, with peaks at about 14,000 cal y BP (Late Glacial) and 4,500 cal y BP (Subboreal). This trend is still visible in fig. 2.8b, while the largest dataset of fig. 2.8c reveals additional peaks around 11,500 cal y BP (Preboreal) and 6,500 cal y BP (Atlantic). All models show a clear low at 9,500 cal y BP (Boreal). Models of green plus orange confidence level data and green plus purple were also modelled and gave intermediate outcomes (not shown).

Based on the available information, most peat initiation sites appear to result either from primary mire formation or paludification (table 2.5). However, one would expect the number of terrestrialisation sites to be larger, as 19 dates were collected in topographic depressions such as pingos (table 2.6, apparently gyttja was only found/sampled at some of the pingo sites). As the study area has been deglaciated since the penultimate glacial, all land in this region has been exposed for the past 130,000 years. Paludification was therefore probably the dominant peat formation process in the study area.

### 2.4.3.2 Peat initiation trends for different landforms and elevations

We grouped landforms into four categories (table 2.6). Both for green confidence level dates and dates from all confidence levels combined KDE models were constructed (fig. 2.9, showing only models from all confidence levels combined). Too little data were available to model only green confidence level dates that were filtered for aboveground terrestrial macrofossils. The distribution for 'Peatlands (unspecified)' in fig. 2.9c shows two peaks similar to model outcomes in fig. 2.8.



**Figure 2.8:** (Figure on previous page). Outputs of KDE models for basal peat ages in the case study dataset. Results are based on model runs of basal ages with (a) green confidence level that were based on aboveground remains of terrestrial macrofossils (n =12), (b) dates with green confidence level with no further filtering applied (n = 50), (c) all confidence levels combined (n = 74). The dark grey area indicates the sampled KDE estimated distribution. The blue line shows the mean of the KDE distribution, the lighter blue band shows the  $\pm 1\sigma$  range. The red crosses show the central values for the entered dates, the black crosses show the medians of the marginal posterior distributions for every dated event. The calibration curve is indicated for reference (*Reimer* et al., 2020).

For these samples the palaeo-landform underlying the organic deposits was unclear (i.e. could not be retrieved from the date's reference). Samples from 'Plains and ridges' (fig. 2.9d) appear to be of younger age, overlapping only with the second peak in the bimodal distributions of fig. 2.8. The model for 'Topographic depressions' (fig. 2.9e) results in a multi-peak distribution that does not show a clear age trend. Samples from 'Valleys' (fig. 2.9f) result in a wide distribution with two small peaks, covering the Late Glacial and entire Holocene. The model for 'Plains and ridges' was also run with samples of all stratigraphical positions, as a way to validate that the basal ages are always oldest, and that retrieved dates that were indicated to originate from higher stratigraphical positions are indeed younger (fig. 2.9g).

Age-elevation plots were constructed for basal peat samples (fig. 2.10c, n = 73) and for samples from all stratigraphical positions (fig. 2.10d, n = 302). Basal peat samples from topographic depressions mostly date from before 6,000 cal y BP. Valleys are located both at lower and higher positions and have ages crossing the Late Glacial and entire Holocene (fig. 2.9f), with high-lying locations being youngest (fig. 2.10c). Basal dates from plains all date from after 6,000 cal y BP. Basal dates from low-lying plains ( $\leq 0$  m O.D.) fit the Relative Sea Level (RSL) curve for the Wadden Sea (fig. 2.10c, Meijles *et al.*, 2018). Basal peats from the high-lying plains (between 5 and 10 m O.D.) are all younger than 5,000 cal y BP. When plotting data from all stratigraphical positions (not only basal dates, fig. 2.10d), several lines are visible in the data, which represent vertical series of these lines indicates the vertical accumulation speed, which lies between 0.35 - 0.57 mm/y. As elevations from non-basal dates were not corrected for potential compaction issues, these lines indicate only the minimum accumulation speed.

Landform grouping	Landform filter	Green confidence level, n:	All confidence levels, n:			
Peatlands	Peatland (unspecified) <sup>1</sup>	17	17			
(unspecified)	Bog (hummock) <sup>1</sup>					
	Bog (hollow) <sup>1</sup>					
Topographic	Pingo	15	19			
depressions	Depression <sup>2</sup>					
Plains and ridges	Plain	9	9			
	Mound					
	Ridge					
Valleys	Valley	8	24			
	Channel fill					
<sup>1</sup> Dates from studies that did not contain information on underlying landform.						

Table 2.6: Landform groupings, specifying applied landform filter and number of dates with green confidence level and all confidence levels combined (only basal dates).

<sup>2</sup> Topographic lows such as deflations in coversand.

#### 2.5Discussion

Here we first discuss the main findings of the case study, followed by experiences regarding data retrieval, representativity of the resulting dataset, and effect of the quality assessment. Based on this we evaluate the proposed workflow.

#### 2.5.1Case study

#### 2.5.1.1Main findings on peatland development

The legacy dataset indicates peat initiation in the study area from at least the Late Glacial onwards (fig. 2.6). KDE model results show a bimodal distribution of basal peat dates, with a first peak during the Late Glacial, a low in the Boreal period, followed by a rise starting in the Atlantic and finally a peak during the Subboreal (fig. 2.8). The majority of data points is located in the northern half of the study area. Here, several spatial clusters indicate areas with simultaneous peat initiation, e.g. during the Atlantic and Subboreal in the east of the province of Groningen (Groenendijk, 1997).

When considering peat initiation for landform groups, several trends can be distinguished (fig. 2.9). Onset of peat growth took place during the Late Glacial



**Figure 2.9:** Overview of landform data. (a) Assessed quality  $(Q, Q_d, Q_T)$  averaged per landform. (b) Locations of data points binned based on landform grouping (table 2.6). Do note that several data points overlap (i.e. multiple samples collected at (nearly) the same location). (c-f) KDE models of peat initiation per landform grouping (detailed in table 2.6, for interpretation of KDE plots see caption fig. 2.8). Results are from model runs where dates from all confidence levels were included. (g) Comparison of model outcomes for landform type 'Plains and ridges' when only basal dates are included versus dates from all stratigraphical positions.



**Figure 2.10:** (Figure on previous page). Comparison of peat initiation data with  $\delta^{18}O$  and sea level rise curves. In (a) the  $\delta^{18}O$  curve is shown (GICC05 NGRIP  $\delta^{18}O$  data accessed through OxCal). The bimodal distribution of peat initiation dates (including all confidence levels), is shown in (b), see fig. 2.8c for details. In the age-elevation plot in (c) only basal peat dates are included (n = 73, note in (b) n = 74, i.e. for one date no elevation data is available). Data points are coloured by landform. In (d) peat dates from all stratigraphical positions are shown (n = 302), data points are coloured by confidence level. Note that sample elevation in (d) for non-basal dates is only indicative as it was not corrected for potential compaction effects. The RSL curve (data from Meijles et al. (2018) was added to (c) and (d). The data points that were used by Meijles et al. (2018) to generate the RSL curve are not part of our case study dataset.

and entire Holocene in river valleys, whereas it started on plains and ridges only during the Subboreal. In topographic depressions peat initiation was rather erratic through time. For sites with unclear palaeo-landform underlying the organic deposits ('Peatlands (unspecified)'), peat initiation follows the bimodal distribution mentioned earlier. Age-elevation plots show a general trend that peat growth started earliest at the lowest locations, and reaches higher positions later in time (fig. 2.10).

Our analyses point to changes in several boundary conditions that, either alone or in combination, may have led to peat initiation (and lateral expansion). The first peak of the bimodal distribution coincides with the Bølling-Allerød interstadial, and ends with the onset of the Younger Dryas (figs. 2.10a and 2.10b). Comparison of fig. 2.8 with fig. 2.9 suggests that this peak primarily consisted of peat initiation in topographic depressions (fig. 2.9e) and onset of peat growth in river valleys (fig. 2.9f). The rise and maximum of the second peak in the bimodal distribution coincide with strong sea level rise (fig. 2.10b and fig. 2.10c, Meijles et al., 2018) and the hypsithermal (Holocene Thermal Maximum; 9,000 to about 5,000-6,000 years ago; Wanner et al., 2008; Renssen et al., 2009). Given favourable climatic conditions for peat growth, combined with sea level rise and related groundwater level rise, peat deposits increasingly filled (higher located) river valleys (fig. 2.9f and fig. 2.10c) and eventually formed on high-lying plains (fig. 2.9d and fig. 2.10c). The drop of the second peak coincides with neoglacial cooling (5,000-6,000 years ago to pre-industrial time; Wanner et al., 2008), perhaps indicating less favourable climatic conditions. However, as peat covered an increasingly large area, further initiation and expansion may also have become limited due to lack of sites suitable for peat growth.

Casparie and Streefkerk (1992) state that for the Netherlands two main phases of climate-induced mire initiation occurred, from 7,000-6,500 BCE (9,000-8,500 cal y BP, start of Atlantic) and around 5,000 BCE (7,000 cal y BP, middle Atlantic). Both periods fall between the start of the second peak and the 'bump'

prior to its maximum in fig. 2.8c, but the legacy dataset shows no indication for a drop of peat initiation between 8,500 and 7,000 cal y BP. Van Geel *et al.* (1998) advocate that the 2,800 cal y BP event is a cause for peat initiation. Locally peat may have initiated at this timing, however, their sampling location may also have been a site overgrown through lateral expansion of a pre-existing, older peatland. Presence of a main initiation period around 2,800 cal y BP is not supported by the bimodal distribution of the legacy data. Based on detailed palynological investigations in the Bargerveen peat remnant, (indicated in fig. 2.2b), Dupont (1986) concludes that human influences can be traced in arboreal pollen data only from 5,500 cal y BP onwards, which suggests that human impact on peat initiation was probably limited in the study area.

On the Dutch national palaeogeographical map series (table 2.7), peat initiation in the study area starts at the earliest around 7,500 cal y BP, slightly later than the rise of the second peak in the bimodal distribution in fig. 2.8. No peat deposits are present on the maps prior to 7,500 cal y BP, whereas our results indicate that a peat initiation peak during the Late Glacial must have resulted in peat cover prior to this date (mainly in topographic depressions and river valleys, fig. 2.9e and fig. 2.9f). According to the map series, maximum extent of peatlands was reached between 3,250 and 2,500 cal y BP (table 2.7). The basal dates in the legacy dataset are mostly older than this, indicating that the majority of peatlands in the study area indeed formed before 2,500 cal y BP. However, some basal dates show younger ages (fig. 2.8), especially in valleys (fig. 2.9f), indicating that peat initiation (or lateral expansion) continued at least at some sites after 2,500 cal y BP. Non-basal dates show that vertical peat growth continued as well (fig. 2.10d), suggesting that maximum extent and maximum thickness of peat deposits were probably not reached at the same time.

	Zagwijn (1986)	Westerhoff <i>et al.</i> (2003)	Vos <i>et al.</i> (2020); Vos (2015a)
Nr. of maps/timeframes	10	6	13
Peat initiation <sup>1</sup>	$\sim$ 7,500 cal y BP	$\sim$ 6,500 cal y BP	$\sim$ 7,500 cal y BP
Maximum extent <sup>1</sup>	~ 3,250 cal y BP	$\sim$ 2,600 cal y BP	$\sim$ 2,500 cal y BP
<sup>1</sup> For our study area			

**Table 2.7:** Comparison of peatland initiation and expansion in the study area as indicated by three Dutch national palaeogeographical map series.

Based on what could be derived from the legacy data, and considering the surface exposure of the study area for 130,000 years (Ter Wee, 1962), paludification

seems to have been the most prominent process causing peat formation in the study area. Paludification may result from environmental factors but also from autogenic processes leading to lateral expansion of peatlands (see section 2.2.2). For our case study, it is often unclear whether dates stem from the same former peatland, as this would already require a clear view of their palaeogeography. Consequently the dataset is not suitable to draw inferences on local peat initiation versus lateral expansion of existing peatlands.

The legacy dataset leads us to tentatively conclude that the study area witnessed two major phases of peat initiation, where the earliest peak was probably mostly driven by climate whereas the second was the result of climate in combination with Holocene sea level rise. We did not consider presence of impermeable deposits in the study area, these may have further enhanced the potential for peat growth, but the degree to which this contributed and on which spatial scale remains so far unclear.

### 2.5.1.2 Experiences regarding data retrieval

Most scientific publications from which data was collected were fairly easy to find using basic literature searches and keyword queries. Reports from contract-based archaeology were easily accessed, however due to the vast amount of reports available it was generally difficult to find relevant information.

Irrespective of data source, we were able to retrieve the labcode for all samples, thus providing insights in the uncalibrated dating results. In case of ambiguities, dates could be retrieved from the Groningen databases. The bulk SampleType was mainly deduced from labcodes. Details for macrofossil samples were retrieved from publications and lab archives. Overall, we found many more dates than anticipated.

Unfortunately, quite often location and sample elevation were not documented in great detail (fig. 2.6e). For our GIS analyses, the spatial error was considered irrelevant due to the fairly large scale of the study area. However, location was needed to calculate former sample elevation relative to m O.D., as for a large number of samples elevation was only reported relative to the (former) surface. With imprecise location and surface levels changing over time e.g. due to peat compaction and oxidation, these calculations only yield estimations for sample position relative to m O.D..

The stratigraphical position of samples was sometimes reported elaborately, e.g. including cross-sectional profiles. However, for a fairly large number of dates (n = 75 out of 313) we were unable to interpret stratigraphy. These dates indicate that organic deposits were present at this location at the dated age, but further implications are much more difficult to deduce.

### 2.5.1.3 Representativity of the legacy dataset

The meta-analysis of Ruppel *et al.* (2013) indicated a lack of data for the northwest European Plain. The legacy dataset of our case study demonstrates that this image is not entirely valid: our search revealed 74 basal peat dates in the studied region. Additionally, sea level research such as the reconstructed RSL curve for the Wadden Sea (Meijles *et al.*, 2018) is based on elaborate datasets of (legacy) basal peat dates.

However, despite our efforts a limited amount of dates was found in the southern half of the study area. This is probably due to two major factors. As can be deduced from fig. 2.6b and fig. 2.7, research traditions and related concentrations of studied sites create a bias in the dataset as a whole. In addition, large scale peat reclamations of the past have largely determined the distribution of surviving peat remnants and consequently potential sites for field study. While interpreting the data these factors should be kept closely in mind.

To address these biases in the dataset future studies may include (legacy) dates that were not performed on peat deposits directly, but on archaeological artefacts that were retrieved from peat layers or from underneath them. It has for example been demonstrated that the coversand landscape underlying the northern part of the former Bourtangermoor (Dutch-German border area, the surviving remnant on Dutch territory is the Bargerveen, fig. 2.2b) is very rich in Mesolithic sites (Groenendijk, 2003). Such finds provide a terminus post-quem for peat initiation, even though potential hiatuses have to be taken into account. Well-preserved overgrown cultural landscapes are also known from northern Germany (e.g. Pantzer, 1986). Well-dated archaeological finds from peat layers may both provide a terminus ante-quem (for underlying peat layers) and/or terminus post-quem (for overlying layers), depending on the local stratigraphy. As archaeological finds from peatlands were often recovered in the distant past during peat-cutting (Van Beek *et al.*, 2015), they do require a quality assessment of their own, tailored for archaeological aspects in addition to taphonomic ( $Q_T$ ) and dating ( $Q_d$ ) quality.

### 2.5.1.4 Effect of the quality assessment

The quality assessment shows that the data points are dispersed through the four confidence levels, indicating that for some samples taphonomic quality is relatively low whereas for others problems lie in the dating quality (fig. 2.5). A significant part of the data points received a green confidence level (n = 121 out of 313), which allows most detailed filtering options as for many aspects sufficient information is available.

The KDE modelling runs with different confidence level groups (fig. 2.8) lead to distributions that are comparable in overall shape, but vary at a more detailed level.

Green confidence level models (with and without filtering, fig. 2.8a and fig. 2.8b) result in a clear bimodal distribution. When including all data (fig. 2.8c) this trend remains visible but becomes less clearly defined. The use of confidence levels provides insight in this confounding effect caused by dates with low taphonomic or dating quality. This approach can however only be applied if the (sub)dataset is large enough, e.g. for the analysis of landform groups this subdivision was not fully possible.

### 2.5.2 Evaluation of approach

The proposed workflow and quality assessment demonstrate the balancing act to reach robustness without being too strict and consequently discarding the majority of data. All data points contain information, the question is how to extract it adequately. The quality assessment has a flexible set-up and depending on the research questions to be answered, assessment criteria can be in- or excluded or made more impactful using the weights. Subsequent filtering allows tailor-made and informed decisions for data analysis. For instance, if for a certain research question (e.g. reconstructing a sea level curve) it is unnecessary to know a detailed location of the date but crucial to know its elevation and stratigraphical position, weights may be adjusted accordingly which will result in a higher penalty for dates that do not match these criteria.

The case study shows that varying criteria have been used to define peat initiation and to subsequently select samples, resulting in divergent approaches to date the onset of peat accumulation. Consequently this led to a range in taphonomic  $(Q_T)$  and dating  $(Q_d)$  quality in our quality assessment. The methods of the studies from which dates were retrieved partly depend on the research objectives, but also reflect methodological possibilities at the time of dating, for instance use of bulk sampling prior to the development of AMS.

Discussions on methodological aspects of dating and 'best practices' are reflected in the quality criteria. For instance, a bulk sample receives a penalty for *SampleType*, as bulk samples are generally large and consist of an uncharacterised mixture of organic compounds (e.g. Törnqvist *et al.*, 1992, 1998). Inherently, this means a penalty is also assigned for *SpeciesType* and *Aboveground*, as it is unknown which species and which plant tissues are contained within the sample. If for a given peatland a reservoir effect is expected (Kilian *et al.*, 1995; Blaauw *et al.*, 2004), then either weights for these properties can be increased, filtering can be applied (to exclude all samples with unknown and aquatic species), or both.

It is important to note that the penalty score is cumulative, not exclusive. For instance, if it is known whether a sample consisted of macrofossils it will receive

no penalty for the property *SampleType*. However, for a sample that consisted of bulk, the overall penalty score may still be low (and resulting confidence level green) if other properties (with an assigned weight above zero) were well-known and few further penalties were assigned. In case *SampleType* is crucial to answer the research question, either its weight should be increased substantially, or a filter should be applied after the quality assessment to generate a list of dates for instance with green confidence level and only macrofossils as *SampleType*.

It is also important to realise that the stricter the boundaries of the confidence levels are defined, and the more subsequent filtering is applied, the smaller the resulting subset of datapoints will be. This may also result in overrepresentation of samples from a few studies from a specific area (as these have comparable taphonomic and dating quality), which may affect how representative outcomes are for the study area as a whole.

Finally, the quality assessment only makes a difference if the dates actually differ for the selected criteria, otherwise the majority will receive the same penalty. This means that the combination of criteria used (i.e. turned on and off by reducing the weight to zero) is crucial to really distinguish dates based on their quality.

### 2.5.3 Implications and recommendations

Data rescue and reuse lead to improved continuity of data (Gil *et al.*, 2016) and development of new, overarching insights (e.g. Tolonen and Turunen, 1996; Ruppel *et al.*, 2013, this study). Based on the process of data rescue and meta-analysis of the case study it appears that the two largest peat remnants in this area, Fochteloërveen and Bargerveen, have so far only been considered by two studies dating one and two vertical cores respectively (Van Geel *et al.*, 1998; Dupont, 1986). These remnants are the main storage sites of the remaining peat archives and have scientific potential yet to be discovered.

The properties that are recorded and their level of detail always depend on the research question to be answered. Additionally, awareness of what is relevant to report may differ between disciplines. However, based on experiences with data reuse in our case study, we emphasize the importance of recording detailed information on basic properties such as geographical location, elevation, stratigraphical position, and sample details. With peat soils further diminishing in spatial extent but also in thickness, we underline the importance of registering coordinates, and where possible elevation in m O.D.. Without this information options for future peat studies that require field data are further reduced. Additionally, sharing data based on the FAIR principles is key (Gil *et al.*, 2016; Wilkinson *et al.*, 2016), otherwise options for reuse decrease rapidly (Savage and Vickers, 2009).

# 2.6 Conclusion

We developed a workflow for reuse of legacy geochronological data in peatland studies, including rigorous quality assessment. The latter can easily be tailored to specific research questions by adjusting the relative weights assigned to penalized aspects.

The proposed approach was tested on a case study of (former) peatlands in the Netherlands. Peat growth started in the Late Glacial (~14,000 cal y BP), dropped during the Boreal (~9,500 cal y BP) and showed a second peak in the Subboreal (~4,500 cal y BP). Peat initiation occurred in the Late Glacial and throughout the Holocene in river valleys, whereas only during the Subboreal on plains and ridges. We tentatively conclude that the earliest peak was mostly driven by climate (Bølling-Allerød interstadial), whereas the second was probably the result of Holocene sea level rise and related groundwater level rise in combination with climatic conditions (hypsithermal).

Studies that reuse legacy data may yield new insights that require a birds-eye view to be discovered. However, their success depends on data retrieval. We therefore emphasize the importance of FAIR sharing detailed information on basic properties such as geographical location, elevation, stratigraphical position, and sample details. These should be recorded irrespective of research aim, to prevent further data loss from peat archives that are at risk of disappearing.

### Author contributions

Funding was secured by RvB. CQ drafted the outline for the research, and improved it based on input from RvB, YvdV and JW. CQ designed the workflow for reusing legacy data, created the database set-up and draft version of the quality assessment. Data collection was initiated by CQ and to a large extent completed by TH, who also elaborately recorded all data. HvdP provided expertise on radiocarbon dating aspects and insight knowledge on the radiocarbon dates performed in Groningen to check and complement the dataset. All authors provided further input for the quality assessment, for which background mathematics were discussed by CQ and JQ. Approach for the data analysis was chosen by CQ and YvdV. JQ programmed the Python scripts for the automation of the quality assessment and to generate data visualisations. CQ performed the data analyses in OxCal and GIS. CQ wrote the main body of text, and finalised it based on feedback from all authors.

### Acknowledgements

We thank Annemie Kersten and Bert Groenewoudt for help with the literature search, several authors whose work was included in the case study for providing details on radiocarbon samples, and Kim Cohen for the discussion and information about databases for legacy radiocarbon dates. We thank Harm Jan Pierik and an anonymous reviewer for their efforts in reviewing an earlier version of this manuscript and the dataset; their feedback was highly appreciated.

# Chapter 3

Dating basal peat: The geochronology of peat initiation revisited

Cindy Quik, Sanne Palstra, Roy van Beek, Ype van der Velde, Jasper Candel, Marjolein van der Linden, Lucy Kubiak-Martens, Graeme Swindles, Bart Makaske, Jakob Wallinga (2022)

> *Quaternary Geochronology*, 72, 1-22: doi.org/10.1016/j.quageo.2022.101278

Dataset at 4TU.Centre for Research Data: doi.org/10.4121/16923358

### Abstract

Attributing the start of peat growth to an absolute timescale requires dating the bottom of peat deposits overlying mineral sediment, often called the basal *peat.* Peat initiation is reflected in the stratigraphy as a gradual transition from mineral sediment to increasingly organic material, up to where it is called peat. So far, varying criteria have been used to define basal peat, resulting in divergent approaches to date peat initiation. The lack of a universally applicable and quantitative definition, combined with multiple concerns that have been raised previously regarding the radiocarbon dating of peat, may result in apparent ages that are either too old or too young for the timing of peat initiation. Here, we aim to formulate updated recommendations for dating peat initiation. We provide a conceptual framework that supports the use of the organic matter (OM) gradient for a quantitative and reproducible definition of the mineral-to-peat transition (i.e., the stratigraphical range reflecting the timespan of the peat initiation process) and the layer defined as *basal peat* (i.e., the stratigraphical layer that is defined as the bottom of a peat deposit). Selection of dating samples is often challenging due to poor preservation of plant macrofossils in basal peat, and the representativity of humic and humin dates for the age of basal peat is uncertain. We therefore analyse the mineral-to-peat transition based on three highly detailed sequences of radiocarbon dates, including dates of plant macrofossils and the humic and humin fractions obtained from bulk samples. Our case study peatland in the Netherlands currently harbours a bog vegetation, but biostratigraphical analyses show that during peat initiation the vegetation was mesotrophic. Results show that plant macrofossils provide the most accurate age in the mineral-to-peat transition and are therefore recommendable to use for <sup>14</sup>C dating basal peat. If these are unattainable, the humic fraction provides the best alternative and is interpreted as a terminus-ante-quem for peat initiation. The potential large age difference between dates of plant macrofossils and humic or humin dates (up to  $\sim 1700$  years between macrofossil and humic ages, and with even larger differences for humins) suggests that studies reusing existing bulk dates of basal peat should take great care in data interpretation. The potentially long timespan of the peat initiation process (with medians of  $\sim$ 1000,  $\sim$ 1300 and  $\sim$ 1500 years within our case study peatland) demonstrates that choices regarding sampling size and resolution need to be well substantiated. We summarise our findings as a set of recommendations for dating basal peats, and advocate the widespread use of OM determination to obtain a low-cost, quantitative and reproducible definition of *basal peat* that eases intercomparison of studies.

# 3.1 Introduction

The start of peat growth represents a major landscape change. Attributing this transition to an absolute timescale requires dating the bottom of peat deposits overlying mineral sediment, often called the *basal peat*. Robust age control of basal peat layers is of paramount importance, not only for studies aimed directly at peatlands, for instance concerning their palaeogeography, development and carbon sequestration (e.g. Gorham *et al.*, 2007; Yu *et al.*, 2011), but also for interdisciplinary research fields that harness the peat archive for climate and sea level reconstructions (e.g. Törnqvist and Hijma, 2012; Morris *et al.*, 2018) or to contextualize wetland archaeology (e.g. Chapman *et al.*, 2013).

Peat growth results from a positive production-decay balance, i.e. where the decay rate of organic material is slower than the rate of production. The decay rate primarily depends on moisture level, which in turn is influenced by various factors such as climate (e.g. Weckström *et al.*, 2010), changes in hydrological base level (sea level rise, e.g. Berendsen *et al.*, 2007) or regional groundwater changes (e.g. Van Asselen *et al.*, 2017), landforms and surface topography (e.g. Almquist-Jacobson and Foster, 1995; Mäkilä, 1997; Loisel *et al.*, 2013), impermeable deposits or resistant layers in the soil profile (e.g. Breuning-Madsen *et al.*, 2018; Van der Meij *et al.*, 2018), and anthropogenic influence (e.g. Moore, 1975, 1993).

A wetland is an area where the substrate is water-saturated or inundated for a substantial period (Charman, 2002c; Joosten and Clarke, 2002). A minimum depth of 30 cm of peat is required to classify a wetland area as a peatland (Charman, 2002c; Joosten and Clarke, 2002; Rydin and Jeglum, 2013c). This implies that during build-up of the first organic deposits, the area is not yet a peatland according to definition, but rather a wetland where peat formation occurs. As a result of these definitions, one could make a distinction between peat initiation (i.e., build-up of the first organic deposits) and peatland initiation (i.e., referring to the moment when 30 cm of organic deposits has formed in a certain area). In the current paper, we focus on peat initiation (prior to the formation of a peat layer with a thickness of 30 cm or more).

Peat initiation may occur through (a combination of) three processes, briefly outlined below based on Charman (2002c) and Rydin and Jeglum (2013d). Terrestrialisation (also called infilling) refers to the process where peat develops in or at the edge of water bodies. Terrestrialisation is characterized by gyttja deposits at the base, which require a water depth of at least 0.5 m to form (Bos, 2010). Paludification refers to peat formation on previously unsaturated mineral substrate, and thus reflects moistening of the landscape. Primary mire formation involves peat growth on newly exposed waterlogged substrate (e.g. after land

uplift from sea). Here, peat growth starts directly on the fresh parent material. When a peat surface rises above the regional groundwater level, consequent strong dependence on rainwater leads to ombrotrophication, which may result in a fenbog transition (Charman, 2002a; Rydin and Jeglum, 2013d; Loisel and Bunsen, 2020). Peatlands not only grow vertically, but also expand laterally, which results in a larger peat-covered area. Paludification of surrounding soils due to poor drainage at the edge of the peatland, and resulting peatland expansion, is known as an autogenic process (Charman, 2002a). However, allogenic factors such as climate and topography influence the rate and extent of lateral spread (e.g. Korhola, 1994; Ruppel *et al.*, 2013).

Peat initiation can be studied at various scales (fig. 3.1a). At the landscape scale, peat initiation refers to the onset of peat growth at a certain locus, that expands over time to cover a larger area. In this case, the term peat initiation refers to the location and time where the nucleus of the resultant peatland developed. In contrast, at the very local scale, peat initiation may be used to indicate the moment of accumulation of the first peat deposits at a specific location, where the location may reflect a development locus but could also be a site that became covered by peat through lateral expansion of a nearby locus. To distinguish lateral expansion from a development locus, one or multiple transects of basal peat dates are usually required (e.g. Mäkilä, 1997; Mäkilä and Moisanen, 2007; Chapman *et al.*, 2013). The approach for dating peat initiation at both scales is similar, while the research aim determines which scale level is of interest.

Peat initiation is a process that takes place during a certain timespan, which is reflected in the stratigraphy as a gradual transition from mineral sediment to increasingly organic material (fig. 3.1b), up to where it is called peat (depending on definitions used). We propose it would be most accurate to speak of a period of peat initiation, which requires a series of vertical dates that encloses the gradual stratigraphical boundary. However, a single date of *basal peat* is often used to reflect peat initiation, potentially for reasons of practicality or feasibility when there is need to date many sites. This requires however an unambiguous and explicit definition of *basal peat*.

So far, varying criteria have been used to define *basal peat*, resulting in divergent approaches to date the onset of peat accumulation (Quik *et al.*, 2021). Current approaches, which are partly dependent on the research objectives, vary from visual determinations and basic laboratory analyses such as loss-on-ignition (e.g. Edvardsson *et al.*, 2014) to detailed micromorphological analyses (e.g. Cubizolle *et al.*, 2007) and studies of plant macrofossils (e.g. Loisel *et al.*, 2013). Depending on the approach taken, the accuracy of resulting dates to represent peat initiation may be called into question, as the (possibly site-specific) definition of the *basal peat* often remains implicit.



#### (a) research scope: peat initiation at local versus landscape scale

#### (b) approach: defining peat initiation



#### (c) method: dating peat initiation

#### challenges in sample selection



Figure 3.1: Conceptual framework for dating peat initiation. (a) Schematic top-view of a landscape (peat is indicated in brown), showing the meaning of peat initiation at both the landscape and local scales. (b) Schematic cores showing [i] stratigraphy that results from peat initiation, [ii] simplified interpretation of this stratigraphy, and [iii] the approach we propose in this study. Here, we propose using the organic matter gradient to characterise the mineral-to-peat transition and to define basal peat. To qualify the material as peat, the OM content should be above a certain value denoted with  $M_d$ , where the first cm of material that has an OM content equal to or above  $M_d$  is defined as the basal peat. (c) Schematic core showing challenges with sample selection, and datable fractions and potential contaminants for <sup>14</sup>C dating basal peat.

The international soil classification of the World Reference Base for Soil Resources states that 'organic material' has  $\geq 20\%$  soil organic carbon in the fine earth fraction (by mass) (IUSS Working Group WRB, 2015). A practical challenge of this definition is that determination of soil organic carbon requires expensive analyses (e.g. elemental analysis), whereas soil organic matter (which includes both organic carbon and, if present, inorganic carbon such as carbonates) can be measured with an inexpensive, simple protocol (loss-on-ignition).

To the best of our knowledge, both the mineral-to-peat transition and the layer called *basal peat* (the stratigraphical level used to reflect peat initiation), are not universally defined based on organic matter (OM) content. A universal definition eases inter-site comparisons, but a site-specific definition may be preferable to cover regional differences. Both require that the properties to define basal peat are explicit and reproducible. For example, Cubizolle et al. (2007) use 30% OM as lower limit for peat in the French Massif Central, whereas for instance Loisel et al. (2013) use 50% OM for an Alaskan peatland. To stimulate the use of quantitative and reproducible definitions, a property such as OM content is recommendable, as it can be measured relatively easily and at low cost. As there is a clear gradient in organic matter (indicated by the variable M) at the transition from mineral sediment to peat (i.e., as a function of distance x upwards in the profile), both the mineral-to-peat transition (i.e., the period of peat initiation) and the basal peat (i.e., the stratigraphical layer that is defined as the bottom of a peat deposit) can be defined by the organic matter content [M(x)] and the organic matter gradient [the derivative given by  $\frac{dM}{dx}$ ] (fig. 3.1b).

After defining *basal peat*, adequate sampling and sample pre-treatment for radiocarbon dating are required to accurately derive the age of the basal peat layer. The discussion on which samples most accurately indicate the age of peat layers started several decades ago (e.g. Törnqvist *et al.*, 1992, 1998; Shore *et al.*, 1995; Nilsson *et al.*, 2001; Brock *et al.*, 2011; Van der Plicht *et al.*, 2019). Various studies have highlighted multiple concerns with the radiocarbon dating of peat. If carbon from other carbon sources is incorporated or mixed with the original sample material and cannot be removed (by manual selection and/or chemically), this may result in apparent ages that are either too old or too young for the peat layer of interest (table 3.1).
Processes when the organism was	Processes after the organism died
Apparent older age	Apparent younger age
<ul> <li>Circumstances where organisms incorporate carbon from a reservoir that is not in equilibrium with the atmosphere (so-called reservoir effect), causing apparent ages that are too old:</li> <li>This typically applies to aquatic samples from marine or freshwater circumstances (the latter is also known as hardwater effect, e.g. Törnqvist <i>et al.</i>, 1992; Philippsen, 2013). Relevance of a reservoir effect for peat samples has been postulated (Kilian <i>et al.</i>, 1995; Blaauw <i>et al.</i>, 2004).</li> <li>Incorporation of older carbon or <sup>14</sup>C-depleted carbon may also occur through:</li> </ul>	<ul> <li>Mixing with younger carbon through:</li> <li>Downgrowth of roots (Törnqvist et al., 1992).</li> <li>Translocation of mobile humic acids downwards in a profile, followed by chemical break-down to humins (Palstra et al., 2021).</li> <li>Contamination during sample storage by microbial growth (Wohlfarth et al., 1998) or laboratory pretreatments. Small samples or samples with low carbon content are particularly sensitive to contamination (Van der Plicht et al., 2019), especially if samples are older than 20 ka.</li> </ul>
• Decomposition of underlying peat layers and subsequent assimilation of $CO_2$ (Smolders <i>et al.</i> , 2001).	
<ul> <li>Assimilation of CH<sub>4</sub> originating from bacterial methanogenesis (Van der Plicht <i>et al.</i>, 2019).</li> </ul>	

**Table 3.1:** Concerns with  ${}^{14}C$  dating of peat samples. 'Organism' refers to a plant of peat-forming vegetation.

Bulk dating is often complicated by difficulties with interpreting the resulting age, which represents a mixture of ages of various organic fractions. The development of AMS in radiocarbon has enabled new possibilities for dating peat deposits due to much lower requirements regarding sample sizes (e.g. Tuniz *et al.*, 1998; Jull and Burr, 2015). The concerns outlined in table 3.1 have led to the recommendation to date short-lived, aboveground plant macrofossils of terrestrial species with AMS (e.g. Piotrowska *et al.*, 2011).

Unfortunately, AMS dating of terrestrial macrofossils is not always possible for mineral-to-peat transitions (fig. 3.1c). Depending on the type of mire and local circumstances, the basal peat layer may have an amorphous peat facies largely devoid of (identifiable) plant macrofossils. Limited presence of plant macrofossils may require resorting to bulk sampling for radiocarbon dating the basal peat layer, which could hamper interpretation of dating results. The magnitude of

this problem appears to vary. For instance, Törnqvist *et al.* (1992) found age differences of up to 600 <sup>14</sup>C year between bulk and macrofossil samples from mid-latitude minerotrophic peats, whereas Berendsen *et al.* (2007), who studied comparable deposits, and Holmquist *et al.* (2016), who looked at basal peat from circum-arctic peatlands, reported no significant age differences. Depending on the duration of the period in which the basal peat forms, sampling plant macrofossils from higher positions in the peat profile to circumvent bulk sampling at the base may no longer reflect peat initiation, as a potentially large age difference between these layers might exist.

Pre-treatment of bulk samples using acids and base solutions results in multiple organic fractions which are defined based on their solubilities (Brock et al., 2011; Van der Plicht et al., 2019): fulvic acids are soluble in both alkaline and acids, humic acids are soluble in alkaline but insoluble in acids, and humins are insoluble in alkaline and acids. Fulvic acids are usually removed during pre-treatment and not used for dating. Because of their high mobility, which allows them to translocate easily through a soil profile, significantly younger dates might be obtained for fulvics than for the humic and humin fraction from the same layer (Shore et al., 1995). As the solubility of humics is determined by pH and lower under acidic conditions (Wüst et al., 2008), their mobility depends on environmental circumstances and may change through time. There is no clear consensus in literature on whether humic or humin dates are most representative for dating peat layers. Examples range from studies where no significant age differences are reported (e.g. Cook et al., 1998; Waller et al., 2006), studies that consider humins to be most appropriate for peat (e.g. Hammond *et al.*, 1991; Van der Plicht *et al.*, 2019) and humics for deposits with low carbon amounts (Van der Plicht et al., 2019), or where a conclusion that one or the other fraction is more reliable could not be drawn unquestionably (Brock et al., 2011). Moreover, hardly any studies focus on *basal peat* layers while investigating the ages of these organic fractions (with the exception of Brock et al., 2011). It is therefore unknown which carbon fractions of these basal peat layers, which might be slightly different in organic carbon composition (especially in carbon content) compared to peat samples from higher positions in peat profiles, are most representative for the time period of peat initiation.

In the current study we aim to formulate recommendations for dating *basal peat*. Issues that we specifically address are (1) peat initiation is a process of a certain timespan rather than an event, (2) *basal peat* needs to be clearly defined, (3) selection of dating samples is typically challenging due to potential poor preservation of plant macrofossils in *basal peat* and (4) the representativity of humic and humin dates for the age of *basal peat* is questionable. We analyse lithological, biostratigraphical and geochronological characteristics of the mineral-

to-peat transition in a bog remnant, focusing on understanding the course of the process of peat initiation and related implications for dating.

# 3.2 Methods

#### 3.2.1 Selection of study area and overview of methods

The Fochteloërveen peat remnant (the Netherlands, fig. 3.2) was selected as case study region. This peatland, with its surface area of approximately 2,500 hectares, is one of the largest raised bog remnants of Northwest Europe. The area is considered representative for many (non-coastal) peatlands of the Northwest European Plain with regard to the widespread distribution of its mineral substrate and characteristic climatic conditions (see section 3.2.2). As many European peatlands are subject to ongoing excavation or affected by reclamation relics from historical peat-cutting, basal peat layers may be damaged. In the Fochteloërveen, several former peat cutting pits (some of which are currently artificial lakes) are present and superficial patterns of historical buckwheat fire culture can be recognized. However, the latter disturbed only the surface of the peatland and therefore basal peat layers are undamaged in the majority of the area. A recent study on peat initiation trends in the northern Dutch coversand landscape (Quik et al., 2021) provides background information on peat growth in the wider study region, and demonstrated that the Fochteloërveen (as one of the few surviving bog remnants) has so far received limited scientific attention regarding its initiation and age.

Three sites (named S17, S18 and S20, fig. 3.2a and 3.2b) in the Fochteloërveen peat remnant were selected for dating (for further details on choices for site selection see section 3.2.3). At each of these sites a core containing the mineral-to-peat transition was obtained. Note that the meaning of *basal peat* in this study is much broader than the 'Basisveen Bed' as known in Dutch stratigraphy (TNO – Geological Survey of the Netherlands, 2021a). For each of the three sites, selected levels in the peat core were analysed for percentage organic matter (OM) through loss-on-ignition (LOI), plant macrofossils (PM) and testate amoebae (TA). The palaeo-environmental setting was reconstructed based on analyses of PM and TA. Based on LOI data, samples for radiocarbon (<sup>14</sup>C) dating were selected from multiple levels within each core. When attainable, (charred) plant macrofossils were selected for dating. Additionally the original bulk material was sampled and chemically processed to derive humic and humin fractions for dating. All steps are explained below.

# 3.2.2 Study Area

The northern Netherlands was covered by a continental ice sheet during the Saalian (MIS 6), which led to deposition of glacial till (Rappol, 1987; Van den Berg and Beets, 1987; Rappol et al., 1989; TNO – Geological Survey of the Netherlands, 2021b) on the Drenthe Plateau or till plateau (Ter Wee, 1972; Bosch, 1990). During the Weichselian (OIS 4-2), aeolian cover sands were deposited over an extensive area of Northwest Europe, forming the European Sand Belt (Koster, 1988, 2005). On the Drenthe Plateau these cover sands occur with a thickness varying from 0.5 to 2 m (Ter Wee, 1979; TNO – Geological Survey of the Netherlands, 2021c). The Fochteloërveen peat remnant is located near the western edge of the Drenthe Plateau and is part of three catchments, draining into the rivers Drentsche Aa, Peizerdiep and Tjonger (fig. 3.2c). Historical data of the 18<sup>th</sup> century indicate that peat thickness at the Fochteloërveen has diminished over the past centuries with as much as 7 m at some locations (Douwes and Straathof, 2019). Our corings (see section 3.2.3) demonstrated that peat thickness currently varies from 20 cm to 225 cm at approximately 100 visited locations distributed over the peatland. Current climate is characterised by average temperatures of 2.8° C in January and 17.5° C in July, average annual rainfall of 805 mm, and a potential evapotranspiration of 566 mm (KNMI, 2021).

As a result of large-scale historical peatland reclamations (e.g. Gerding, 1995; Van Beek *et al.*, 2015) currently only small remnants of the former extensive Northwest European peat landscapes remain (fig. 3.2d). The Fochteloërveen remnant is protected as Natura 2000 area and harbours a wide range of plant and animal species (Provincie Drenthe, 2016). Main threats to the quality and continuity of the area include atmospheric nitrogen deposition and desiccation due to intense drainage for surrounding agriculture. Since the 1980s nature conservation is aimed at peatland restoration (Altenburg *et al.*, 2017).

## 3.2.3 Site selection and stratigraphy

We performed an elaborate field exploration of the peat remnant consisting of around 100 corings (some grouped in transects of 185 to 575 m long). For each core the stratigraphy was described (see table 3.2 for details).

As the basal peat is not oligotrophic (see 3.3 Results for further information), field determination of the degree of humification using the scale for ombrotrophic peat by Von Post (Aaby, 1986) does not fully apply. Additionally, the use of Munsell colour charts for fresh organic deposits is often difficult as the material changes in colour following exposure to oxygen. Instead, we applied asimplified version of the organic-facies determination key by Bos *et al.* (2012), which is originally intended for organic sediments in deltaic settings. Our basic field classi-



**Figure 3.2:** (a) Location of the Netherlands and the Fochteloërveen peat remnant in Europe. (b) Topographical map of Fochteloërveen, indicating sampling locations. Dataframe coordinates are in metres (Dutch RD-new [Rijksdriehoeksstelsel] projection). (c) Digital Elevation Model (DEM) of Fochteloërveen and surroundings, showing the main drainage pattern. Elevation is in metres relative to Dutch Ordnance Datum (O.D., roughly mean sea level). (d) Reconstructed palaeogeography for ~2,500 cal years BP, indicating assumed former extent of the peatland area around Fochteloërveen. Sources: topography (OpenSimpleTopo, 3200 pixels/km) by Van Aalst (2021); DEM of the Netherlands (AHN3; horizontal resolution 5 m, vertical resolution 0.1 m) from AHN (2021a,b); rivers from Ministerie van Verkeer en Waterstaat (2007); Natura 2000 area from Ministerie van Economische Zaken - Directie Natuur & Biodiversiteit (2018), palaeogeographical map (500 BC) from Vos and Vries (2013); Vos et al. (2020)

fication differentiates amorphous organic material (similar to *amorphous organics* in Bos *et al.*, 2012), and non-decomposed peat (similar to *oligotrophic peat* in Bos *et al.*, 2012) where further botanical specification is obtained later through microscopic analyses of plant macrofossils.

Following the field exploration, 21 cores originating from sites distributed over the peatland were collected for future analyses. To address the current research aim, three cores were selected based on a set of criteria considering lithological representativity, spatial distribution and elevation (see table 3.3 for a comparison of these properties and further details on selection criteria). Each core was collected from a transect along a cover sand ridge that underlies the peat deposits. Site S17 (transect in fig. 3.3) is located in what is probably a valley or topographic low in the sand landscape underlying the peat deposits, cores S20 and S18 (transects not shown) are located at peat-covered flanks of sand ridges.

**Table 3.2:** Lithology and lithogenetic interpretation of the stratigraphical layers occurring from the surface downwards (modified from Bos et al., 2012).

Lithology	Symbol	Lithogenetic interpretation
Peat with brown colouring and clearly recognisable plant re- mains.	V3	Non-decomposed peat
Peat with blackish-brown colour- ing, greasy consistency and very few recognisable plant remains.	V3*	Amorphous peat (highly humified; <i>sapric</i> cf. IUSS Working Group WRB, 2015)
Mixture of peat with very fine to moderately fine sand, dark brown colouring.	ZV/VZ	Peaty sand/sandy peat (gradual tran- sition zone from Pleistocene min- eral deposits to overlying organic de- posits)
Very fine to moderately fine sand with colour varying from dark brown to light grey, locally loamy, sporadically containing pebbles (ø 1-5 mm).	Ρ	Pleistocene mineral deposits

**Table 3.3:** (a) Criteria for the selection of sites for a vertical dating series, with their respective rationale. (b) Properties of the three sampled sites compared with the minimum (Min), maximum (Max) and average (Avg) of the in total 21 sampled sites of the peat remnant (i.e., where a core for analyses was collected). The surface elevation was measured with a vertical precision of ~10 mm. The total thickness of organic deposits is the sum of V3, V3\* and ZV/VZ. The top of the Pleistocene mineral deposits was derived from the surface elevation minus the total thickness of organic deposits as determined visually in the field (i.e., might deviate slightly from the basal peat layer that was defined later based on OM%). NA = not applicable. For other stratigraphical abbreviations see explanation in table 3.2.

(a) Selection criteria	Ration	ale										
The site is part of a coring transect (distance between 185 and 575 m long).	The coring transect provides relevant background in- formation regarding the landscape position of the site											
The obtained core contains the (vi- sual) mineral-to-peat transition.	Cores of be mos	Cores containing the mineral-to-peat transition will be most straightforward to analyse and are not com- promised by suboptimal sampling conditions										
For the three selected sites, the cores have V3* layers of varying thickness, and at least one site contains a ZV layer.	Analyse ness ra cover t the stu other re recomn	es of thre nge of V3 he stratig dy area ( egions. TI nendation	e sites wi 3* and pr raphical c (see table his ensure s will have	th a rep esence o liversity 3.3b) a s that ar e a wide	oresentative of a ZV I that is preand poter by methome applicab	ve thick- ayer will resent in ntially in dological ility.						
The three selected sample sites are well-distributed spatially and of varying elevation.	Our approach should be applicable to different land- scape positions.											
(b) Property	S17	S18	S20	Min	Max	Avg						
Surface elevation (m. O.D.)	9.308	11.999	10.781	9.31	12.00	10.70						
Total thickness organic deposits (cm)	216	30	35	30	216	94						
Thickness V3 (cm)	182	16	24	16	182	74						
Thickness V3* (cm)	34	6	11	6	34	17						
Thickness ZV/VZ (cm)	0	8	0	0	8	3						
Top of Pleistocene mineral deposits (m. O.D.)	7.15	11.70	10.26 <sup>1</sup>	7.15	11.70	9.70						
Location	East	West	Central	NA	NA	NA						



**Figure 3.3:** Example cross section, showing stratigraphical context of core S17 (on the right). The red box indicates the sampled reach of the profile. About 1 m further in S-E direction, a wide ditch hampered additional corings to extend the transect.

### 3.2.4 Collection of cores

The three cores were collected in 2019 with a hand-operated stainless-steel peat corer (Russian type) of 50 cm long and 60 mm outer diameter, with an equivalent core volume of 0.5 dm<sup>3</sup> (Eijkelkamp Soil & Water, 2018). This type of corer was found to be most useful for sampling both peat and water saturated mineral sediments underneath in one core, with minimal disturbance and low risk of contamination. Other types of corers were considered unsuitable for this purpose. For instance, augers or gouge corers often disturb the sample and do not protect it from contamination as there is no closed coring chamber. A Van der Staay suction corer (Wallinga and Van der Staay, 1999), which is used to sample water saturated mineral deposits, cannot sample peat layers as these block the suction mechanism. The Russian corer is generally used for sampling deeper (i.e. mostly catotelm) peat layers (Vleeschouwer et al., 2010). Field testing demonstrated that this corer was able to simultaneously sample both peat and the top of water saturated mineral deposits adequately (fig. 3.4). In areas where the mineral deposits are compacted so firmly that hand-operated corers cannot be pressed down to a sufficient depth, the use of percussion drilling equipment might be useful (e.g. Eijkelkamp Agrisearch Equipment, 2022).

Each time prior to sampling a new core, the coring chamber and pivoting blade (fin or lid) were cleaned with a fresh microfiber cloth, followed by a thorough rinse with deionised water. Directly after this cleansing routine the corer was pressed down with the pivoting blade closed. After reaching the desired sampling depth the corer was turned 180° clockwise to collect the sample, upon which the pivoting blade closed the coring chamber and the corer was retrieved. Subsequently the corer was kept horizontally with the pivoting blade facing upwards. The blade was carefully turned to retrieve the (undisturbed) core. At this point the core was photographed. Retrieval and packaging of the core followed the procedures proposed by Vleeschouwer et al. (2010) and Givelet et al. (2004), and proceeded as follows. The core was covered with plastic cling film. Then a PVC half-circular pipe was placed over the core, upon which the corer was rotated to transfer the plastic-covered core to the PVC pipe. The exposed side was covered with the remaining plastic film and the core was secured with plastic tape. Through these steps, which took about 5 minutes from retrieval to packaging, handling of the core in the field was minimized. The name, top and bottom of the core were marked with water-resistant labels. The PVC pipes were transported in horizontal orientation to prevent damage to the cores and stored in a refrigerator of 3° C within 12 hours. Location and elevation of all sampling sites were recorded with a Topcon 250 Global Navigation Satellite System (GNSS) receiver, with a horizontal precision of  $\sim 5$  mm and vertical precision of  $\sim 10$  mm (RTK; TOPCON, 2017).



**Figure 3.4:** Photograph of core S17 taken directly after collecting the core in the field, (a) overview photo (bottom at the right), (b) detail of the mineral-to-peat transition.

### 3.2.5 Core processing and subsampling

The cores were opened in the laboratory of BIAX Consult (Zaandam, the Netherlands). Each core was photographed again, the stratigraphy was described and based on visual inspection the approximate mineral-to-peat transition was determined. Around this transition, a range of contiguous 1-cm-thick slices was cut from the core. Outer edges of each slice were carefully cleaned to prevent contamination. Total sample volume for each cleaned level amounted to ~7 cm<sup>3</sup>. The samples were subsampled for multi-proxy analysis based on a priority flowchart (fig. 3.5), which is further explained below.

A bulk subsample (2 cm<sup>3</sup>) was collected for dating. The remaining material (around 5 cm<sup>3</sup>) was used for PM to analyse plant species and to select suitable plant macrofossils for radiocarbon dating. The PM subsamples were obtained from the filtrate after gently rinsing with warm water over a 0.25 mm sieve. A pollen sample (0.5 ml) was collected from the sieving water for future study and stored at 3° C. Sieving water (0.5 ml) and any remaining non-sieved material (ranging between 1 up to 3 cm<sup>3</sup> depending on how much material remained) was collected for TA analysis. Considering its destructive protocol, LOI was performed only on non-sieved material that remained after TA analysis (typically between 1 and 2 cm<sup>3</sup>). To gain more insight in the organic matter gradient in the cores, additional levels were sampled where only LOI was performed (without biostratigraphical analyses and <sup>14</sup>C dating). For these levels 2 cm<sup>3</sup> of unsieved material was used for LOI, the remainder was stored at 3° C for future reference. Table 3.4 provides an overview of all collected subsamples for the three cores.



**Figure 3.5:** Flowchart showing allocation of sample material from each 1-cm-thick core slice, subsequent processing steps and resulting fractions for <sup>14</sup>C dating. PM = plant macrofossil analysis, TA = testate amoebae analysis, LOI = loss-on-ignition.

**Table 3.4:** Overview of analyses and resulting fractions for <sup>14</sup>C dating for each investigated level of the cores S17, S18 and S20. PM = plant macrofossil analysis, TA = testate amoebae analysis, LOI = loss-on-ignition.

Core	From (m O D )	To (m () D )	РМ	ТА	LOI	<sup>14</sup> C bulk	<sup>14</sup> C bulk humin	<sup>14</sup> C macro- fossils
\$17	7 /05	7 505	x	x		X	X	X
517	7.495	7.305	X	x	x	X	×	X
	7 475	7 485	x	X		x	x	X
	7 430	7 440	X		x			
	7.390	7.400	X		X			
	7.350	7.360			Х			
	7.300	7.310	Х	Х	Х	Х	Х	Х
	7.290	7.300	Х	Х	Х	Х	Х	Х
	7.280	7.290	Х	Х		Х	Х	Х
	7.230	7.240			Х			
	7.220	7.230			Х			
	7.210	7.220	Х	Х	Х	Х	Х	Х
	7.200	7.210	Х	Х	Х	Х	Х	Х
	7.190	7.200	Х	Х		Х	Х	Х
	7.180	7.190	Х	Х	Х		Х	
	7.170	7.180	Х	Х	Х	Х	Х	
S18	11.80	11.81	Х	Х	Х	Х	Х	
	11.79	11.80	Х	Х	Х	Х	Х	Х
	11.78	11.79	Х	Х	Х	Х	Х	Х
	11.77	11.78	Х	Х		Х	Х	Х
	11.76	11.77			Х			
	11.75	11.76			Х			
	11.74	11.75			Х			
	11.73	11.74	Х	Х		X	X	X
	11.72	11.73	Х	Х	Х	X	X	X
	11.71	11.72	X	X	X	X	X	
	11.70	11.71	X	X		X	X	
	11.69	11.70	v	V	X	~~~		
	11.68	11.69	X	X		X	X	
	11.07	11.08				×	×	
	11.00	11.07		- <del></del>	^	~	~	
	11.05	11.00	^	^		^	^	
S20	10.40	10.41			Х			
	10.39	10.40			X			
	10.38	10.39	X	X	X	X	X	
	10.37	10.38	X	X	X	X	X	X
	10.30	10.37	X	X	X		X	<b>X</b>
	10.35	10.30	× .	~	X		*	
	10.34	10.35	v	~	$\sim$	~	· · · · · · · · · · · · · · · · · · ·	~
	10.33	10.34	- <del></del>	$\sim$		$\sim$	$\sim$	×
	10.32	10.33	Ŷ	Ŷ	^	Ŷ	×	×
	10.31	10.32	Ŷ	X	×	Ŷ	Ŷ	^
	10.30	10.31	Ŷ	Ŷ	^	Ŷ	. <u>^</u>	
	10.23	10.30	^	~	x	~	~	
	10.20	10.25			x			
	10.26	10.20	X	X		X	X	X
	10.25	10.26	X	x	X	X	X	X
	10.24	10.25	Х	Х	Х	Х	Х	х

## 3.2.6 (Bio)stratigraphical analyses

The percentage organic matter was determined using LOI (see e.g. Chambers *et al.*, 2011; Kennedy and Woods, 2013). The used subsamples for LOI had a volume of 1 to 2 cm<sup>3</sup>. Sample dry weight was determined after drying for 24 hours at 105° C, followed by combustion at 550° C. Subsample dry weight was ~0.65 g on average, of which the remaining mineral component (dry ash) after combustion amounted to ~0.29 g. A microbalance (0.0001 g) was used to maximise measurement precision for these small subsample sizes.

PM analyses were conducted at BIAX Consult in Zaandam, the Netherlands. Mosses and seed remains of vascular plants in the filtrate were identified with a Leica binocular incident light microscope at magnifications of x6 to x50. Identifications followed Körber-Grohne (1964); Körber-Grohne (1991), Berggren (1969, 1981), Anderberg (1994), Smith (2004), and Cappers *et al.* (2006).

TA subsamples were analysed at Queen's University Belfast, United Kingdom. These subsamples were wet-sieved at 300  $\mu$ m and back-sieved at 15  $\mu$ m following standard procedures described by Booth *et al.* (2010). Two slides per sample (2 x 21 x 21 mm cover glasses) were studied using a high power binocular microscope under x20 to x40 magnification.

## 3.2.7 Defining basal peat (determining $M_d$ value)

To gain a thorough understanding of the OM gradient around the mineral-topeat transition in our study area prior to selecting a defining OM value (i.e.,  $M_d$ , fig. 3.1b) above which the material is called *peat*, a vertical series of LOI measurements was performed for five duplicate cores (i.e., additional cores that were collected approximately 10 cm next to the locations of the 21 sampling sites). In this way, the OM gradient could be determined for a continuous sequence (of 22 to 26 cm long) with a resolution of 1 cm (note that this was not fully possible for the cores of sites S17, S18 and S20 that were collected for dating, as for some investigated levels no unsieved material remained after completion of the biostratigraphical analyses to perform LOI). The resulting OM gradients were analysed to derive a substantiated value for  $M_d$ .

## 3.2.8 Radiocarbon dating

The material reserved for bulk sampling was used without removal of roots and provided sufficient humics and humins to date for nearly each level (table 3.4). For plant macrofossil samples, only charred aboveground plant material was selected for radiocarbon dating; waterlogged (uncharred) belowground plant remains like rootlets, radicelles and rhizomes were present abundantly, but aboveground wa-

terlogged plant remains were scarce. From 22 levels of the three cores, charred above ground plant remains from terrestrial plants could be retrieved (tables 3.4 and 3.5).

Radiocarbon measurements were performed at the Centre for Isotope Research of the University of Groningen (the Netherlands), using a MICADAS Accelerator Mass Spectrometer (Ionplus AG; Synal *et al.*, 2007). For background information on the principles of radiocarbon dating we refer to e.g. Bayliss *et al.* (2004); Ramsey (2008b); and Törnqvist and Hijma (2012). For a full description of the (pre-treatment) methods in Groningen we refer to Dee *et al.* (2020). Here we only concisely describe details of chemical pre-treatment and other relevant characteristics of our dating samples.

The acid-base-acid (ABA) method was applied to all the charred plant remains, with respective temperatures of 80°, 80° and 20° C. For the humin and humic fractions the bulk sample material was first pre-treated with acid and base, both at 80° C. Then the base solution was kept separate and the humic fraction was obtained by addition of acid (at 20° C). The solid humic fraction was rinsed with decarbonized water to almost neutral pH and dried in an oven at 80° C. The solid material (humin fraction) that remained after the base step was rinsed to neutral pH and then treated with acid (at 20° C), rinsed with decarbonized water to neutral pH and dried in an oven at 80° C. The solid material was not sieved during the entire procedure, to secure that the very small organic particles in the bulk material were retained. Instead, a centrifuge was used to separate the solid and liquid fractions. Some of the humin fraction samples contained a lot of sand and little organic remains, which resulted in a very low carbon yield of the combusted subsample (tables 3.7-3.9). Also the obtained humic fraction yields were very low for several samples.

After chemical pre-treatment (sub)samples were weighed in tin capsules and combusted to  $CO_2$  in an elemental analyser (IsotubeCube NCS). This analyser is coupled to an Isotope Ratio Mass Spectrometer (Isoprime 100) for measurement of  $\delta^{13}C$  in the sample material. Resultant  $CO_2$  was graphitized to carbon using hydrogen and an iron catalyst. The graphite was pressed into aluminium cathodes and measured on  ${}^{12}C$ ,  ${}^{13}C$  and  ${}^{14}C$  atoms with the MICADAS. The samples measured as graphite in the AMS, can be divided in two groups. Part of the humic and humin fractions were relatively small (up to 1 mg carbon) and these samples were measured in an AMS batch for small-sized samples. The measurement error for these samples is around  $\pm 40$  yrBP. The other part of the samples was measured as graphite in a regular AMS batch (for masses >1 mg and <2.5 mg C) and the measurement uncertainty for these samples is in general below  $\pm 30$  yrBP.

Three charred plant remains samples of S17 (M6, M12 and M3) and one humic fraction of S18 (M12) had very small sizes (< 0.5 mg) and were treated

in a different way. These samples were combusted with an elemental analyser (Isotube Cube) to  $CO_2$ . The  $CO_2$  was led into the AMS and measured directly on carbon isotopes. Since much lower carbon masses are measured in the AMS (in a much shorter time period) when introduced as  $CO_2$  gas compared to graphite samples, the measurement uncertainty for these samples is larger ( $\pm$  60-80 yrBP) compared to the samples measured as graphite.

The <sup>14</sup>C measurement results (F<sup>14</sup>C and <sup>14</sup>C age in tables 3.7 to 3.9) are calculated according to the conventions (Stuiver and Polach, 1977), OX-II (SRM 4990C) was used as calibration standard, and the results are corrected for background signals using background reference materials and for isotopic fractionation using the  $\delta^{13}$ C value measured with AMS.

## 3.2.9 Calibration and age-depth modelling

The radiocarbon dates were calibrated using IntCal20 (Reimer *et al.*, 2020) in the OxCal program (version 4.4; Ramsey, 1995). The calibrated ages are presented as likelihoods in age-depth plots in OxCal (i.e., initially no further assumptions were applied in a Bayesian modelling framework). Subsequent modelling was based on the following assumptions:

- Plant macrofossil ages provide the best estimate of the age of a peat layer (based on correct chronology as shown in fig. 3.7a/e/i and in line with the consensus in literature to date short-lived, aboveground plant macrofossils of terrestrial species with AMS, see e.g. Piotrowska *et al.*, 2011.
- For reasons outlined in fig. 3.1, humic and humin samples may potentially yield different ages than plant macrofossil samples.
- Agreement of humic/humin ages with plant macrofossil ages indicates that the humic/humin samples are contemporaneous with the peat layer from which they were obtained. As such, they provide a representative indication of the age of the peat layer.
- In contrast, disagreement of humic/humin ages with plant macrofossil ages indicates that the humic/humin samples are not contemporaneous with the peat layer from which they were obtained, and do not accurately represent the age of the peat layer.

For each core, the dates of plant macrofossils were modelled using the P\_Sequence function (Ramsey, 2008a) as follows. Start and end of the P\_Sequence were defined with a Tau\_Boundary and regular Boundary respectively. The levels of the start of the peat initiation process, basal peat layer and end of the peat

initiation process (i.e., as based on %OM data, see fig. 3.7) were specified in the P\_Sequence. If no macrofossil date for these levels was available, the Date command was added as query to generate an age distribution. The Difference command was used to calculate a distribution for the amount of time that passed between the start and end of the peat initiation process. A t-type outlier model (Ramsey, 2009) using a T(5) distribution and U(0,4) scale was combined with the P\_Sequence. The t-type outlier model is intended for cases where the measured sample might not relate to the event being dated (Ramsey, 2009). The prior probability for a macrofossil date to be an outlier was set to 5% (i.e., one out of twenty might be an outlier). The resulting age-depth models were calculated based on a model averaging approach, where macrofossil dates that are more probable to be outliers are down-weighted (Ramsey, 2009). See the link to the dataset on page 57 for the OxCal scripts.

The likelihoods of humic and humin ages were plotted together with the macrofossil-based P\_Sequences. The degree of overlap of the humic/humin likelihoods with the 95% confidence interval of the P\_Sequences indicates the accuracy of the humic/humin dates in representing the age of the peat layer from which they were obtained.

**Table 3.5:** Overview of the charred aboveground plant remains that were selected for radiocarbon dating. When a number is given, this is the exact amount encountered, cf. = resembles, + = present, ++ = frequent. The sample weight is the mass of the sample before the start of the chemical pre-treatment.

Core	From (m O.D.)	To (m O.D.)	Aboveground plant remains (all charred) for $^{14}C$ dating	Sample weight (mg)
S17	7.495	7.505	<i>Calluna vulgaris</i> , twig fragments +	8.63
			Herbaceous stem, fragments $+$	
			Sphagnum, stem fragments +	
	7.485	7.495	<i>Erica tetralix</i> , leaf fragments +	20.78
			<i>Calluna vulgaris</i> , twig fragments +	
			Calluna/Erica, twig fragments $+$	
			Herbaceous stem, fragments +	
			<i>Bryales</i> , stem fragments +	
	7.475	7.485	<i>Erica tetralix</i> , 3 leaves	4.21
			Eriophorum vaginatum, 1 spindle	
			<i>Calluna/Erica</i> , twig fragments +	
	7.300	7.310	Deciduous wood, undetermined	1.36
	7.290	7.300	Herbaceous stem fragments +	0.87
	7.280	7.290	Herbaceous stem ~18 fragments (incl. cf. <i>Juncus</i> )	11.75
	7.210	7.220	Herbaceous stem fragments (cf. <i>Eriophorum</i> ) ++	9.70
	7.200	7.210	Herbaceous stem and stem base fragments (cf. Eriopho-	12.99
			rum) ++	
	7.190	7.200	Herbaceous stem ~16 small fragments	3.07
S18	11.79	11.80	<i>Calluna vulgaris</i> , twig fragments +	2.34
			<i>Calluna/Erica</i> , twig fragments +	
			Herbaceous stem, fragments $+$	
			Sphagnum, stem fragments +	
	11.78	11.79	<i>Calluna/Erica</i> , twig fragments +	0.57
			Herbaceous stem, fragments +	
	11.77	11.78	Cyperaceae (cf. <i>Eriophorum</i> ), stem base 3 fragments	3.30
			Herbaceous stem, fragments +	
	11.73	11.74	Calluna/Erica, twig fragments (lower parts) $+$	8.16
			Herbaceous stem, fragments $+$	
			Sphagnum, stem 1 fragment	
	11.72	11.73	Calluna/Erica, twig fragments $+$	7.44
			Herbaceous stem, fragments $+$	
S20	10.37	10.38	Erica tetralix, 1 leaf	4.98
			Calluna/Erica, twig fragments $+$	
	10.36	10.37	Erica tetralix, 2 leaves	18.09
			Calluna/Erica, twig fragments $+$	
	10.33	10.34	Erica tetralix, 1 leaf	4.86
			Calluna/Erica, twig fragments $+$	
			Bryales, stem fragments +	
	10.32	10.33	Calluna/Erica, twig fragments	5.26
	10.31	10.32	<i>Calluna/Erica</i> , twig fragments +	4.47
			Herbaceous stem, fragments $+$	
	10.26	10.27	Herbaceous stem, 2 small fragments	0.04
	10.25	10.26	Herbaceous stem, fragments +	1.58
			Charcoal, small fragments +	
	10.24	10.25	Herbaceous stem, 1 fragment	1.04
			Charcoal, 1 small fragment	

# 3.3 Results

# 3.3.1 The organic matter gradient of the mineral-to-peat transition: defining basal peat $(M_d)$

The five vertical series of LOI measurements to derive the OM gradient at the mineral-to-peat transition are shown in fig. 3.6. The data show a clear and abrupt rise of OM over a distance of a few cm. Based on this outcome *peat* was defined as material with an OM percentage of 40% or higher (i.e., the defining value  $M_d = 40\%$ ). The layer called *basal peat* is therefore the first cm of material with OM  $\geq 40\%$ .

# 3.3.2 Plant macrofossils (PM) and testate amoebae (TA)

Overall the samples that originated from the stratigraphical layer described as amorphous organic matter (table 3.2) contained few macrofossils. In the non-decomposed (waterlogged) peat (table 3.2) mostly belowground remains such as rootlets, radicelle and rhizomes were preserved. Few waterlogged aboveground plant tissues were present, however charred aboveground remains could often be identified (table 3.6).

In general, all investigated levels of the three cores contained small unidentifiable rootlets. Part of the rootlets was of Cyperaceous origin, of which some were identified as Carex radicelle. The bottom levels of S18 and S20 contained fine sand, and in some of these levels also sclerotia of the mycorrhizal fungus Cenococcum geophilum were present. In the core of lowest elevation, S17, the bottom levels (7.17 to 7.20 m O.D.) contained a few leaves of Sphagnum austinii (S. imbricatum) and six megaspores of Selaginella selaginoides. From 7.20 to 7.22 m O.D. many charred herbaceous stems were present, of which some likely derived from stem bases (corm) of *Eriophorum*. Slightly higher in the profile (7.28 to 7.31 m O.D.) fewer herbaceous stems were observed and many sclerotia of the fungus Cenococcum geophilum were found. The upper three levels, from 7.475 to 7.505 m O.D., contained vegetative remains of Eriophorum vaginatum, Erica tetralix, Calluna vulgaris, Bryales species, Sphagnum austinii and other Sphagnum species. In core S20, between 10.30 and 10.34 m O.D., leaf and/or stem remains of *Poaceae* (including *Phragmites* and other species) were present, which were not found in core S17 and S18. In the upper levels of cores S18 and S20, charred herbaceous stems and twigs of *Calluna vulgaris* and/or *Erica tetralix* were found. In addition, the upper levels of core S20 contained some stems of *Bryales* mosses, whereas in core S18 some charred Sphagnum stems were found. Both core S18 and S20 contained waterlogged seeds of Juncus (Juncus conglomeratus/effusus) in their upper levels.

Almost none of the investigated levels contained suitable material for testate amoebae analyses. In four subsamples only a few damaged or broken tests were present, which could be identified as *Difflugia pristis* or other *Difflugia* species.



**Figure 3.6:** (a) Organic matter data of five cores (for locations, see fig. 3.2). The dashed horizontal line shows the value of  $M_d$ , which was set at 40% OM. The dotted horizontal lines indicate an OM content of 35% and 45% for ease of comparison. For one 1-cm-thick layer of core 20 no data were available, this causes the discontinuity in the green line. (b) Schematic core rotated 90° clockwise, showing conceptual organic matter gradient (see also fig. 3.1b.

Core	From (m 0.D.)	To (m O.D.)	Sphagnum, stem fragments (c)	Sphagnum austinii, leaves	<i>Bryales</i> , stems (c)	Calluna vulgaris, twig fragments (c)	Calluna vulgaris, leaves (c)	Calluna vulgaris, leaves	<i>Calluna/Erica</i> , twig fragments (c)	Erica tetralix, leaves, fragments (c)	Erica tetralix, leaves, fragments	Poaceae (excl. Phragmites), leaf/stem epider	Poaceae (incl. <i>Phragmites</i> ), leaf/stem epider	<i>Eriophorum vaginatum</i> , spindle (c)	Cyperaceae, cf. Eriophorum, stem base (c)	Herbaceous stem fragments (c)	Juncus, seeds	Juncus conglomeratus/effusus, seeds	<i>Carex</i> , radicelle	Cyperaceae, rootlets/radicelle	Undetermined rootlets/radicelle	Undetermined rhizome epidermis	Wood, decidous, fragments (c)	Cenococcum geophilum, sclerotia	Selaginella selaginoides, megaspores	Difflugia	Difflugia pristis	Charcoal, fragments	Fine sand	Stratigraphy
17	7.495	7.505	+	+		+		2								+					+++	+								peat
17 17	7.465	7.495		+	+	+		2	+	+3				1		+					+ +++	+								peat
											:														:			::		
17	7.430	7.440				8	2			12	1															NA	NA			peat
17	 7 390	7 400			+											+	+							+		NA	NA			neat
											1													·····	:			::		pour
17	7.300	7.310																		+	+	++	+	++++						peat
17	7.290	7.300														+				+	+++	+++		+++						peat
11	1.200	1.290														10						· · · ·								реас
17	7.210	7.220													+	++			+	+	+	+								MtP
17	7.200	7.210													+	++			+	+	+	+								BP
17	7.190	7 100		+												10			+	+	+	+								MtP M+P
17	7.170	7.180		+															+	+	+				6		2			MtP
18	11.79	11.80	+			+			+							+		++	+	+	+									peat
18	11.78	11.79							+						2	+		++	+	+	+									MtP
18	11.77	11.78													3	+		++	+	+	+									MITP
18	11.73	11.74	+						+							+			+	+	+					1				BP
18	11.72	11.73							+							+			+	+	+								+	MtP
18	11.71	11.72																	+	+	+								+	mineral
10	11.70	11./1																	+	++	++			+					+	mineral
18	11.68	11.69																	+	++	++			+						mineral
18	11.67	11.68						•••••• 	<u>.</u>										+	+	+								+++	mineral
18	11.66	11.67						ļ											+	+	+								+++	mineral
$\frac{10}{20}$	10.37	10.38		:		:		-	+	1	: :							++	+	++	++		-	+					+++	neat
20	10.36	10.37							+	2									+	+	++									peat
20	10.33	10.34			+				+	1			+							+	++	+								MtP M+D
20	10.32	10.33							+							+				++	++	+								MtP
20	10.30	10.31										+							+	+	++									BP
20	10.29	10.30																		+	++					1				MtP
20	10.20	10.07											;			2									:			;;		- main and I
20 20	10.20	10.27																	+	+++++++++++++++++++++++++++++++++++++++	++			++				+	++	mineral
20	10.24	10.25														1			+	+	++			1 1				1	++	mineral

**Table 3.6:** (Table on previous page). Results of the analyses of plant macrofossils and testate amoebae. When a number is given, this is the exact amount encountered, (c) = charred, cf. = resembles, + = present, ++ = frequent, +++ = abundant, ++++ = extremely abundant, NA = not available. Final column indicates stratigraphy (also see table 3.2), MtP (indicated with light grey shading) = mineral-to-peat transition (see fig. 3.7), BP = basal peat.

### 3.3.3 Dating results

The radiocarbon dating results for the investigated levels for each core are shown in tables 3.7 to 3.9. The calibrated ages, modelled P\_Sequences, and OM content are shown in fig. 3.7. Overall, the chronological order of the dates from macrofossils concurs with stratigraphical position (older at the bottom and younger towards the top). However, there is one reversal in core S20 at 10.31 to 10.32 m O.D.. The chronological order of the humic dates is also largely correct, with a few exceptions of minor reversals (S17: 7.19 and 7.28 m O.D.; S18: 11.65 m. O.D.; S20: 10.24 m. O.D.). For higher levels (above the mineral-to-peat transition), humin ages converge with those based on humics and plant macrofossils. For lower levels however, humin ages are scattered.

Dates of plant macrofossils, humics and humins diverge for samples from the mineral-to-peat transition (i.e., in fig. 3.7 between 'Start of peat initiation process' and 'End of peat initiation process'), especially for core S17. The higher in the profile and the further away from the mineral-to-peat transition, dates of plant macrofossils and both humic and humin fractions are increasingly in agreement. At the levels where the radiocarbon ages diverge, dates of plant macrofossils represent the oldest fraction in cores S17 and S18. For core S17 the difference between the macrofossils and humics is relatively constant for the samples below 7.22 m O.D., while also the dates of these fractions are relatively constant with increasing depth. In core S20, no plant macrofossils were available from the sandy layers at the bottom. Here, the humics are generally oldest, humin ages are very dispersed.

The part of the stratigraphy with the rising OM gradient ('period of peat initiation' as explained in fig. 3.1b, also see fig. 3.6) reflects the timespan (duration) of the peat initiation process. Based on the P\_Sequences presented in fig. 3.7, this timespan was modelled (fig. 3.8). Results show that the peat initiation process took a median of 1073 years at the location of core S17 (91 – 2706 years at 95.4% probability). At the site of S20 the process took a bit longer with a median duration of 1343 years (472 – 2040 at 95.4%). The process took longest at site S18, here the median lies at 1510 years (1301 – 1714 years at 95.4%). Figure 3.7: (Figure on next pages). Overview of dating results and OM percentages for core S17 (a-d), S18 (e-h) and S20 (i-l). (a/e/i) Age-depth plot showing the likelihoods of all dated fractions; green = macrofossils, orange = humics, pink = humins. Note that some likelihoods overlap, see tables 3.7-3.9 for an overview of all dated fractions per layer. (b/f/i) Age-depth plot showing the result of a P Sequence based on macrofossil dates, accompanied with an outlier model (see Methods for details). The probability for a date to be an outlier is indicated behind each date at the left side of the plot in the format [O:x/5], where x gives the posterior probability and 5 the prior probability that was entered in the outlier model (always set to 5%). Blue shading = 95.4% confidence interval. + = median of modelled posterior distribution, orange = likelihood of humics. pink = likelihood of humins. Degree of overlap of the humic/humin likelihoods with the confidence interval of the P sequence indicates accuracy of the humic/humin dates in representing the age of the peat layer from which they were obtained. (c/g/k) Organic matter data, the combination of dashed and dotted lines indicates which sample is the first with OM > 40%, dark grey shading = samples that encompass the peat initiation process (mineral-to-peat transition). (d/h/l) Schematic cores showing conceptual stratigraphy as in fig. 3.1b and 3.6b.







From	То						<sup>14</sup> C				
(m	(m	Sample	Dated				age	$\pm$	$\delta^{13}C$	$\pm$	
Ò.D.)	Ò.D.)	name	fraction	Lab-ID	F <sup>14</sup> C	$\pm$ (1 $\sigma$ )	(yrBP)	<b>(1</b> σ)	(IRMS)	<b>(1</b> σ)	%С
7.495	7.505	S17-M9-B	humin	GrM-23376	0.5775	0.0019	4410	26	-27.09	0.15	62.1
7.485	7.495	S17-M8-B	humin	GrM-23797	0.5759	0.0018	4433	26	-27.17	0.15	55.0
7.475	7.485	S17-M7-B	humin	GrM-23515	0.5354	0.0017	5019	26	-27.62	0.15	64.5
7.300	7.310	S17-M6-B	humin	GrM-23516	0.4345	0.0015	6696	29	-27.56	0.15	63.7
7.290	7.300	S17-M5-B	humin	GrM-23517	0.4504	0.0016	6407	29	-28.57	0.15	55.5
7.280	7.290	S17-M4-B	humin	GrM-23520	0.4255	0.0016	6864	30	-27.79	0.15	56.3
7.210	7.220	S17-M12-B	humin	GrM-23731	0.4973	0.0017	5612	27	-28.28	0.15	13.9
7.200	7.210	S17-M11-B	humin	GrM-23732	0.4817	0.0017	5868	29	-28.08	0.15	5.9
7.190	7.200	S17-M3-B	humin	GrM-23733	0.4247	0.0016	6878	30	-28.25	0.15	9.5
7.180	7.190	S17-M2-B	humin	GrM-23734	0.4644	0.0017	6162	29	-28.05	0.15	10.7
7.170	7.180	S17-M1-B	humin	GrM-23736	0.5023	0.0017	5532	27	-28.16	0.15	7.2
7.495	7.505	S17-M9-B	humic	GrM-23285	0.5821	0.0019	4346	26	-28.13	0.15	57.2
7.485	7.495	S17-M8-B	humic	GrM-23829	0.5759	0.0019	4433	27	-27.09	0.15	50.5
7.475	7.485	S17-M7-B	humic	GrM-23830	0.5477	0.0018	4836	27	-28.45	0.15	49.1
7.300	7.310	S17-M6-B	humic	GrM-23831	0.4527	0.0016	6365	30	-28.23	0.15	50.3
7.290	7.300	S17-M5-B	humic	GrM-23832	0.4486	0.0016	6439	29	-28.45	0.15	56.8
7.280	7.290	S17-M4-B	humic	GrM-24024	0.4585	0.0016	6264	29	-27.49	0.15	9.3
7.210	7.220	S17-M12-B	humic	GrM-24025	0.4408	0.0016	6580	29	-26.79	0.15	15.7
7.200	7.210	S17-M11-B	humic	GrM-23865	0.4346	0.0016	6695	30	-26.99	0.15	20.9
7.190	7.200	S17-M3-B	humic	GrM-23866	0.4465	0.0016	6477	30	-27.30	0.15	15.0
7.170	7.180	S17-M1-B	humic	GrM-23867	0.4383	0.0016	6625	30	-27.57	0.15	22.1
7.495	7.505	S17-M9	CPR	GrM-23521	0.5791	0.0017	4388	24	-26.32	0.15	66.1
7.485	7.495	S17-M8	CPR	GrM-23522	0.5786	0.0019	4395	26	-26.55	0.15	61.5
7.475	7.485	S17-M7	CPR	GrM-23523	0.5315	0.0018	5077	27	-27.85	0.15	67.1
7.300	7.310	S17-M6	CPR	GrM-23491	0.4032	0.0041	7300	80	NA	NA	NA
7.280	7.290	S17-M4	CPR	GrM-23524	0.4097	0.0016	7168	30	-26.30	0.15	64.0
7.210	7.220	S17-M12	CPR	GrM-23492	0.3513	0.0033	8400	80	NA	NA	NA
7.200	7.210	S17-M11	CPR	GrM-23525	0.3556	0.0013	8305	30	-25.42	0.15	63.1
7.190	7.200	S17-M3	CPR	GrM-23493	0.3528	0.0036	8370	80	NA	NA	NA

**Table 3.7:** Dating results for core S17. CPR = charred plant remains, NA = not available.

From	To						<sup>14</sup> C				
(m	(m	Sample	Dated				age	±	$\delta^{13}C$	$\pm$	
Ò.D.)	Ò.D.)	name	fraction	Lab-ID	F <sup>14</sup> C	$\pm$ (1 $\sigma$ )	(yrBP)	<b>(1</b> σ)	(IRMS)	<b>(1</b> σ)	%С
11.80	11.81	S18-M12-B	humin	GrM-23819	0.8017	0.0022	1775	22	-27.79	0.15	49.9
11.79	11.80	S18-M11-B	humin	GrM-23820	0.7969	0.0021	1823	22	-28.20	0.15	24.4
11.78	11.79	S18-M10-B	humin	GrM-23822	0.7979	0.0022	1814	22	-28.13	0.15	42.9
11.77	11.78	S18-M9-B	humin	GrM-23825	0.7971	0.0028	1821	29	-27.74	0.15	38.0
11.73	11.74	S18-M8-B	humin	GrM-23826	0.6973	0.0022	2896	26	-27.30	0.15	20.1
11.72	11.73	S18-M7-B	humin	GrM-24009	0.8177	0.0044	1615	45	NA	NA	1.4
11.71	11.72	S18-M6-B	humin	GrM-24010	0.9203	0.0049	665	45	NA	NA	2.8
11.70	11.71	S18-M5-B	humin	GrM-24011	0.5993	0.0035	4115	45	NA	NA	2.8
11.68	11.69	S18-M4-B	humin	GrM-24012	0.6145	0.0033	3910	45	NA	NA	0.08
11.67	11.68	S18-M3-B	humin	GrM-24013	0.5070	0.0028	5455	45	NA	NA	2.4
11.66	11.67	S18-M2-B	humin	GrM-24014	0.6602	0.0036	3335	45	NA	NA	0.05
11.65	11.66	S18-M1-B	humin	GrM-24015	0.8293	0.0048	1505	45	NA	NA	0.05
11.80	11.81	S18-M12-B	humic	GrM-23499	0.7958	0.0064	1840	60	NA	NA	NA
11.79	11.80	S18-M11-B	humic	GrM-23868	0.7963	0.0021	1830	21	-27.91	0.15	49.0
11.78	11.79	S18-M10-B	humic	GrM-23870	0.7938	0.0020	1855	21	-28.22	0.15	53.4
11.77	11.78	S18-M9-B	humic	GrM-23871	0.7790	0.0021	2006	22	-28.66	0.15	53.2
11.73	11.74	S18-M8-B	humic	GrM-23872	0.7117	0.0019	2732	22	-28.35	0.15	49.1
11.72	11.73	S18-M7-B	humic	GrM-23873	0.6861	0.0021	3026	24	-28.31	0.15	38.2
11.71	11.72	S18-M6-B	humic	GrM-23875	0.6029	0.0019	4064	26	-28.04	0.15	36.6
11.70	11.71	S18-M5-B	humic	GrM-23878	0.5718	0.0019	4491	26	-28.31	0.15	36.3
11.68	11.69	S18-M4-B	humic	GrM-23879	0.5526	0.0018	4765	26	-28.93	0.15	39.8
11.67	11.68	S18-M3-B	humic	GrM-23880	0.5410	0.0018	4935	27	-28.91	0.15	37.9
11.66	11.67	S18-M2-B	humic	GrM-23881	0.5380	0.0018	4980	27	-29.06	0.15	32.0
11.65	11.66	S18-M1-B	humic	GrM-23882	0.5512	0.0018	4785	26	-29.16	0.15	28.5
11.79	11.80	S18-M11	CPR	GrM-23287	0.7902	0.0024	1892	24	-28.18	0.15	62.9
11.73	11.74	S18-M8	CPR	GrM-23827	0.6863	0.0020	3024	24	-27.69	0.15	64.0
11.72	11.73	S18-M7	CPR	GrM-23828	0.6659	0.0020	3267	24	-27.62	0.15	62.2

**Table 3.8:** Dating results for core S18. CPR = charred plant remains, NA = not available.

From	То						<sup>14</sup> C				
(m	(m	Sample	Dated				age	$\pm$	$\delta^{13}C$	$\pm$	
0.D.)	0.D.)	name	fraction	Lab-ID	F14C	$\pm$ (1 $\sigma$ )	(BP)	<b>(1</b> σ)	(IRMS)	<b>(1</b> σ)	%С
10.38	10.39	S20-M10-B	humin	GrM-23798	0.7686	0.0043	2115	45	-28.06	0.15	49.1
10.37	10.38	S20-M9-B	humin	GrM-23799	0.7692	0.0023	2108	24	-28.55	0.15	47.2
10.36	10.37	S20-M8-B	humin	GrM-23800	0.7693	0.0023	2107	24	-27.04	0.15	27.8
10.35	10.36	S20-M7-B	humin	GrM-23801	0.7423	0.0022	2394	24	-28.10	0.15	37.5
10.33	10.34	S20-M12-B	humin	GrM-23802	0.7131	0.0024	2716	27	-29.16	0.15	16.2
10.32	10.33	S20-M11-B	humin	GrM-23803	0.6628	0.0020	3304	24	-28.60	0.15	26.7
10.31	10.32	S20-M6-B	humin	GrM-23812	0.6608	0.0020	3328	24	-29.69	0.15	29.5
10.30	10.31	S20-M5-B	humin	GrM-23813	0.6518	0.0021	3438	26	-30.09	0.15	34.9
10.29	10.30	S20-M4-B	humin	GrM-24006	0.7512	0.0022	2298	24	NA	NA	7.7
10.26	10.27	S20-M3-B	humin	GrM-24007	0.7451	0.0043	2365	45	NA	NA	2.4
10.25	10.26	S20-M2-B	humin	GrM-23737	0.6413	0.0033	3570	40	NA	NA	13.2
10.24	10.25	S20-M1-B	humin	GrM-24008	0.7963	0.0040	1830	40	NA	NA	1.3
10.38	10.39	S20-M10-B	humic	GrM-23968	0.7723	0.0021	2076	22	-28.07	0.15	53.4
10.37	10.38	S20-M9-B	humic	GrM-24018	0.7707	0.0025	2093	26	-27.94	0.15	42.8
10.36	10.37	S20-M8-B	humic	GrM-24017	0.7543	0.0026	2265	27	-26.83	0.15	16.6
10.33	10.34	S20-M12-B	humic	GrM-23883	0.6954	0.0021	2918	24	-28.05	0.15	35.1
10.32	10.33	S20-M11-B	humic	GrM-24019	0.6635	0.0025	3295	30	-28.20	0.15	27.8
10.31	10.32	S20-M6-B	humic	GrM-23884	0.6530	0.0021	3424	26	-28.22	0.15	49.1
10.30	10.31	S20-M5-B	humic	GrM-24021	0.6419	0.0022	3561	27	-28.77	0.15	46.7
10.29	10.30	S20-M4-B	humic	GrM-23969	0.6473	0.0033	3495	40	-29.29	0.15	32.7
10.26	10.27	S20-M3-B	humic	GrM-23885	0.6247	0.0020	3779	26	-29.11	0.15	30.0
10.25	10.26	S20-M2-B	humic	GrM-24022	0.6135	0.0020	3924	27	-28.75	0.15	14.9
10.24	10.25	S20-M1-B	humic	GrM-23886	0.6290	0.0023	3725	29	-28.85	0.15	20.2
10.37	10.38	S20-M9	CPR	GrM-23814	0.7580	0.0020	2225	22	-24.54	0.15	66.4
10.36	10.37	S20-M8	CPR	GrM-23815	0.7577	0.0020	2229	22	-25.24	0.15	63.9
10.33	10.34	S20-M12	CPR	GrM-23208	0.6962	0.0027	2910	30	-26.94	0.15	60.6
10.32	10.33	S20-M11	CPR	GrM-23817	0.6746	0.0022	3163	26	-26.90	0.15	71.3
10.31	10.32	S20-M6	CPR	GrM-23818	0.6906	0.0021	2974	24	-26.53	0.15	63.1

**Table 3.9:** Dating results for core S20. CPR = charred plant remains, NA = not available.

# 3.4 Discussion

Here we discuss the process of peat initiation as reflected by the mineral-topeat transition in the stratigraphical record, the resulting definition of *basal peat* (3.4.1), followed by the reconstructed palaeoenvironment (3.4.2), the course and timespan of peat initiation (3.4.3), and the age assemblage of the different carbon fractions (3.4.4). Based on this, recommendations for dating *basal peat* are formulated (3.4.5).

## **3.4.1** Mineral-to-peat transition and basal peat (defining $M_d$ )

Our analyses demonstrate that the organic matter content shows a clear and steep rise over a distance of a few centimetres, starting at low values of about 10% and increasing to more than 90% (fig. 3.6). This gradient reflects the stratigraphical mineral-to-peat transition. The drastic rise in OM content occurs around an OM percentage of 40%. In the explanation of the Dutch soil classification system by De Bakker and Schelling (1966), a range of organic matter classes is defined based on OM and clay percentages (by mass). For soils containing 0 to 8% clay (as in our case study area), the material is called *peat* when containing >35% OM (in case of 8% clay) or >40% OM (when containing 0% clay). This is in strong agreement with the results of our LOI tests (fig. 3.6), based on which we have selected 40% OM as cut-off value ( $M_d$ ) above which we define the material as *peat*. The first 1-cm-thick subsample that contained  $\geq$  40% OM is therefore defined as the basal peat layer.

## 3.4.2 Palaeoenvironment

The investigated cores contained a limited amount of well-preserved plant macrofossils. Part of these remains is charred (table 3.6). In several levels sclerotia of *Cenococcum geophilum*, a mycorrhizal fungus that usually lives in the sandy subsoil, were observed (table 3.6). In peat, the presence of *C. geophilum* may indicate relatively dry conditions (Van Geel, 1978). Presence of charred and uncharred plant remains in multiple levels of the cores suggests (local) wildfires (for more information on (palaeo)wildfires in peatlands see e.g. Zaccone *et al.*, 2014; Nelson *et al.*, 2021; Rein and Huang, 2021). Dry periods allow further breakdown of material, which could explain the rather poor preservation of uncharred (waterlogged) macrofossils.

It is important to note that even though a bog remnant is studied here, the basal peat is in fact fen peat (which is quite often the case; see e.g. Korhola, 1994; Cubizolle *et al.*, 2007). The mineral-to-peat transition is later followed by a fen-bog transition as a result of ombrotrophication (for more information on

the latter transition see e.g. Almquist-Jacobson and Foster, 1995; Hughes, 2000; Hughes and Barber, 2004; Väliranta *et al.*, 2017; Loisel and Bunsen, 2020). Our findings indicate that at all three cored locations the peat-forming vegetation was initially mesotrophic. The local vegetation was dominated by sedges, with some presence of *Juncus* (table 3.6). After the peat initiation process, conditions became more oligotrophic, and the vegetation at S17 and S18 developed probably to an oligotrophic bog with *Calluna vulgaris, Erica tetralix* and *Sphagnum* (ombrotrophic conditions). At the location of S20 no *Sphagnum* remains were found, but the vegetation likely changed to a moss (*Bryales*) and heather vegetation.

The investigated cores appeared to be very low in testate amoebae content. However, the presence of *Difflugia* species in a few samples suggests very wet conditions. This taxon and the presence of abundant diatoms (which were observed during TA analysis but not subject of further study) suggest a fen environment (rather than an ombrotrophic bog), which is in agreement with the botanical data. Testate amoebae are often poorly preserved under fen-type conditions, possibly due to predation, physical disaggregation or chemical dissolution, or a combination of these (Roe *et al.*, 2002; Swindles and Roe, 2007; Swindles *et al.*, 2020).

In the investigated levels, *C. geophilum* does not occur simultaneously with *Difflugia* species. This mutual exclusion suggests respectively drier and wetter conditions that alternated during the timespan of the peat initiation process. Additionally, a detailed study by Sullivan and Booth (2011) has shown that several *Difflugia* species (including *D. pristis*) are able to cope with fairly high levels of short-term variability in environmental conditions at the peat surface. This variability appeared to be higher at locations with loose-growing *Sphagnum* (rather than dense *Sphagnum* cover) or where vegetation was dominated by vascular plants and non-*Sphagnum* bryophytes (Sullivan and Booth, 2011). We suggest that such conditions may be similar to those in our study area during peat initiation and the transition to an oligotrophic bog.

## 3.4.3 Time span of the peat initiation process

The stratigraphical reach with the rising OM gradient (fig. 3.1b and 3.6) reflects the timespan of the peat initiation process. Based on the P\_Sequences in fig. 3.7, these timespans were modelled for each core (fig. 3.8). Results show that this process lasted for 1073, 1510 and 1343 years (medians) for core S17, S18, and S20 respectively. In the stratigraphy, this is reflected in a vertical distance of respectively 8, 7 and 6 cm. This means that apparent vertical accumulation during these first stages of peat development varied between the sites with values of 0.07, 0.05 and 0.04 mm/year for core S17, S18, and S20 respectively. A typical value given for the apparent peat accumulation rate in the catotelm is 1 mm/year,

and may be lower further down in the catotelm due to compaction and anaerobic decomposition (Rydin and Jeglum, 2013e). A low apparent accumulation rate is indeed the case here, with rates far below 1 mm/year. This implies that 1 cm of peat, at the slowest rate of 0.04 mm/year, reflects about 250 years of peat growth.

Very few studies have investigated the timespan that is reflected in the first centimetres above the layer they define as basal peat. For two cores, Berendsen *et al.* (2007) dated a pair of vertically spaced samples (taken 11 cm apart in one core and 9 cm apart in the other), and found age differences of respectively 60 and 120 calendar years. Based on this, they conclude that within-core sampling resolution is less critical than previously assumed.

The difference between our results and those of Berendsen *et al.* (2007) highlights that the peat initiation process cannot be assumed to be rapid in all cases and is influenced by environmental setting. Depending on the timespan of the peat initiation process and the apparent accumulation rate, a high vertical sampling resolution and small sample thickness can be crucial to obtain accurate dates. The duration of peat initiation also determines to which degree a date of basal peat is representative to use as starting point for build-up of peat deposits.

It is important to note that the reconstructed timespans and apparent accumulation rates are based on age-depth relationships (instead of a mass-age relationship). These age-depth relationships do not consider gross accumulation and subsequent decay separately, but only the apparent vertical increase (potentially affected by decomposition and/or compaction). Due to water-saturated conditions for significant periods of time, bioturbation by soil fauna is presumably low during the peat initiation process. This assumption is corroborated by the intact chronostratigraphy of macrofossils shown in our cores (fig. 3.7a/e/i). As the timespan of the peat initiation process is potentially long, we emphasize that sampling resolution and sample thickness are key points to consider when dating the start of peat growth.

**Figure 3.8:** (Figure on next page). Timespan (duration in years) of the peat initiation process in cores S17, S18 and S20, derived from the P\_Sequence models shown in fig. 3.7(b/f/j) (see Methods for details). Apparent accumulation rate (expressed in mm/year) was calculated by dividing the stratigraphical distance of the peat initiation process (provided in cm, see also fig. 3.7(b/f/j)) by the modelled median of the timespan. Note that these accumulation rates are based on an age-depth relationship (instead of a mass-age relationship) and do not consider gross accumulation and subsequent decay separately, but only the apparent vertical increase (potentially affected by decomposition and/or compaction).



#### 3.4.4 Age assemblage of carbon fractions

Our data demonstrate that macrofossils (i.e., in situ material) in the basal peat layer are oldest, and that both humics and humins generally show younger ages. In line with the general consensus in literature (e.g. Piotrowska et al., 2011), we consider the macrofossils to reflect the 'true' age, i.e. representative for the timing when the vegetation accumulated at the specific location. Aboveground remains (no roots) of terrestrial plants are expected to have been in equilibrium with atmospheric <sup>14</sup>C values until they died and we therefore do not expect any reservoir effect (also see Blaauw et al., 2004). In this study, samples were carefully selected, cleaned and pre-treated with the full ABA-protocol to minimise the presence of any contamination with carbon from sources other than the original plant materials (see sections 3.2.6 and 3.2.8). The macrofossils show a clear chronological order, with samples dating younger upwards in the profile, which suggests that macrofossil relocation through bioturbation is unlikely. There is only one reversal for the macrofossils in core S20 at 10.31 m O.D., while the humic and humin fractions of this same layer do not show a deviation in chronology. The cause for this deviation remains unclear. The outlier model shows that there is an increased chance that either S20M11macro or S20M6macro is an outlier; through the model averaging approach these dates are corrected in the P\_Sequence (fig. 3.7j).

Despite the correct chronological order of the humic dates, the age difference between humics and plant macrofossils in core S17 from 7.19 to 7.30 m O.D. shows that the absolute humic ages are likely too young (fig. 3.7a and 3.7b). This difference in age ranges from about 1700 to 800 years. The same applies to core S18 at depths 11.72 and 11.73 m O.D. (fig. 3.7e and 3.7f), with age differences of about 250 to 400 years. This is different in core S20, where two humic samples dated older than macrofossils from the same layer. At 10.31 m O.D., the macrofossil sample deviates from the chronology (see above), which causes a fairly large difference with the humic age of approximately 530 years (fig. 3.7i and 3.7j). At 10.32 m O.D. however, where the macrofossil date concurs with chronological order, the humic date is only about 120 years older.

The humins, both in the *basal peat* but also below in the Pleistocene deposits, show younger ages than the humics and plant macrofossils, and remarkably little coherence. In the sandy Pleistocene layers the organic matter content was low and sand content high. Separating this small amount of carbon from the sand in the lab appeared to be challenging, resulting in several humin samples with fairly low %C (tables 3.7 to 3.9). If some younger material is incorporated here, the influence on the resulting age will be larger due this small sample size. This may account for part of the observed variation.

Overall, both the humic and humin fractions derived from the mineral-to-peat

transition result in younger ages. Fluctuating water tables during the process of peat initiation (discussed above), might explain the origin of younger carbon in these first peat layers. Changes in hydraulic head may lead to both upward and downward (and also sideward) movement of soluble organic compounds (Waddington and Roulet, 1997). Low water levels specifically allow downward water movement through the profile, potentially transporting mobile humic acids that contain young carbon to lower levels. As the mobility of humics is pH-dependent (Wüst *et al.*, 2008), the initial mesotrophic conditions allow higher mobility than the more acid, ombrotrophic conditions that follow later in time (see above).

Brock *et al.* (2011) dated humic and humin fractions from three grain sizes obtained through wet-sieving (63-125  $\mu$ m, 125-250  $\mu$ m and >250  $\mu$ m), originating from a 1-cm-thick layer positioned 1 cm below the level they regarded to reflect peat initiation. Their results show that for two of these grain sizes, the humic and humin dates are not significantly different from each other. However, both the dates of the humics and the humins become older with increasing grain size, suggesting that the fine particulate matter may be responsible for younger contaminations in both fractions. In our study, the bulk material was not sieved during the entire pre-treatment procedure, to secure that the very small organic particles in the bulk material were retained. Presence of this fine fraction may be responsible for the observed younger dates of the humics and humins, but does not explain the erratic pattern of the humin dates.

Downgrowth of roots may also cause age differences in bulk samples compared to aboveground plant macrofossils, a problem that appears to be of particular relevance in the case of slowly accumulating (fen) peat (Streif, 1972; Törnqvist *et al.*, 1992) such as encountered here during the peat initiation process. The effect may also depend on the botanical composition of the peat-forming vegetation, as certain species such as *Phragmites* or *Eriophorum vaginatum* (table 3.6) can produce fairly deep roots (Kohzu *et al.*, 2003; Iversen *et al.*, 2015). As humics and humins result from decomposition, their ages may result partly from in-situ carbon and partly from younger carbon that originated from mobile fulvic and humic acids and roots. Additional dating of separated rootlets at multiple levels slightly above the mineral-to-peat transition may shed more light on the sources of error in the ages of the humics and humins for dating peat initiation.

Holmquist *et al.* (2016), who compared radiocarbon dating results for plant macrofossil and bulk samples obtained from basal peat in circum-arctic peatlands, found no significant difference in ages. Based on this they conclude that evidence for a consistent systematic bias introduced by the incorporation of bulk peat dates in large basal <sup>14</sup>C databases from peatlands is lacking. In contrast, the large age difference between dates of plant macrofossils and humic or humin dates (up to ~1700 years between macrofossil and humic ages in our case study peatland,

and with even larger differences for humins, fig. 3.7a and 3.7e) indicates that studies reusing existing bulk dates of basal peat should take great care in data interpretation. Some of these studies, which concentrate on regional or global reconstructions of peat initiation (e.g. Tolonen and Turunen, 1996; Macdonald *et al.*, 2006; Ruppel *et al.*, 2013), have important implications for climate research and carbon budgets. Depending on the sample type obtained from the dated basal peat layers, dates are potentially interpreted more safely as terminus-ante-quem dates for peat initiation, or should be subjected to rigorous quality assessment prior to data analysis (Quik *et al.*, 2021).

Higher in the peat profile however, dates of plant macrofossils, humics and humins converge, indicating homogeneity regarding carbon fractions and ages. Some of these dates do not differ significantly, others fall within a (very) short timeframe (fig. 3.7b, 3.7f and 3.7j). We suggest that as water tables started fluctuating less, peat accumulation speed began to increase (compare slope in the P\_Sequences, most clearly visible for core S18 in fig. 3.7f), and conditions became more ombrotrophic (start of *Sphagnum* growth, table 3.6). As a result, downward water flow declined and mobility of humics decreased. In line with Törnqvist *et al.* (1992) and Blaauw *et al.* (2004), our findings suggest that when focus is not directed towards the basal peat, but to higher layers in the peat profile that are characterised by more stable water tables and higher accumulation rates, one might obtain accurate dates through bulk sampling (both humics and humins).

In the mineral soil horizons, i.e. those with <40% OM, generally no (aboveground) plant remains could be recognised. Samples from the stratigraphical layers peaty sand/sandy peat and Pleistocene deposits (i.e., the palaeosoil that became covered with peat, table 3.2) could therefore only be radiocarbon dated using the humic and humin fractions. Humic acids are considered to be the most reliable fraction for dating organic matter in soils if no plant remains and (almost) no organic carbon is present (Van der Plicht et al., 2019). Indeed, the humin ages of the Pleistocene layers display poor coherence (fig. 3.7) with frequent stratigraphical inconsistencies. Humic ages in contrast provide results that are stratigraphically consistent. In core S18 for example (fig, 3.7e), the four humic samples between 11.65 to 11.69 m O.D. all date around 5500 cal y BP. This consistency could suggest that these samples represent the slow build-up of organic matter in the sandy soil (i.e., at the time prior to peat growth). Some of the humin dates in the mineral horizons show remarkably young ages, perhaps due to the small amount of total carbon in these samples as mentioned above (tables 3.7 to 3.9), and relatively larger quantities of carbon from other carbon sources. Dating soil organic matter in mineral soils is complicated and involves different processes than peat initiation (see e.g. Goh and Molloy, 1978; Van Mourik et al., 1995; Van der Plicht et al., 2019), further study of these dates is therefore beyond the scope of the present study.

### 3.4.5 Recommendations for dating peat initiation

Our study highlights that peat initiation is a process rather than an event, which has implications for dating peat initiation and *basal peat*. This process is reflected in the stratigraphy as a gradual boundary between mineral sediments and overlying peat deposits. The mineral-to-peat transition can be characterised using the organic matter gradient ( $\frac{dM}{dx}$ , fig 3.1b). The use of biostratigraphical indicators to define basal peat, as for instance in the approach used by Törnqvist *et al.* (1998), is not always possible due to limited presence of plant macrofossils and potential lack of testate amoebae. However, if material is present and resources allow, including additional biostratigraphical analyses to characterise the palaeoenvironment of the peat initiation process is valuable.

The layer that is interpreted as *basal peat* should be defined clearly and quantitatively, which ensures reproducibility and eases intercomparison of studies. To move towards a quantitative definition of *basal peat*, a simple parameter such as OM is useful as it is easy to measure at low cost, which enables widespread use. Based on the obtained organic matter gradient  $\left(\frac{dM}{dx}\right)$  that reflects the peat initiation process, a value  $(M_d)$  can be chosen for the organic matter percentage above which the material is called *peat*. The first cm that has an OM% equal to or above this value is defined as the *basal peat* layer.

Based on our results, an  $M_d$  value of 40% OM would be recommendable to define *basal peat* in areas comparable to our case study peatland. This value agrees very well with the Dutch soil classification (De Bakker and Schelling, 1966), especially given the low clay content of the soils in our study area (table 3.2). However, for peatlands in other regions or with a different botanical composition near the base, LOI-testing may result in a different value for  $M_d$ .

As organic matter measurements using LOI require burning the subsample, care should be taken beforehand to ensure sufficient allocation of sample material to all required analyses. Additionally, it should be kept in mind that post-depositional changes such as downgrowth of roots may change OM content, therefore determining the OM gradient over a reach of several cm is useful to contextualise single measurements. We therefore highly recommend to investigate a stratigraphical range to properly contextualise the mineral-to-peat transition and for selecting an OM value to define the *basal peat*.

An additional advantage of organic matter content determination is that this information may help in estimating chances to obtain sufficient amounts of plant macrofossils for <sup>14</sup>C dating from specific layers. For the three cores investigated in this study, nearly all (12 out of 14) of the macrofossil samples for dating originated
from layers with an organic matter content of 40% or higher. Most of the dated layers with OM below 40% did not contain sufficient macrofossils for dating and were dated solely using humics and humins. Depending on the organic matter gradient and local conditions, this potentially varies between study regions, but may be taken into account for sample selection.

Depending on the timespan of the peat initiation process and apparent accumulation rate, sampling resolution and sample thickness may affect the accuracy of dates in representing the start of peat growth. If accumulation rates are high, a lower vertical sampling resolution or sample size of several cm's might be adequate, whereas lower accumulation rates may require detailed sampling with small sample sizes depending on the research question to be answered. If the timespan of peat initiation and related apparent accumulation rate are unknown, studies aiming to date basal peat with a higher accuracy than several hundred (potentially thousand) years should take great care regarding vertical sampling resolution and sample size, as the assumption of a rapid process is not always valid.

An elaborate dating inventory comparable to the approach of the present study is valuable as it provides detailed information on the peat initiation process and on the accuracy of different carbon fractions for dating. To gain insight in the timespan of peat initiation, and consequently to ensure accurate dates of the peat initiation process and basal peat, dating several vertically spaced samples within one core as a preceding test is very useful. If time allows, such a layered or multi-stage approach for dating peat initiation and lateral expansion is recommendable (also see staged approaches for dating as suggested by Bayliss, 2009 and Piotrowska et al., 2011). For instance, after obtaining the OM gradient dating plant macrofossils and humics of three levels dispersed over the mineral-topeat transition would allow substantiated choices for sampling size and resolution for subsequent spatial dating schemes. These preliminary dates and insights may be used to run simulations in OxCal to evaluate potential outcomes of alternate approaches for more extensive (spatial) sampling and dating. If a multi-stage approach is not possible and only one level of each core is dated without prior tests, assumptions on the timespan of peat initiation and related choices in sampling resolution and sample size should be explicitly discussed.

Our findings show that plant macrofossils provide the most reliable age near the mineral-to-peat transition and the basal peat layer, and are therefore recommendable to use for <sup>14</sup>C dating. The full ABA pre-treatment of macrofossils lowers the risk of chemical contamination and should be applied when possible. If plant macrofossils are unattainable, either due to poor preservation or limited resources for biostratigraphical analyses, dates of the humic fraction provide the best alternative regarding chronological order. These dates may however deviate from those of plant macrofossils, and our data show that humics can therefore

only provide a terminus-ante-quem for the dated levels. If bulk samples are used, it is important to clearly report which fraction was obtained from the sample and used for dating. If marker layers such as tephra are present, these can be used to correlate sites or even provide independent age control if a date can be obtained for the event.

Above the mineral-to-peat transition, our results indicate that dating results of plant macrofossils, humics and humins converge. This implies that humics and humins might be a useful alternative to plant macrofossils for dating peat layers higher in the profile. Justifying this choice would however require a-priori dating knowledge of the peatland under study, and occasional cross-checks with macrofossil dates to ensure accuracy of results.

If the peatland under study is protected as a nature reserve (i.e., not a peat extraction site or location with an outcrop), digging a trench for sampling all the way down to reach the mineral-to-peat transition creates a large disturbance (and would be severely hampered by practical challenges with water infilling). Coring provides an efficient alternative, even though commonly used coring tools such as the Russian corer have a limited sample volume. This limits the options for dating in combination with multiproxy study at a high resolution (Piotrowska *et al.*, 2011). This combination is possible (e.g. as in our study), but requires careful allocation of material to various analyses and involves dating (very) small samples. If chances for obtaining plant macrofossils are encouraging (e.g. based on OM%), one could choose to sacrifice the bulk subsample intended for dating and process it for plant macrofossil analysis. If the expectation regarding plant macrofossils is low, then keeping a bulk sample would offer the possibility to obtain at least a terminus-ante-quem for peat initiation using the humic fraction extracted from the bulk sample.

#### Textbox 3.1: Summary of recommendations for dating peat

- Study the mineral-to-peat transition using the organic matter gradient, and if resources allow, by including biostratigraphical analyses (see [A] in the textbox figure below);
- Take the timespan of the peat initiation process into account when deciding upon sample size and sampling resolution. A layered or multi-stage approach is useful to gain insights on the duration of peat initiation prior to executing elaborate spatial dating schemes (see [B] and [C]);
- Define basal peat based on the organic matter gradient to obtain a low-cost, quantitative and reproducible definition that eases intercomparison of studies (see [D]);
- Regarding which fraction to use for <sup>14</sup>C dating basal peat (see [E]), our data show that plant macrofossils provide the most accurate age in the mineral-to-peat transition and are therefore recommendable to use. If these are unattainable, dates of the humic fraction provide the best alternative regarding chronological order, but may deviate significantly from the 'true age'. Our results show that humic dates are best interpreted as terminus-ante-quem dates;
- Potentially limited options for sampling and resulting small sample volumes require detailed consideration of allocating material to analyses (see fig. 3.5 for an example).



# 3.5 Conclusions

# 3.5.1 Dating peat initiation

In this study we aimed to formulate updated recommendations for dating peat initiation. We based our approach on a conceptual framework (fig. 3.1) that supports the use of the organic matter (OM) gradient for a quantitative and reproducible definition of the mineral-to-peat transition (i.e., the stratigraphical range reflecting the timespan of the peat initiation process) and the layer defined as *basal peat* (i.e., the stratigraphical layer that is defined as the bottom of a peat deposit). Subsequently we analysed the mineral-to-peat transition for a case study peatland in the Netherlands, based on three detailed series of radiocarbon dates that include plant macrofossils, humics and humins. Our findings demonstrate that plant macrofossils, even though their presence in basal peat is often limited, provide the most reliable dating results. If insufficient plant macrofossils are retrieved, the humic fraction provides the best alternative for dating, however dating results are most safely interpreted as a terminus-ante-quem for peat initiation. The potential large age difference between dates of plant macrofossils and humic or humin dates (up to  $\sim 1700$  years between macrofossil and humic ages, and with even larger differences for humins) indicates that studies reusing existing bulk dates of basal peat should take great care in data interpretation. The potentially long timespan of the peat initiation process (with medians of  $\sim 1000$ ,  $\sim 1300$  and  $\sim 1500$  years within our case study peatland) demonstrates that choices regarding sampling size and resolution need to be well substantiated. Our findings are summarised as a set of recommendations for dating *basal peat* in textbox 3.1.

# 3.5.2 Palaeoenvironment

Our case study peatland in the Netherlands currently harbours a bog vegetation, but biostratigraphical analyses show that during peat initiation the vegetation was mesotrophic. This vegetation was dominated by sedges, with some presence of *Juncus*. The data indicate that the peat initiation process initially involved fluctuating water tables and that wetter and drier conditions probably alternated. Frequent presence of charred plant remains demonstrates that wildfires occurred regularly. After the peat initiation process, conditions became more oligotrophic, and the vegetation developed probably to an oligotrophic bog with *Calluna vulgaris, Erica tetralix* and *Sphagnum* at two of the studied locations and to a moss (*Bryales*) and heather vegetation at the third location.

#### Author contributions

RvB secured funding for this research. CQ proposed the research outline, which was further improved based on feedback by JW, JC, BM, YvdV, and RvB. JC and CQ explored the study area, performed test corings and decided upon a coring strategy. CQ organised the fieldwork and completed the collection of data and cores in the field with help of several students (see acknowledgements). CQ coordinated all subsequent analyses that were performed on the obtained cores. Cores were opened and subsampled by MvdL and CQ. LK performed the microscopy work for analyses of plant macro remains and selection of datable material. MvdL interpreted and reported the resulting vegetation data. GS analysed the subsamples for testate amoebae. CQ performed the majority of LOI analyses, a small part of the subsamples were analysed by the CBLB (see acknowledgements). SP worked on the radiocarbon dating at the CIO laboratory in Groningen. All resulting data were combined and analysed by CQ. CQ wrote the main body of text, with input from the reports by MvdL and GS, and from discussions with JW, JC, BM, YvdV and RvB. The manuscript was finalized based on feedback from all authors.

#### Acknowledgements

We thank Roos Veeneklaas (Natuurmonumenten Noordenveld) for permission to do fieldwork at the Fochteloërveen and for sharing detailed field knowledge; Hans Beens (Staatsbosbeheer Kop van Drenthe) for providing access to private roads to be able to reach the field sites; Teun Fiers, Klais Blaauw, Tom Harkema, Marte Stoorvogel, and Gibran Leeflang for field assistance; Mieke Hannink for administrative assistance with organising the fieldwork; Marcello Novani (Laboratory for Geo-information Science, WUR) for providing GNSS equipment; Jim Quik for construction work for core transport; Wouter van der Meer (BIAX Consult) for his help with opening and subsampling the cores; Harm Goorens for assistance with LOI analyses at the Soil Hydro Physics Lab (WUR) and Piet Peters for providing storage space in the cool room; Anne Roepert and colleagues of the CBLB for LOI analyses; Romy Koudijs for sharing OM data on the five duplicate cores; and Bas van Geel for his kind introduction to analysis of plant macrofossils. We thank Julie Loisel and an anonymous reviewer for their constructive comments to an earlier version of this manuscript; their feedback was much appreciated.

# Chapter 4

Faded landscape:

Unravelling peat initiation and lateral expansion at one of NW-Europe's largest bog remnants

> Cindy Quik, Ype van der Velde, Jasper Candel, Luc Steinbuch, Roy van Beek, Jakob Wallinga (2023)

> > *Biogeosciences*, 20, 695–718: doi.org/10.5194/bg-20-695-2023

Dataset at 4TU.Centre for Research Data: doi.org/10.4121/16923358

#### Abstract

In the mainland of Northwest Europe generally only remnants of former peat landscapes subsist. Due to the poor preservation of these landscapes, alternative approaches to reconstruct peat initiation and lateral expansion are needed compared to regions with intact peat cover. Here we aim (1) to find explanatory variables within a digital soil mapping approach that allow us to reconstruct the pattern of peat initiation and lateral expansion within (and potentially beyond) peat remnants, and (2) to reconstruct peat initiation ages and lateral expansion for one of the largest bog remnants of the Northwest European mainland, the Fochteloërveen. Basal radiocarbon dates were obtained from the peat remnant, which formed the basis for subsequent analyses. We investigated the relationship between peat initiation age and three potential covariates: (1) total thickness of organic deposits, (2) elevation of the Pleistocene mineral surface that underlies the organic deposits, and (3) a constructed variable representing groundwater-fed wetness based on elevation of the mineral surface and current hydraulic head. Significant relationships were found with covariate (1) and (3), which were hence used for subsequent modelling. Our results indicate simultaneous peat initiation at several loci in the Fochteloërveen during the Early Holocene, and continuous lateral expansion until 900 cal y BP. Lateral expansion accelerated between 5,500 -3,500 cal y BP. Our approach is spatially explicit (i.e., results in a map of peat initiation ages), and allows for a quantitative evaluation of the prediction using the standard deviation and comparison of predictions with validation points. The applied method based on covariate (1) is only useful where remnant peat survived, whereas covariate (3) may ultimately be applied to reconstruct peat initiation ages and lateral peatland expansion beyond the limits of peat remnants.

# 4.1 Introduction

Peat initiation and subsequent lateral expansion of peatlands represent a significant change in the palaeoenvironment. Knowledge on the timing, process rates and spatial dynamics of peat initiation and expansion is essential to develop our understanding of peatland functioning and development, carbon dynamics, climate change, and long-term human-landscape interactions in peatland environments (e.g. Van der Velde *et al.*, 2021; Tolonen and Turunen, 1996; Van Beek *et al.*, 2015; Van Beek, 2015; Chapman *et al.*, 2013).

Peat initiation may result from terrestrialisation (also called infilling), paludification, or primary mire formation (Charman, 2002c; Rydin and Jeglum, 2013d). Peat deposits form if the decay rate of biomass is slower than the rate of production, i.e. where there is a positive production-decay balance. The decay rate of organic material is mainly influenced by the degree of moisture (Charman, 2002b), which is dependent on a range of factors, including climate (e.g. Weckström *et al.*, 2010), changes in hydrological base level (resulting from sea level rise, e.g. Berendsen *et al.*, 2007; or regional groundwater changes, e.g. Van Asselen *et al.*, 2017), impermeable deposits or resistant layers in the soil profile (e.g. Breuning-Madsen *et al.*, 2018; Van der Meij *et al.*, 2018), landforms and surface topography (e.g. Almquist-Jacobson and Foster, 1995; Mäkilä, 1997; Loisel *et al.*, 2013), and anthropogenic influence (e.g. Moore, 1975, 1993).

Peat initiation can be studied at landscape scale and local scale (fig. 4.1a; Chapter 3). Landscape scale peat initiation refers to the development of peat at a certain locus, i.e. the oldest core of a peatland, that subsequently expands laterally and covers an increasing surface area. At local scale, peat initiation refers to the accumulation of the first organic deposits at this particular site, irrespective of its landscape position, i.e. the site could either be a development locus or become covered with peat through lateral expansion of one or more nearby loci.

Studies on the spatio-temporal dynamics of peat initiation and lateral expansion of peatlands appear to have focused mostly on boreal and circum-arctic peatlands, e.g. in Scandinavia (Mäkilä and Moisanen, 2007; Edvardsson *et al.*, 2014), Siberia (Peregon *et al.*, 2009), Canada (Bauer *et al.*, 2003), and Alaska (Loisel *et al.*, 2013; Jones and Yu, 2010). During the past decades several supra-regional to global syntheses were published that describe large-scale trends (Ruppel *et al.*, 2013; Korhola *et al.*, 2010; Macdonald *et al.*, 2006; Crawford *et al.*, 2003; Morris *et al.*, 2018). So far, limited attention has been paid to the palaeogeographical development of the former extensive peat landscapes of the Northwest European mainland (for an indication of their former extent see, e.g. Vos *et al.*, 2020; Casparie, 1993), with the exception of coastal and alluvial peatlands in the Rhine-Meuse delta (e.g. Berendsen and Stouthamer, 2000, 2001; Hijma, 2009; Cohen *et al.*, 2014; Pierik *et al.*, 2017). This is partly due to their large-scale disappearance following reclamation activities in the past few centuries (e.g. Gerding, 1995), leaving only small peat remnants behind in the current landscape. These remnants are under increasing threat of drainage (e.g. Swindles *et al.*, 2019), pollution (e.g. Limpens, 2003), and locally continuing excavation. The exploitation of their scientific potential is therefore of poignant urgency.

In the range of studies where (boreal) peatland initiation and long-term lateral development are studied, methodologies can roughly be divided in three approaches. In the first category, lateral expansion rates are deduced using transects of basal dates and distance between dating points, but the palaeogeographical pattern of lateral development is not visualised (e.g. Almquist-Jacobson and Foster, 1995; Turunen et al., 2002a; Anderson et al., 2003; Turunen and Turunen, 2003; Peregon et al., 2009; Robichaud and Bégin, 2009; Weckström et al., 2010; Loisel et al., 2013; Zhao et al., 2014). In the second category, transects of basal dates are manually converted to isochrones, i.e. lines of equal age that are deduced from the spatial distribution of obtained ages (fig. 4.1b). The isochrones visualise the pattern and rate of lateral development (e.g. Bauer et al., 2003; Edvardsson et al., 2014; Foster et al., 1988; Korhola, 1994, 1996; Mäkilä, 1997; Mäkilä and Moisanen, 2007). As a third category, numerical peat growth models can be distinguished that are based on hydrological and ecohydrological feedbacks. These simulate vertical peat growth (age-depth) for a peat column (e.g. the Holocene Peat Model, Frolking et al., 2010; the DigiBog model, Baird et al., 2012 and Morris *et al.*, 2012; the coupled DigiBog-STREAM model, Swinnen *et al.*, 2021). However, models that include lateral expansion are so far unavailable (see e.g. discussion on peat models by Baird *et al.*, 2012).

The use of transects of basal dates across a peatland is generally applied in areas where the natural extent of the peatland(s) under study is still intact. In regions where large areas of peatland have disappeared, the placing of such transects is questionable as the orientation of peat remnants within the former extensive peat landscape is unknown (fig. 4.1c). Additionally, peat-cutting (and ongoing excavation) may have damaged basal peat layers. Consequently, an adapted strategy is needed to collect (field) data from peat remnants. The number of studies that focus on peat remnants appears to be very low compared to studies of peatlands of which the extent is still intact (with some exceptions, e.g. the studies by Chapman *et al.*, 2013 and Crushell *et al.*, 2008).

Here we aim (1) to find explanatory variables within a digital soil mapping approach that allow us to reconstruct the pattern of peat initiation and lateral expansion within (and potentially beyond) peat remnants, and (2) to reconstruct peat initiation ages and lateral expansion for one of the largest bog remnants of the Northwest European mainland, the Fochteloërveen, in a former peat landscape

of which the majority has been lost during the past centuries. The elevation of the Pleistocene surface relative to its surroundings is likely the primary control on the moment of peat initiation at Fochteloërveen: the lowest points in a region tend to grow over by peat first. However, as a secondary control, large-scale geomorphology may influence local wetness: locations that are situated relatively far from a draining river tend to have higher groundwater tables that come closer to the surface, than locations adjacent to a river. Based on the hypothesis of a twofold control of peat initiation as discussed above, we investigate the relationship between peat initiation age and three potential covariates: (1) total thickness of organic deposits, (2) elevation of the Pleistocene mineral surface that underlies the organic deposits, and (3) a constructed variable based on elevation of the mineral surface and hydraulic head. Covariate (1) is only useful where remnant peat survived, whereas covariates (2) and (3) are potentially useful for reconstructing peat initiation age beyond peat remnants. The Holocene period up to 900 cal y BP (i.e., approximate start of the High Middle Ages in the study area) forms the temporal scope for our study, because of increased human influence from then onwards (see section 4.2.2).

## 4.2 Study area

#### 4.2.1 Selection and description of study area

The Fochteloërveen peatland in the Netherlands (fig. 4.2) was chosen as case study area. The Fochteloërveen ( $\sim$  2,500 ha) is the largest Dutch bog reserve (Joosten et al., 2017a), and one of the largest bog remnants in the Northwest European mainland. It was part of an extensive peat landscape (fig. 4.2c) and is protected as a Natura 2000 area (Provincie Drenthe, 2016). The widespread occurrence of its mineral substrate and characteristic climatic conditions (see below) make this peatland area representative for larger parts of the Northwest European mainland. Because of the availability of earlier obtained radiocarbon dating evidence (Quik et al., 2022a) and detailed subsurface data from national databases, we consider the Fochteloërveen as an ideal case study to investigate temperate peatland development. In addition, background information on peat initiation trends in the wider region is available from a recent study based on a large set of legacy radiocarbon dates (Quik et al., 2021). Various important archaeological finds have been done in the vicinity of Fochteloërveen, including a Mesolithic aurochs butchering site (Prummel and Niekus, 2011), wooden trackways from the Iron Age (Casparie, 1985), and a Roman-period settlement site that is assumed to have been deserted due to rising groundwater levels (Van Giffen, 1958).

In the north of the Netherlands a continental ice sheet was present during



**Figure 4.1:** Conceptual illustration of peat initiation and lateral expansion of peatlands. (a) Schematic top-view of a landscape (peat is indicated in brown), showing the meaning of peat initiation at the landscape and local scale (redrawn from Chapter 3). (b) Approach to reconstruct peat initiation and the pattern of lateral expansion using dating transects and isochrones for an intact peatland. (c) Situation when only remnants of the former peat landscape remain.

the Saalian (MIS 6). This led to deposition of glacial till (Rappol, 1987; Rappol *et al.*, 1989; Van den Berg and Beets, 1987; TNO – Geological Survey of the Netherlands, 2021b), on the Drenthe Plateau (Bosch, 1990; Ter Wee, 1972). Deposition of aeolian coversands over Northwest Europe during the Weichselian (OIS 4-2) resulted in formation of the European Sand Belt (Koster, 1988, 2005). Coversands occur with a thickness of approximately 0.5 - 2 m on the Drenthe Plateau (TNO – Geological Survey of the Netherlands, 2021c; Ter Wee, 1979). Fochteloërveen is located close to the western edge of the Drenthe Plateau. Below the coversands a discontinuous till layer with a thickness up to 3.5 m is present underneath the peat remnant (Provincie Drenthe, 2022). Fochteloërveen is part of three catchments (fig. 4.2b). Currently average temperatures are  $2.8^{\circ}$  C in January and  $17.5^{\circ}$  C in July, average annual rainfall amounts to 805 mm, and the potential evapotranspiration is 566 mm (KNMI, 2021). Throughout the paper, we indicate elevation in metres O.D., i.e. relative to Dutch Ordnance Datum

(+NAP), which is roughly equal to mean sea level.

Fochteloërveen does not fit within a single definition or classification as it probably formed through coalescence of multiple smaller mires (see results in section 4.4) that formed on a non-coastal and non-alluvial topographic plain. However, in the hydromorphological classification (cf. Charman, 2002c), the resultant composite peatland can probably best be described as a plateau-raised bog. Fochteloërveen started off as a fen (minerogenous mire), but later on transitioned to a bog (ombrotrophic mire) (more information below; Quik *et al.*, 2022a).

Biostratigraphical analyses by Quik *et al.* (2022a) show that the vegetation was mesotrophic during peat initiation. This vegetation was dominated by sedges, with some presence of *Juncus*. Water tables fluctuated during the peat initiation process and wetter and drier conditions probably alternated. Wildfires must have occurred regularly, as indicated by the frequent presence of charred plant remains. Conditions became more oligotrophic after the peat initiation process. The vegetation developed probably to an oligotrophic bog with *Calluna vulgaris, Erica tetralix* and *Sphagnum* at two of the locations studied by Quik *et al.* (2022a), and to a moss (*Bryales*) and heather vegetation at the third studied location.

The vegetation of Fochteloërveen is currently dominated by *Sphagnum* mosses occurring in a hummock and hollow topography (Provincie Drenthe, 2016). Species include amongst others *S. magellanicum*, *S. papillosum*, *S. rubellum*, and vascular species typical for ombrotrophic conditions such as *Eriophorum vaginatum*, *Andromeda polifolia* and *Vaccinium oxycoccos*. Fochteloërveen harbours several protected animal species, including the common ringlet (*Coenonympha tullia*), subarctic darner (*Aeshna subarctica*), smooth snake (*Coronella austriaca*), common European adder (*Vipera berus*) and a wide range of bird species. From 2001 onwards, crane birds (*Grus grus*) settled in the area (Provincie Drenthe, 2016). Since the 1980s nature conservation is directed at peatland restoration (Altenburg *et al.*, 2017; Provincie Drenthe, 2016). Main threats for nature conservation include atmospheric nitrogen deposition and desiccation due to intense drainage for surrounding agriculture.



**Figure 4.2:** (Figure on previous page). (a) Location of the Fochteloërveen peat remnant within Europe (ESRI, 2022); (b) Digital Elevation Model (DEM) of the wider area around Fochteloërveen (version AHN3, horizontal resolution 5 m, vertical resolution 0.1 m; AHN, 2021a,b), indicating the main drainage pattern (Ministerie van Verkeer en Waterstaat, 2007). Coordinates are in metres (Dutch RD-new [Rijksdriehoeksstelsel] projection). Extent of the Fochteloërveen Natura 2000 area is indicated (Ministerie van Economische Zaken - Directie Natuur & Biodiversiteit, 2018); (c) Palaeogeographical map of Fochteloërveen and surroundings, showing reconstructed situation for 500 BCE (2,450 cal y BP) as indicated by the Dutch national palaeogeographical map series (Vos et al., 2020; RCE, 2022); (d) Topographical map of Fochteloërveen (OpenTopo; Van Aalst, 2022), showing coring locations (see sections 4.3.1.1 and 4.3.1.2) and position of archaeological validation points (for details see table 4.2). Peatland is indicated with purple colours. The area surrounding Fochteloërveen shows the landscape structure that resulted from historical peat colonies, peat-cutting activities and agricultural reclamation.

#### 4.2.2 Peatland development and decline in the (wider) study area

There is a hiatus between the deposition of coversand during the Weichselian and the formation of peat in the coversand areas in the eastern half of the Netherlands. This can be deduced from the occurrence of soil profiles in coversand (e.g. podzols) underneath the peat, and sometimes by the presence of bog wood (i.e. evidence of previous vegetation cover; Staring, 1983; Jongmans *et al.*, 2013). Theories deviate on the timing when peat growth started in the coversand landscape, on the period when these peatlands expanded, and when they reached their maximum extent (fig. 4.3). Note that in the text below, we repeat the 'old' chronostratigraphic terms that were used in the cited papers. We have added cal y BP ages to ease interpretation and comparison with the new formal subdivision of the Holocene (Walker *et al.*, 2019).

The national palaeogeographic maps created by Zagwijn (1986) indicate that peat formation started during the Early Atlantic ( $\sim$ 7,450 cal y BP). By the Late Atlantic ( $\sim$ 6,050 cal y BP) large raised bog complexes had formed and reached their maximum extent. From then onwards they remained laterally stable. Zagwijn (1986) placed the reaching of maximum extent in the Late Atlantic, but indicates that the true timing remains uncertain due to lack of data as a result of peat-cutting.

According to the recent palaeogeographic map series by Vos *et al.* (2020), peat initiation also started during the beginning of the Atlantic (~7,450 cal y BP). However, the peatlands continued to expand gradually during the Atlantic and Subboreal, and reached their maximum extent at the beginning of the Subatlantic (~2,450 cal y BP). From ~1500 CE (~450 cal y BP) onwards the peat-covered area rapidly declines according to the map series of Vos *et al.* (2020) due to agricultural reclamation and peat-cutting activities. The national palaeogeographic maps are

based on elaborate and detailed data on the development of river deltas and coastal areas in the Netherlands (Vos, 2015b). In contrast, the reconstructions of peatlands in the coversand region have relatively large uncertainty as the amount of data on these areas is low (Vos, 2015a; Van Beek, 2009; Spek, 2004).

Few studies with regional palaeogeographical focus are available for Fochteloërveen and surroundings, but important exceptions include the work of Fokkens (1998) and Waterbolk (2007), who deduce patterns of peatland development based on archaeological find distributions. In doing so, Waterbolk (2007) deduces presence of peat through the absence of archaeological finds, whereas Fokkens (1998) considers an approach using archaeological finds only as terminus-post-quem dates for peat initiation most appropriate (i.e., absence of finds is not used as an indication for presence of peat).

Waterbolk (2007) deduces that large areas were already covered by peat during the Early and Middle Neolithic (4,900 – 2,850 BCE; 6,850 – 4,800 cal y BP). This situation remained stable for several thousand years. Rapid peatland expansion during the Iron Age, which Waterbolk (2007) linked to climate change as discussed by Van Geel *et al.* (1998), left the majority of the area uninhabited during the Roman Period (19 BCE – 450 CE; 1,969 – 1,500 cal y BP). This is in contrast to the conclusions of Fokkens (1998), who assumed that (oligotrophic) peat on the plateau was largely absent during the Middle Neolithic (5,000 cal y BP), except for very local sites.

Until recently, radiocarbon dating evidence from a (near) basal peat layer was only available for a single site in Fochteloërveen (Klaver, 1981; later published by Van Geel *et al.*, 1998), which indicated an age of 2,920 - 2,736 cal y BP (at 95.4% confidence interval). Fokkens (1998) mentions that this site represents the nucleus of the area, and consequently places peat initiation in the Fochteloërveen area between the beginning of the Early Iron Age and the end of the Roman period (800 BCE – 400 CE; 2,750 - 1,550 cal y BP). He also assumes that the maximum extent was reached during this period. Waterbolk (2007) assumes that the peatland reached its maximum extent slightly later during the Early Middle Ages (450 - 1000 CE; 1500 - 950 cal y BP), prior to the onset of systematic reclamation activities.

New dating evidence at three sites presented by Quik *et al.* (2022a) indicates that within the boundaries of the Fochteloërveen nature reserve peat developed from ~9,000 cal y BP onwards and that new areas became covered with peat at least until ~3,500 cal y BP, suggesting that landscape scale peat initiation occurred much earlier than suggested in the studies mentioned above (Vos *et al.*, 2020; Fokkens, 1998; Waterbolk, 2007; Zagwijn, 1986). The substantial differences between the studies discussed above and the overview presented in fig. 4.3 highlights the need to better constrain the timing of peat initiation and the pe-



**Figure 4.3:** Timing of peat initiation and period of lateral development in the Dutch coversand landscape (upper two) and at Fochteloërveen specifically (lower three), according to different studies. The x-axis is equal to fig. 4.10, to ease comparison with the outcomes of the present study.

riod of lateral development in the coversand landscape and at Fochteloërveen specifically.

Even though local peat cutting (i.e., on a household level) took place since the Middle Ages, large-scale reclamations of the Fochteloërveen area only started in the 17<sup>th</sup> century (Gerding, 1995; Douwes and Straathof, 2019). This happened mainly for turf production, and lasted (in the last decades on a smaller scale) until the 1970s. In a large part of the area superficial peat layers are affected by buckwheat cultivation, which was at its height in the late 18<sup>th</sup> and 19<sup>th</sup> century (Douwes and Straathof, 2019). However, this practice did not affect the basal peat layers that are of interest to our study. Data from the 18<sup>th</sup> century indicate that peat thickness at Fochteloërveen has locally declined with as much as 7 m during the past three centuries (Douwes and Straathof, 2019).

# 4.3 Methods

# 4.3.1 Methods Part 1: Collection of field data and radiocarbon dating evidence

#### 4.3.1.1 Field approach and site selection

Our field approach is directed at obtaining an extensive set of basal radiocarbon dates, which forms the basis for subsequent modelling steps. Testing the influence of the elevation of the Pleistocene surface and large-scale geomorphology requires insight both in vertical and horizontal landscape dimensions within the Fochteloërveen peat remnant. Transects of basal peat dates are generally used for reconstructions of peat initiation and lateral development (fig. 4.2b). In sea level research, where focus lies on the vertical dimension, it is custom to date basal peat samples that overlie compaction-free sediments where the groundwater level that steered peat growth can be related to former sea level (e.g. Törnqvist *et al.*, 1998). To obtain the required insight in vertical and horizontal landscape dimensions at Fochteloërveen, we chose a hybrid approach that combines spatially distributed transects with elevation gradients in the (compaction-free) mineral deposits underlying the organic deposits (also see Chapman *et al.*, 2013).

Our field exploration consisted of 93 gouge corings, mostly grouped in transects of 185 to 575 m long that were placed perpendicular to the elevation gradients of subsurface coversand ridges and depressions (fig. 4.2d and fig. 4.4). At a few sites a central gouge coring was surrounded by four corings in a radial pattern to derive the subsurface topography. For each core the stratigraphy was described (see Quik *et al.*, 2022a for details). After the field exploration 21 sites were selected for sampling, taking the distribution over the study area into account. Additionally, it was ensured that the combination of sample sites stretched the elevation range of the mineral surface underlying the organic deposits (samples cover an elevation range of 7.2 - 11.7 m O.D.). Collection and subsampling of cores, (bio)stratigraphical analyses, selection of dating samples, and radiocarbon dating procedures followed Quik *et al.* (2022a) and are summarized in sections 4.3.1.2 - 4.3.1.4.

#### 4.3.1.2 Collection and subsampling of cores

Cores were collected in 2019 with a hand-operated stainless-steel peat corer (Russian type) with a core volume of 0.5 dm<sup>3</sup> (Eijkelkamp Soil & Water, 2018). Prior to sampling the corer was cleaned with deionised water. After retrieving the core it was carefully packaged in a PVC half-pipe and placed in a refrigerator of 3° C within 12 hours. Location and elevation of all sampling sites were recorded with a Topcon 250 Global Navigation Satellite System (GNSS) receiver, with a horizontal

precision of ~5 mm and vertical precision of ~10 mm (RTK; TOPCON, 2017).

Directly after opening the cores the stratigraphy was described (additional to field descriptions) and the approximate mineral-to-peat transition (following Quik *et al.*, 2022a) was determined through visual inspection. Around this transition 6 to 12 contiguous 1-cm-thick slices were cut from the core. Outer edges of these slices were carefully cleaned to avoid contamination. From each slice a subsample of ~2 cm<sup>3</sup> was collected for loss-on-ignition; all remaining material (~5 cm<sup>3</sup>) was reserved to select plant macrofossils for radiocarbon dating after subsample selection (see section 4.3.1.3).

#### 4.3.1.3 (Bio)stratigraphical analyses and selection of dating samples

The organic matter (OM) content was measured using loss-on-ignition. Sample dry weight was determined after drying for 24 hours at 105° C, followed by combustion at 550° C. After obtaining the OM% for the range of slices cut from a particular core, the lowermost slice that contained >40% OM was defined as the basal peat layer (following Quik et al., 2022a). Charred and uncharred plant macrofossils (aboveground tissues) were selected at BIAX Consult in Zaandam, the Netherlands for radiocarbon dating (table 4.1). For 12 cores radiocarbon samples were obtained from the basal peat slice. These basal dates form the calibration dataset for subsequent modelling Quik et al., 2022a. In addition, five radiocarbon samples were obtained from slices of five cores that represent terminus-ante-quem (TAQ) and terminus-post-quem (TPQ) dates for peat initiation (i.e., from slices with >40% OM and slices with <40% OM respectively). The TAQ and TPQ dates are part of our validation dataset (together with archaeological validation data, see section 4.3.2.2. A table with organic matter gradients on which selection of dating samples is based, and a table with all encountered plant macrofossils (i.e., including material not selected for dating) are available online (see reference to the dataset on page 105). For three additional cores, dating information of the basal peat layer is available from Quik et al. (2022a,b).

#### 4.3.1.4 Radiocarbon dating

Radiocarbon measurements were performed at the Centre for Isotope Research of the University of Groningen (the Netherlands). A full description of the methods used at this laboratory can be found in Dee *et al.* (2020); methods applied to our samples are only concisely explained below. Samples were either pre-treated using the acid-base-acid (n = 9) method, or only with acid (n = 5), or not pre-treated in case of very small and delicate samples (n = 3) (see reference to the dataset on page 105 for details per sample). Samples were measured using a MICADAS Accelerator Mass Spectrometer (lonplus AG; Synal *et al.*, 2007).

Depending on sample weight after the pre-treatment, samples were measured as graphite in a regular batch (in case of 1.0 - 2.5 mg C), in a batch for small-sized graphite samples (0.1 - 1.0 mg C), or measured directly as CO<sub>2</sub> after combustion (< 0.15 mg C). F<sup>14</sup>C and <sup>14</sup>C age were calculated according to the conventions (Stuiver and Polach, 1977). Results are corrected for isotopic fractionation using the  $\delta^{13}$ C value measured with AMS. Dates were calibrated using IntCal20 (Reimer *et al.*, 2020) in the OxCal program (version 4.4, Ramsey, 1995).

Core	Subsample	From (m O.D.)	To (m O.D.)	Aboveground plant remains for <sup>14</sup> C dating (charred unless indicated otherwise)
S1	M4	10.53	10.54	<i>Calluna vulgaris</i> (stem 2 fragments [charred, very small fragments])
S1	M5	10.54	10.55	Ericaceae (twig fragments 3); <i>Erica tetralix</i> (leaf 1)
S2	M9	10.79	10.80	<i>Erica tetralix</i> (leaf 1, leaf 1 [un-charred]); <i>Calluna/Erica</i> (twig fragments $+$ )
S3	M2	10.12	10.13	<i>Eriophorum vaginatum</i> (leaf and stem fragments with sclerenchyma tissue and epidermis ++, stem base (corm) with spindles 1, isolated spindles ++ [all un-charred])
S4	M9	7.16	7.17	<i>Carex</i> cf. <i>riparia</i> (5 [un-charred]), <i>Carex</i> cf. <i>pilulifera</i> (6 fragments [un-charred]; Ericaceae (twig 1 fragment); undetermined (herbaceous stem 3 fragments)
S5	M5	10.13	10.14	<i>Erica tetralix</i> (leaves 7); cf. <i>Erica</i> (twig fragments +)
S6	M8	9.74	9.75	<i>Erica tetralix</i> (leaves 8); Ericaceae (twig 3 fragments); <i>Carex</i> (3 fragments [un-charred])
S7	M7	10.55	10.56	<i>Eriophorum vaginatum</i> (spindle 1); undetermined (herbaceous stem 1 fragment)
S7	M8_9	10.56	10.58	Cf. Ericaceae (stem base/root 2 fragments); <i>Persicaria</i> (1 fragment [un-charred]); undetermined (herbaceous stem 2 fragments);
S8	M3	11.55	11.56	<i>Erica tetralix</i> (leaves 2); cf. <i>Erica</i> (flower 1); <i>Andromeda polifolia</i> (seed 1 [un-charred]); undetermined (herbaceous stem fragments $+$ )
S9	M1	10.59	10.60	NA
S9	M2	10.60	10.61	<i>Reseda luteola</i> (seed 1 [un-charred]); undetermined (herba- ceous stem 2 fragments)
S10	M6	10.78	10.79	Eriophorum vaginatum (spindles 3); Ericaceae (stem base 1); undetermined (herbaceous stem fragments $+$ )
S12	M7	10.91	10.92	Ericaceae (stem base 4 fragments); Sphagnum (stem 1 fragment [un-charred]); undetermined (herbaceous stem fragments $+$ )

**Table 4.1:** Overview of the aboveground plant remains that were selected for radiocarbon dating. When a number is given, this is the exact amount encountered, cf. = resembles, + = present, ++ = frequent, NA = not available.

Continued on next page

Fable 4.1 –	Continued
-------------	-----------

Core	Subsample	From (m O.D.)	To (m O.D.)	Aboveground plant remains for <sup>14</sup> C dating (charred unless indicated otherwise)
S13	M10	8.99	9.00	Calluna vulgaris (twig fragments ++, leaves +); Erica tetralix (leaves 4, twig fragments +); undetermined (herba- ceous stem fragments +)
S14	M10	8.65	8.66	Calluna vulgaris (twigs with leaves ++, stem 1 fragment [un-charred]); Erica tetralix (leaves ++); Calluna/Erica (flowers +); Sphagnum (stem 1 fragment)
S15	M5	8.13	8.14	Erica tetralix (leaves 3, stem base 1); undetermined (herbaceous stem fragments $+$ )
S16	M2	7.57	7.58	Undetermined (herbaceous stem fragments +)
S16	M4	7.59	7.60	NA
S19	M8	10.44	10.45	Cyperaceae (stem 2 fragments); undetermined (herbaceous stem fragments $+$ )

#### 4.3.2 Methods Part 2: Reconstructing peat initiation age spatially

#### 4.3.2.1 Covariates and construction of covariate maps

The relationship with median peat initiation age was tested for: (1) the total thickness of organic deposits [O], (2) the elevation of the Pleistocene mineral surface  $[z_P]$  underlying the organic deposits, and (3) a constructed variable based on elevation of the mineral surface and hydraulic head, that captures the effect of groundwater-fed wetness that results from geomorphological position (fig. 4.5). The latter covariate, denoted with  $z_PH$ , is defined as the peat initiation height (i.e., the elevation of the Pleistocene mineral surface,  $z_P$ ) at location x minus the current hydraulic head ( $H_{t0}$ ) at location x.

Two Dutch national databases with subsurface data, managed by the Dutch Geological Survey (TNO-GDN), were consulted (see below) for the construction of the covariate maps. Geological coring data for a region surrounding Fochteloërveen (see extent of corings in fig. 4.4a) were downloaded from Dl-NOloket (DINOloket - TNO, 2022) and selected for presence of peat or gyttja (n = 485). In addition, information from gouge corings (n = 71) and sample corings (n = 21) from the fieldwork (see section 4.3.1.2) was used. For each coring (total n = 577), the elevation of the Pleistocene surface [ $z_P$ ] and total thickness of organic deposits [O] was registered (for geological corings the sum was used of peat and gyttja if present). Using ArcGIS Pro (version 2.3.3), a palaeoDEM of the Pleistocene surface was interpolated based on the  $z_P$  values through Inverse Distance Weighing (IDW). In addition, a map of the total thickness of organic deposits was interpolated through IDW based on the O values. The resulting rasters have a resolution of 50 x 50 m and a support of ~10 datapoints/km<sup>2</sup>.



**Figure 4.4:** Two cross sections showing example transects perpendicular to coversand ridges underlying the organic deposits in the Fochteloërveen peat remnant. See fig. 4.2 for location of the transects; (a) transect covering one side of a ridge, and (b) transect covering both sides of a ridge. Corings are indicated with vertical bars at the top of each cross section. The numbers (of the format Sxx) refer to the sampling site codes (see table 4.1 and table 4.3).

For the current hydraulic head  $[H_{xt0}]$  the map of the upper layer of the national hydrological model was downloaded from Grondwatertools (Grondwatertools - TNO, 2022), reflecting the phreatic groundwater level (the upper unconfined aquifer). This raster has a resolution of 250 x 250 m. To enable maps of peat initiation age with a 50 x 50 m resolution, the  $H_{xt0}$  raster was resampled to 50 x 50 m and smoothed through focal statistics (using the mean with a neighbourhood of 3 x 3 cells). A  $z_PH$  raster was calculated by subtracting the  $H_{xt0}$  raster from the  $z_P$  palaeoDEM. For each coring location (n = 577) the value of  $z_PH$  was obtained to be used in linear regression analysis (see section 4.3.2.2). Covariate maps and a table with all used coring data are available online (see reference to the dataset on page 105).

#### 4.3.2.2 Linear regressions and prediction maps of peat initiation age

Using R, the relationships between median peat initiation age and each of the three covariates  $[O, z_P, z_PH]$  were analysed with linear regression. Based on several checks (including the normality of the residuals, homoscedasticity, and leverage) the assumptions underlying linear regression were deemed valid for our data (results not shown but available through the R script, (see reference to the dataset on page 105). The three linear models were assessed based on their p-value and adjusted R<sup>2</sup>; those with significant results were used for subsequent predictions of peat initiation age.

Using the linear models, covariate maps were converted to prediction maps of peat initiation age using the R "raster" package. The corresponding standard deviation maps were obtained from the limits of the prediction interval of the regressions. The resulting rasters were exported as geotiffs and opened in ArcGIS Pro (2.3.3). Values <900 cal y BP were set to 'No Data'. Isochrones (contour lines) with an interval of 1000 years were added to the map to show the pattern of modelled peatland expansion.

A histogram and density function of the predicted peat initiation ages were created to visualize the acceleration in lateral expansion through time. A cumulative density function was created to show the increase in peat covered area through time within the boundaries of the Fochteloërveen peat remnant.

#### 4.3.2.3 Assessment of modelling results

To evaluate the agreement between the predictions of peat initiation age, the resulting prediction rasters were subtracted from each other. In addition, predictions were evaluated using validation points consisting of radiocarbon dates of peat samples that indicate a terminus-ante-quem (n = 3) and terminus-post-quem



**Figure 4.5:** Schematic depiction of the covariates  $[O, z_P, z_PH]$ . For explanation see text.

(n = 2) for peat initiation, and archaeological sites that indicate a terminus-postquem for peat initiation (n = 4; fig. 4.2d and table 4.2). Predicted peat initiation ages at the locations of validation points were obtained and compared with the TPQ/TAQ information of these points. To ease interpretations the predicted ages and TPQ/TAQ validation ages were plotted in OxCal (version 4.4, Bronk Ramsey, 1995). Confidence intervals of the validation points are presented both as oneand two-sigma intervals. Table 4.2: Archaeological validation points within the study area. Identification numbers can be used to obtain detailed find information is not entirely certain, coordinates are only approximate. TPQ = terminus-post-quem, TAQ = terminus-ante-quem. Ages in BCE are from Archis III (RCE, 2019). Coordinates are in metres (Dutch RD-new [Rijksdriehoeksstelsel] projection). As the exact find location indicated with a minus sign.

Identification Archis III	RD-x	RD-y	Lo cation, t <i>op onym</i>	Concise description	Interpretation	Constraint for peat initiation	Archaeological period	From (BCE/CE)	To (BCE/CE)	From (cal y BP)	To (cal y BP)
2929700100	221450	556300	Appelscha	Late Palaeolithic and Mesolithic flint material.	Habitation was possible during the Mesolithic (imprecise date).	ТРQ	Late Palaeolithic to Mesolithic	-15000	-5000	16950	6950
3010261100	222500	557800	Fochteloo, De Bonghaar	Cluster of Mesolithic flint material.	Habitation was possible during the Mesolithic (imprecise date).	ТРQ	Mesolithic	-8000	-5000	9950	6950
2929758100	224400	554500	Ravenswoud	Cluster of Late Neolithic flint artefacts.	Habitation was possible during the Late Neolithic.	ТРQ	Late Neolithic	-2850	-2000	4800	3950
2688418100	228000	558270	Norgervaart, Norgervaart 09	Single find of a silver coin dated to the first half of the 9th century.	Human activity in or after 9th century. Unclear if this find points to habitation.	۲.	Early Middle Ages	814	840	1136	1110
3030577100	221450	560500	Veenhuizen, <i>De</i> Bieuw	Find assemblage of high medieval pottery.	Habitation was possible during the High Middle Ages.	ТРQ	High Middle Ages	006	1300	1050	650

# 4.4 Results

## 4.4.1 Results Part 1: Collection of field data and radiocarbon dating evidence

A table with all collected coring data is available online (see reference to the dataset on page 105). The dating results for the obtained samples from the cores are listed in table 4.3. Both peat initiation dates and terminus-post-quem/terminus-ante-quem dates for validation are included. Ages range from 1650  $\pm$  40 yrBP (1690 – 1411 cal y BP at 95.4% confidence interval) to 8305  $\pm$  30 yrBP (9433 – 9142 cal y BP at 95.4%), indicating that the period of peat initiation and subsequent lateral expansion stretched over at least ~7.500 calendar years.

#### 4.4.2 Results Part 2: Reconstructing peat initiation age spatially

The linear regressions of median peat initiation age versus the total thickness of organic deposits [O] is highly significant (p-value < 1e-3) and explains a reasonable amount of variation with an adjusted R<sup>2</sup> of 0.57 (fig. 4.6a). The linear regression of median peat initiation age versus the elevation of the Pleistocene mineral surface underlying the organic deposits  $[z_P]$  has a p-value of 0.09 and and adjusted R<sup>2</sup> of 0.14. Based on these values, this covariate was rejected for further analyses. For the third covariate that was tested, groundwater-fed wetness that results from geomorphological position  $[z_PH]$ , the linear regression has an adjusted R<sup>2</sup> of 0.61 and is again highly significant (p-value < 1e-3) (fig. 4.6b).

The interpolated Pleistocene surface underlying the organic deposits ([ $z_P$ ], fig. 4.7a) indicates that the mineral substrate covers an elevation range of 6 – 13 m O.D.. The thickness of organic deposits ([O], fig. 4.7b) varies from 0.0 – 3.5 m. Using the Pleistocene surface ([ $z_P$ ], fig. 4.7a) and present-day hydraulic head ([ $H_{t0}$ ], fig. 4.7c), the covariate raster for groundwater-fed wetness that results from geomorphological position ([ $z_PH$ ], fig. 4.7d) was calculated (see section 4.3.2.1).

Using the covariate maps in fig. 4.7b and 4.7d and the linear regressions in fig. 4.6, the prediction maps of peat initiation age in fig. 4.8 were generated. Standard deviations of the predictions are shown in insets in fig. 4.8. The higher the standard deviation at a particular location, the less certain the prediction of peat initiation age is for that point. Lower certainty mainly occurs at points with a predicted peat initiation age  $\geq$  6,000 cal y BP and at points with an age of  $\leq$  1,500 cal y BP, as the number of datapoints above and respectively below these ages are limited (see data points in fig. 4.7). Overall, both predictions in

fig. 4.8 show a similar pattern. The difference between both predictions as shown in fig. 4.9a, demonstrates that they deviate near the edge of the Fochteloërveen area and mostly in the northern part of the area. As this northern part is currently forested (see fig. 4.2d), present-day groundwater levels may not reflect a natural pattern here, causing deviations in the prediction based on  $z_P H$ . The validation points indicate that for the three terminus-ante-guem radiocarbon dates (fig. 4.9b), which should be younger than the predicted peat initiation ages, two are indeed younger than the mean of the predictions and one is slightly older but falls within the 1-sigma confidence interval of the predictions. Of the terminuspost-quem radiocarbon dates (fig. 4.9b), which are expected to be older than the predicted peat initiation ages, one has a comparable age as the mean of the predictions, and one is older than the mean of the predictions. Of the terminuspost-quem archaeological validation points (fig. 4.9c), three are indeed older. One is younger than the mean of the predictions but still falls within the 2-sigma confidence interval for the prediction based on O and within the 1-sigma confidence interval for the prediction based on  $z_P H$ . Overall, comparison with the validation points suggests validity of the predictions.

Loci of early peat initiation are distributed over the lower central, west and northwest parts of the Fochteloërveen area (fig. 4.8a and 4.8b), indicating that landscape scale peat initiation (see fig. 4.1a) occurred simultaneously at multiple sites. The west and northwest loci are located on low-lying topography ( $\leq$ 7.0 m O.D.) of the Pleistocene surface (compare with fig. 4.7a), but the loci in the lower central part of the area are located on somewhat higher ground (between 9.0 and 10.5 m O.D.). Even on the highest parts of the Pleistocene surface (between 12.0 and 13.0 m O.D.), both predictions indicate a peat cover from about 3,000 cal y BP onwards. This suggests that as the peat cover grew with time, even coversand ridges that initially protruded above the peat landscape became covered with peat as time progressed. The distance between the isochrones in fig. 4.8a and 4.8b indicates the rate of lateral expansion. Where isochrones are drawn close together, the peat cover expanded slowly. This is the case in the blueish coloured parts of the maps, pointing to initial slow expansion of peat initiation loci. Later in time the peat-covered area expanded more rapidly, with the strongest expansion between 5,500 - 3,500 cal y BP (see fig. 4.10c; 5,500 - 3,500 cal y BP for the prediction based on O, and 5,500 – 3,000 cal y BP for the prediction based on  $z_PH$ ). Half of the Fochteloërveen area was covered with peat by ~4,000 cal y BP according to the prediction based on O (fig. 4.10c), and by ~3,500 cal y BP according to the prediction based on  $z_P H$ . Peat covered nearly the entire area by  $\sim$ 2,500 and  $\sim$ 900 cal y BP respectively.



**Figure 4.6:** Linear regressions, showing the relationships between median peat initiation age and (a) the total thickness of organic deposits [O], and (b) groundwater-fed wetness that results from geomorphological position  $[z_PH]$ . Both regressions were significant with p < 1e-3. Observations are indicated by the black circles. The regression line is shown in black, the 95% confidence interval of the regression line is indicated in red, and the 95% prediction interval in blue. The range of the covariate maps is indicated in purple and also visible in fig. 4.7b and 4.7d.

**Figure 4.7:** (Figure on next page). (a) Interpolated elevation of the Pleistocene surface  $[z_P]$  based on data obtained through sample/gouge corings and DINOloket corings (DI-NOloket - TNO, 2022). (b) Interpolated current thickness of organic deposits [O] based on data obtained through sample/gouge corings and DINOloket corings (DINOloket - TNO, 2022). Radiocarbon dates of basal peat samples show peat initiation ages (see section 4.4.1). (c) Current hydraulic head  $[H_{t0}]$  based on the LHM model (Grondwatertools - TNO, 2022), resampled to 50 × 50 m and smoothed through focal statistics. (d) Groundwater-fed wetness that results from geomorphological position, calculated by subtracting the hydraulic head  $[H_{t0}]$  in (c) from the Pleistocene surface  $[z_P]$  in (a). For further details on each map see section 4.3.2.1.





**Figure 4.8:** (a/b) Reconstructed peat initiation ages for Fochteloërveen, with (a) based on thickness of organic deposits [O] (see fig. 4.7b), and (b) based on groundwater-fed wetness that results from geomorphological position  $[z_PH]$  (see fig. 4.7d). Contours represent isochrones (lines of equal age) and have an interval of 1000 years. Note that peat initiation age legends are equal, but that legends of the standard deviation of the prediction differ.



**Figure 4.9:** (Figure on previous page). (a) Difference between the mean predictions of peat initiation age in fig. 4.8a and 4.8b (i.e., prediction of fig. 4.8a minus prediction of fig. 4.8b). The locations of the validation points (of which results are presented in panels (b) and (c)) are also indicated on this map (for further details on the archaeological sites see table 4.2). In (b) and (c) validation points are compared with the predicted peat initiation age as predicted by both covariates (i.e., each validation point is compared with the prediction of fig. 4.8a and with the prediction of fig. 4.8b). The central circles show the mean, bars indicate 1-sigma confidence interval, and grey-coloured blocks the 2-sigma confidence interval. Note that terminus-ante-quem validation points should be younger (see section 4.4.2). (b) Validation points consisting of radiocarbon dates of peat samples that indicate a terminus-ante-quem (n = 3) and terminus-post-quem (n = 2) for peat initiation. (c) Validation points consisting of archaeological sites that indicate a terminus-post-quem for peat initiation (n = 4; for one archaeological site the relationship with peat growth is unclear, therefore this site was not included here, see table 4.2 for details).



**Figure 4.10:** (a) Histogram of predicted peat initiation ages for Fochteloërveen, showing results for the prediction based on thickness of organic deposits [O] (see fig. 4.8a), and for the prediction based on groundwater-fed wetness that results from geomorphological position  $[z_PH]$  (see fig. 4.8b). Start of the Holocene follows Walker et al. (2009). Temporal scope of the predictions runs to 900 cal y BP. Note that certainty of both predictions is different, see standard deviation maps in fig. 4.8a and fig. 4.8b. (b) Density plot of predicted peat initiation ages for Fochteloërveen (i.e., normalised to a graph surface area of one). (c) Plot showing cumulative fraction of peat-covered area within the Fochteloërveen area as indicated by the two predictions.

# 4.5 Discussion

Here we discuss peat initiation and lateral expansion at Fochteloërveen as indicated by our predictions (section 4.5.1), followed by an evaluation of our approach (section 4.5.2).

#### 4.5.1 Peat initiation and lateral expansion at Fochteloërveen

Peat initiation results from terrestrialisation, paludification and/or primary mire formation (Charman, 2002b; Rydin and Jeglum, 2013d). As the substrate in the study region has been deglaciated and exposed at the surface since the penultimate Glacial (OIS6, see section 4.2.1), primary mire formation is not the case here. The question then remains whether peat initiation resulted from paludification or terrestrialisation, or both. Gyttja is often found at the base of terrestrialisation sequences. We did not encounter gyttja in the gouge and sample corings of the field survey, but in part of the DINOloket coring descriptions gyttja is mentioned. However, interpretation and terminology of the amorphous peat layer (highly humified; sapric cf. IUSS Working Group WRB, 2015; see table 3.2 in Chapter 3, page 68) that is regularly found near the bottom of peat deposits may differ. The lithology of this layer can be described as peat with blackish-brown colouring, greasy consistency and very few recognisable plant remains. Depending on definitions used, it is plausible that in the DINOloket corings reference is made to a similar facies as the amorphous peat layer with the term gyttja. Proximity of some of the DINOloket corings with gyttja in the description to our sample corings with an amorphous peat layer suggests that both terms refer to the same layer. As this layer does not meet the requirements of true gyttja (an organic lacustrine deposit) according to Bos (2010) and Bos et al. (2012), we conclude that peat initiation in the study area was largely caused by paludification, i.e. waterlogging of previously unsaturated sediments. Note that we used an OM percentage of 40% to define peat based on Quik et al. (2022a), which is important to keep in mind for comparison of results with other studies.

Trophic status is used in many peatland classification systems. Reference can be made to the current trophic status (suffix 'trophic'), and to trophic status during peatland initiation (suffix 'genous') (Charman, 2002c). Geogenous or minerogenous conditions indicate that a peat-forming vegetation receives ground-water or surface runoff, i.e. water that has been in contact with mineral soil. Ombrogenous conditions are present when a peat-forming vegetation receives water solely from precipitation (Rydin and Jeglum, 2013c; Charman, 2002c; Joosten and Clarke, 2002; IPS, 2022). The water source during formation of the organic deposits at Fochteloërveen could be deduced from the botanical data (table 4.1,

see reference to the dataset on page 105). Of the peat initiation samples five can be classified as geogenous, nine as transitional from geogenous to ombrogenous, and only one as truly ombrogenous. This suggests that peat initiation was strongly influenced by groundwater and surface runoff, and only to a limited degree by perched groundwater tables that result from precipitation and poor drainage caused by impermeable (sub)surface layers (i.e., glacial till which can be found close to the surface, see section 4.2.1).

The weak linear relationship between elevation of the Pleistocene mineral surface and peat initiation age (p-value = 0.09, adjusted  $R^2 = 0.14$ ) shows that not all basal peat layers of equal elevation are of the same age. In contrast, the relationship between peat initiation age and groundwater-fed wetness that results from geomorphological position is highly significant (p-value < 1e-3, adjusted  $R^2 = 0.61$ ). This demonstrates that the influence of groundwater in initiating peat growth at Fochteloërveen cannot be explained solely by elevation, but that it is strongly related to groundwater-fed wetness resulting from position within the large-scale geomophology of high topographic plains and insized valleys. Sea level was rising during the entire period of peat development at Fochteloërveen (Meijles *et al.*, 2018). As a result, the isohypse pattern gradually rose through time.

Transects of basal peat dates are useful to distinguish development loci from lateral expansion areas (e.g. Mäkilä, 1997; Mäkilä and Moisanen, 2007; Chapman *et al.*, 2013). At Fochteloërveen, landscape scale peat initiation (fig. 4.1) occurred simultaneously at multiple sites (fig. 4.8). Some of the loci of peat initiation are located at positions that are lower compared to surrounding topography (areas of accumulated flow or sinks). Our predictions indicate that even cover sand ridges eventually became covered with peat (fig. 4.8), suggesting that lateral expansion was not slope-limited in this area or below its threshold. The pattern (fig. 4.8) and pace (fig. 4.10b and 4.10c) of lateral expansion show that after initial slow lateral growth of peat initiation loci, lateral growth accelerated. The strongest expansion occurred between 5,500 - 3,500 cal y BP.

Some of these findings are contrasting with previous palaeogeographic reconstructions by Fokkens (1998), Waterbolk (2007) and Vos *et al.* (2020). Fokkens (1998) placed peat initiation in the Fochteloërveen area between the beginning of the Early Iron Age and the end of the Roman period (800 BCE – 400 CE; 2,750 - 1,550 cal y BP). This is much later than our dating results (table 4.1) and predictions (fig. 4.8) demonstrate. Waterbolk (2007) on the other hand assumes that large areas were already covered by peat during the Early and Middle Neolithic (4,900 - 2,850 BCE; 6,850 - 4,800 cal y BP). This is roughly in agreement with our results; according to our predictions half of the Fochteloërveen area was covered with peat by ~4,000 cal y BP (fig. 4.10c). However, Waterbolk (2007) concludes that this situation remained stable for ~3,000 years followed by
rapid peatland expansion, which left the area largely abandoned by the Roman Period (19 BCE – 450 CE; 1,969 – 1,500 cal y BP). This is in strong contrast with our findings, which do not indicate a period of stability that precedes further lateral expansion. In the national-scale reconstructions by Vos *et al.* (2020), it was assumed that peatlands expanded gradually until they reached their former maximum extent, but the authors indicate that this is mainly due to a lack of data. Our results demonstrate that peat loci at Fochteloërveen probably expanded in a non-gradual fashion, with a phase of accelerated lateral expansion between 5,500 – 3,500 cal y BP (fig. 4.10b and 4.10c).

Ruppel *et al.* (2013) analysed an extensive dataset of basal radiocarbon dates reflecting both peat initiation and lateral expansion in Northern Europe and in North America. Their data on lateral growth demonstrate that the expansion of existing peatlands accelerated between approximately 5,000 – 3,000 ka, both in Northern Europe and in North America (Ruppel *et al.*, 2013). Similarly, Korhola *et al.* (2010) found that high-latitude peatlands in Europe expanded most drastically after 5 ka. Our data fit within this large-scale trend as they indicate a phase of accelerated lateral expansion at Fochteloërveen between 5,500 – 3,500 cal y BP (fig. 4.10b and 4.10c). Ruppel *et al.* (2013) suggest that this trend may be related to Neoglacial cooling (Wanner *et al.*, 2008).

In a core from Fochteloërveen studied by Klaver (1981); Van Geel *et al.* (1998), a radiocarbon date of a plant macrofossil sample collected near the visual mineralto-peat transition indicated an age of 2920 - 2736 cal y BP (95.4% confidence interval, uncalibrated age of  $2690 \pm 50$  BP; core location indicated in fig. 4.9a). Based on this and a comparison with several other regions, Van Geel *et al.* (1998) infer an influence of the 2.8 ka event where a change in climate results in peat initiation on previously unsaturated soils. However, our predictions of peat initiation age (fig. 4.8) indicate that the site studied by Van Geel *et al.* (1998) probably became covered with peat through lateral expansion, and does not reflect landscape scale peat initiation (fig. 4.1). An effect of the 2.8 ka event does not become clear from our results (fig. 4.10).

#### 4.5.2 Evaluation of approach

Palaeogeographical studies of former extensive peat landscapes are challenging as vast areas have lost their former peat cover, affecting the natural archive formed by the peat and thus limiting the options for collecting field data (fig. 4.1b and 4.1c). Consequently, alternative approaches are needed to reconstruct peatland development on the Northwest European mainland and other areas where peat is poorly preserved. Our field strategy consisted of spatially distributed transects that were placed perpendicular to elevation gradients in the mineral subsurface

underlying the organic deposits. This resulted in an extensive set of basal radiocarbon dates that stretches both the lateral and vertical dimensions of the Fochteloërveen peat remnant, and as such formed the basis for subsequent modelling steps. This approach was found useful when options for transects covering a whole peatland (as in fig. 4.1b) are hampered by limited a-priori knowledge on the position of peat remnants within the former peat landscape (fig. 4.1c). The availability of coring data within national databases (see section 4.3.2) was highly useful for constructing a palaeoDEM of the Pleistocene mineral surface underneath the organic deposits. For areas where comparable data are not available, the sampling scheme may need to be expanded as it must also provide sufficient data for interpolation to a palaeoDEM (for examples where basin morphometry is studied, see e.g. Anderson *et al.*, 2003; Bauer *et al.*, 2003; Chapman *et al.*, 2013).

To reconstruct peat initiation age for non-sampled sites within the peat remnant, we applied a digital soil mapping approach. For data-intensive approaches such as geostatistics (Oliver and Webster, 2014) or random forests (Breiman, 2001) the amount of data (specifically radiocarbon dates) is generally too low. Therefore, a digital soil mapping technique was needed that is less data-intensive. In our study, we found linear regression to be the best option as it involves only few assumptions, which were valid for our dataset.

Advantages of this approach are that it is spatially explicit (i.e., results in a map of predicted peat initiation ages), and it offers a quantitative alternative compared to manual deduction of isochrones from transects of basal dates (e.g. as in Bauer *et al.*, 2003; Foster *et al.*, 1988; Korhola, 1994, 1996; Mäkilä, 1997; Mäkilä and Moisanen, 2007). In addition, our approach allows for a quantitative evaluation of the prediction using the standard deviation and comparison of predicted ages with validation points.

To the best of our knowledge, digital soil mapping approaches have so far hardly been explored for reconstructing peat initiation and lateral expansion. An important exception is the work of Chapman *et al.* (2013), who made use of second-order polynomial regression to reconstruct peat initiation ages based on empirical relationships between basal peat age and DEM-derivatives for two remnants of a floodplain raised mire (Hatfield and Thorne Moors, UK). They tested relationships between peat initiation age and (1) elevation, (2) proximity to river courses, and (3) flow accumulation. Similar to our results, they found a weak relationship between peat initiation age and elevation (R<sup>2</sup> values of 0.15 and 0.20 for linear and second-order polynomial regression respectively (p-value not reported), compared to an adjusted R<sup>2</sup> of 0.14 and p-value of 0.09 for the linear regression in our study). Proximity to rivers also yielded a weak relationship, whereas flow accumulation produced an R<sup>2</sup> of 0.39 for linear regression and of 0.91 for second-order polynomial regression. Hence they apply the second-order polynomial relationship with flow accumulation as covariate to reconstruct peat initiation ages for their study area. Unfortunately, they do not provide an indication of certainty of their prediction, but do compare the prediction result with five validation points. Based on the relationship with flow accumulation, they conclude that peat growth initiated in the area at locations where surface run-off accumulates and results in a terrestrialisation process.

In our study, we reasoned with two points in mind, and used a combination of process-informed choices within a statistical approach. Firstly, we hypothesized that peat initiation at Fochteloërveen would be subjected to a two-fold control consisting of elevation and position within the large-scale geomorphology of high topographic plains and insized valleys. Secondly, we valued a covariate that is independent of peat presence or thickness, as this may have the potential to estimate peat initiation age for areas that are no longer covered by peat. Based on the assumption that the current isohypse pattern within the Fochteloërveen peat remnant reflects the isohypse pattern of the past (but in the past positioned at lower elevation due to lower sea level), we constructed the variable  $z_PH$  based on elevation of the mineral surface and present-day hydraulic head. This covariate then allows to explain peat initiation with groundwater-fed wetness that results from geomorphological position.

As stated above, in our model based on  $z_P H$ , we assume that the current isohypse pattern reflects the natural situation as it was once present in the area (but with the pattern as a whole currently positioned at higher elevation). Within the boundaries of the Fochteloërveen peat remnant this assumption is largely true, as nature conservation measures require high groundwater tables. In addition, hydraulic conductivity within the area is not subjected to change (i.e. for the deposits underneath the peat). However, as groundwater levels in the surrounding area are much lower due to artificial drainage, an edge effect is probably present near the border of the peat remnant. This is reflected in the prediction map of peat initiation ages based on  $z_P H$  in fig. 4.8b, which shows much younger ages near the edges (especially in the north) compared to the prediction map based on O in fig. 4.8a. This is also suggested by the modelling results if we would not hold on to our temporal scope with 900 cal y BP as limit. In that case, the  $z_PH$  model predicts that peat near the edges of Fochteloërveen is of very recent age and locally even later than 0 cal y BP (i.e., these are otherwise set to 'No Data'). However, an historical map of 1664 CE indicates that peat was probably present at these locations (Pynacker, 1664). We encountered a comparable problem during an exploratory attempt to reconstruct peat initiation ages outside the peat remnant, which results in peat initiation ages that are likely too young. This shows that to obtain a reliable prediction, the method requires the isohypse pattern to reflect the natural pattern as closely as possible, and cannot predict peat initiation ages

for areas that are subject to strong artificial drainage.

Our model based on total thickness of organic deposits [O, is based on the concept that the thicker the organic deposit, the longer ago peat was initiated at that location. This is of course dependent on peat growth rates, decay rates, and degree of compaction. However, despite these complicating factors, our regression model of peat initiation age versus total thickness of organic deposits is highly significant (p-value < 1e-3) and has a decent fit (adjusted R<sup>2</sup> = 0.57). We made use of newly collected and existing coring data to obtain thickness values, which reflect the thickness of the peat layer that is still present. If the natural peat thickness (from before the onset of reclamations and peat-cutting) would be known, perhaps the fit would increase further.

Where remnant peat survived, covariate O is useful and may provide an estimation of the peat initiation age based on the thickness of the remaining peat layer. Covariate  $z_P H$  is potentially useful for reconstructing peat initiation age beyond peat remnants, given that the data on which this covariate is constructed reflect the natural situation. A combination of both may offer new options to reconstruct peatland development within and beyond peat remnants. For the  $z_P H$ model, this requires data on the natural topography of the mineral surface that used to be covered with peat (where peat is lost, the mineral surface may have been subject to levelling activities) and on the natural isohypse pattern. If this pattern could be derived for the region surrounding a peat remnant using hydrological modelling, the  $z_P H$  model could provide insights in peatland initiation and lateral development for areas where this information cannot be collected from the field. It is important to keep in mind that complicating factors may have played a role during peatland development, for instance where peat growth in valleys changes regional base level or where the formation of peat domes affects drainage divides. As peat may be largely absent outside (protected) peat remnants, options to obtain radiocarbon dates to verify model results are probably very limited. Another option would be to use TAQ and TPQ dates obtained from archaeological data. In the example of Fochteloërveen, the surrounding area contains a fairly large number of archaeological finds which could offer a regional-scale validation dataset.

# 4.6 Conclusion

Reconstructions of peat initiation and lateral expansion in areas where the former peat cover is largely lost, such as the Northwest European mainland, are severely hampered by the limited options for collecting field data. In this study we aim (1) to find explanatory variables within a digital soil mapping approach that

allow us to reconstruct the pattern of peat initiation and lateral expansion within (and potentially beyond) peat remnants, and (2) to reconstruct peat initiation ages and lateral expansion for one of the largest bog remnants of the Northwest European mainland, the Fochteloërveen in the Northern Netherlands. Basal radiocarbon dates that were obtained from the peat remnant formed the basis for subsequent analyses. Significant relationships were found between peat initiation age and total thickness of organic deposits, and between peat initiation age and a constructed covariate on groundwater-fed wetness based on the present-day hydraulic head relative to the mineral palaeosurface underneath the peat cover. In contrast, a weak relationship was found between peat initiation age and elevation of the mineral palaeosurface. These findings indicate a strong influence of position within large scale geomorphology (high plains and insized valleys) on peat initiation at Fochteloërveen. The digital soil mapping approach based on thickness of organic deposits is only useful where remnant peat survived. whereas the constructed covariate on groundwater-fed wetness may ultimately be applied beyond the limits of peat remnants. Thereby this novel approach has the potential to shed light on the pattern, timing and pace of peatland initiation and lateral expansion in areas where this information can no longer be obtained from the field. For the Fochteloërveen our results indicate simultaneous peat initiation at several loci during the Early Holocene, and continuous lateral expansion until 900 cal y BP. Lateral expansion accelerated between 5,500 - 3,500 cal y BP.

#### Author contributions

RvB secured funding for this research. CQ proposed the research outline and adjusted it based on discussions with JC, YvdV, JW and RvB. JC and CQ explored the study area, performed test corings and decided upon a coring strategy. CQ organised the fieldwork and completed the collection of data and cores in the field with help of several students (see Acknowledgements). CQ subsampled the cores and coordinated all subsequent analyses on the subsamples. CQ analysed the data with input from YvdV, JC, JW and RvB; LS advised and assisted with the statistical analysis. CQ wrote the main body of text. The manuscript was finalized based on feedback from all authors.

#### Acknowledgements

We are thankful for the collaboration with Natuurmonumenten Noordenveld and Staatsbosbeheer Kop van Drenthe for access to and inside knowledge of the field sites; Teun Fiers, Klais Blaauw, Tom Harkema, Marte Stoorvogel, and Gibran Leeflang for field assistance; Mieke Hannink for administrative assistance with organising the fieldwork; Marcello Novani (Laboratory for Geo-information Science, WUR) for providing GNSS equipment; Jim Quik for construction work for core transport; and Piet Peters for providing storage space in the cool room (Soil Hydro Physics Lab, WUR). The contributions of Anne Roepert and colleagues (CBLB, Wageningen) for LOI analyses; Marjolein van der Linden and Lucy Kubiak-Martens (BIAX Consult, Zaandam) for selecting plant macrofossils for radiocarbon dating; and Sanne W.L. Palstra for the radiocarbon measurements at the Centre for Isotope Research (University of Groningen) are gratefully acknowledged. We appreciate the constructive comments of two anonymous reviewers, which helped to improve an earlier version of this paper.

# Chapter 5

General synthesis

# 5.1 Introduction

In the first chapter of this PhD thesis I identified two key research deficits, regarding (I) the need for methodological developments to constrain the spatio-temporal development of peatlands more accurately, and (II) the limited understanding of spatio-temporal trends in and steering factors for peatland initiation and lateral development in the coversand landscape of the Northwest European mainland. In this chapter, I synthesize and reflect on the methodological advances that were developed in this thesis (section 5.2), and on the knowledge that was gained regarding peatland development (section 5.3). Subsequently, I summarise the main conclusions (section 5.4), discuss recommendations for future research (section 5.5), and end with implications of the thesis results (section 5.6).

# 5.2 Methodological advances

# 5.2.1 Reuse of existing radiocarbon dates in peatland geochronology

In the geosciences, data rescue and reuse are of increasing interest (Wyborn *et al.*, 2015). However, for geochronological peat research no overviews exist of factors that need to be taken into consideration for reuse of radiocarbon dates, and standardized workflows or designs for quality assessments of peat dates are lacking (section 1.2 and 1.3). In Chapter 2, I developed a workflow for data rescue and reuse of legacy radiocarbon dates in peatland studies, including a rigorous quality assessment. The latter is intended to provide insight in potential sources of error and enables working with confidence levels based on subsets of data with increasing uncertainty. In this way, accuracy and robustness of conclusions can be verified. The developed workflow consists of:

- i. A database set-up that can be tailored as required by the study scope;
- ii. A complementary set of quality criteria with flexible weights to suit specific research questions;
- iii. A script for automated quality assessment of the recorded legacy data using the weights defined in point (ii), to make the approach suitable for evaluating large legacy datasets.

In the quality assessment of point (ii), a penalty is assigned to each date based on criteria that consider taphonomic quality (i.e., sample provenance) and dating quality (i.e., sample material and method used). The degree to which a

radiocarbon date represents the event of interest is determined by its dating and taphonomic quality. The quality criteria cover two types of information, i.e. aspects that are considered negative, and the availability of information about a certain property. The first type is based on 'best practices' in scientific literature regarding methodological aspects of dating (for comparable approaches, see e.g. Small *et al.*, 2017). For example, dates based on bulk samples are generally considered to be less accurate than those based on plant macrofossil samples (Törnqvist *et al.*, 1992, 1998; Piotrowska *et al.*, 2011). The second type considers whether sufficient information is available to make informed choices with regard to data analysis. Here the focus is not on the property itself (for instance, the location itself is not judged), but on knowledge about the property (do we know the location well or not).

In Chapter 2, the value of the workflow was demonstrated through application to a case study area, for which the northern Dutch coversand landscape was selected. Data rescue for the case study proved to be cost-efficient and provided access to information of which part is no longer available in the field. The comprehensive dataset for the case study area and overarching trends in peat growth that emerged from the meta-analyses highlight the potential of data reuse for palaeogeographic reconstructions.

## 5.2.2 Defining and dating basal peat

The lack of a universally applicable and quantitative definition for basal peat, combined with multiple concerns that have been raised previously regarding the radiocarbon dating of peat, may result in apparent ages that are either too old or too young for the timing of peat initiation (section 1.2 and 1.3). To contribute to resolving these challenges, I analysed a detailed series of radiocarbon dates in Chapter 3, comparing ages of multiple organic fractions from the mineral-to-peat transition in three cores obtained at Fochteloërveen. The approach was based on a conceptual framework (fig. 3.1) explaining the difference between local and landscape scale peat initiation, the issues with defining the mineral-to-peat-transition (which reflects the timespan of peat initiation) and basal peat layer, the challenges in sample selection, and the datable fractions and potential contaminants.

Chapter 3 illustrated the value of studying the mineral-to-peat transition using the organic matter gradient, and to base the definition of *basal peat* on this gradient. The timespan of the peat initiation process that is reflected in the mineral-to-peat transition (a stratigraphical distance of 6, 7 and 8 cm in the studied cores), lasted for 1343, 1073 and 1510 years (medians) respectively. It is important to take this significant timespan into account when deciding upon a

sample size and sampling resolution. The *basal peat* layer can be defined using the organic matter gradient to provide a quantitative and reproducible definition that can be obtained through a simple protocol and at low-cost.

The dating results presented in Chapter 3 demonstrate that the ages obtained from plant macrofossil samples are most accurate in the mineral-to-peat transition and are therefore recommendable to use. However, the mineral-to-peat transition may be low in plant macrofossil content. If plant macrofossil samples cannot be obtained, dates of the humic fraction provide the best alternative regarding chronological order, but may deviate significantly from the 'true age' and are most safely interpreted as terminus-ante-quem dates. As the options for collecting dating samples might be limited and resulting sample volumes might be small, it is important to assign material to analyses based on a comprehensive (sub)sampling scheme. The findings of Chapter 3 were summarised as a set of recommendations for defining and dating peat initiation that can be applied in various peatland settings (see Textbox 3.1 in Chapter 3 on page 57).

# 5.2.3 Reconstructions based on peat remnants

Adapted strategies are needed to obtain field data and to model peat initiation and lateral expansion based on peat remnants (section 1.2 and 1.3). To contribute to addressing this need, I searched for explanatory variables within a digital soil mapping approach in Chapter 4. I found significant linear relationships between median peat initiation age and the total thickness of organic deposits, and between median peat initiation age and a constructed covariate on groundwater-fed wetness. The latter was based on the present-day hydraulic head relative to the mineral palaeosurface underneath the peat cover.

To reconstruct peat initiation age for non-sampled sites within the Fochteloërveen peat remnant, a digital soil mapping approach was applied where either a map of total thickness of organic deposits or a map of groundwater-fed wetness was converted into a map of peat initiation ages using the abovementioned significant relationships. Key benefits of this method are that it directly results in a map of predicted peat initiation ages, and a quantitative evaluation of the prediction can be obtained using the standard deviation of the prediction and comparison of predicted ages with validation points. As such, it offers an alternative where uncertainty is quantified compared to manual deduction of isochrones from transects of basal dates (e.g. as in Foster *et al.*, 1988; Korhola, 1994, 1996; Mäkilä, 1997; Bauer *et al.*, 2003; Mäkilä and Moisanen, 2007).

The use of total thickness of organic deposits as covariate is only useful where remnant peat survived and a thickness of this layer can be derived. In contrast, use of the constructed covariate on groundwater-fed wetness may ultimately be applied beyond the limits of peat remnants, as it does not require information from peat deposits. As such, this novel approach enables reconstructions of the pattern, timing and rate of peatland initiation and lateral expansion in areas where the peat cover is (largely) lost.

For the reconstruction based on groundwater-fed wetness, we made the assumption that the current isohypse pattern reflects the natural situation as it was once present in the area, but with the pattern as a whole currently positioned at higher elevation. This assumption is reasonable within the boundaries of the Fochteloërveen nature reserve where groundwater levels are kept high (Provincie Drenthe, 2016; Altenburg *et al.*, 2017). However, in the surrounding area artificial drainage causes much lower groundwater levels. The results presented in Chapter 4 illustrate that this probably causes the prediction of younger ages near the boundary of Fochteloërveen. During a trial to reconstruct peat initiation ages outside the peat remnant, which by default has to be modelled based on groundwater-fed wetness as organic deposits are lacking, we came across a similar issue with modelling results indicating peat initiation ages that are likely too young. This underlines the importance that the isohypse pattern must reflect the natural pattern as closely as possible to obtain a reliable prediction, and that the method is not applicable to areas where groundwater levels are artificially lowered.

# 5.2.4 Relevance and integration of methodological tools

The methodological tools presented in this thesis aim to advance science on peatland reconstructions. Depending on research objectives, they can be applied separately, or concurrently in a complementary manner.

# Relevance

The workflow for data reuse and quality assessment (Chapter 2) were designed with adaptability in mind, making them suitable for application in studies with divergent research questions and study regions. To obtain this flexibility, weights were applied to the quality criteria that may be adjusted based on research focus. Depending on which information can be retrieved (and the resulting penalty score and confidence level), data may be filtered prior to data analysis (for example, first including only sites with well-known location, then analysing sites with uncertain location as well). This allows a purposeful assignment of dates to various analyses and helps to ensure veracity of conclusions. To make the workflow and quality assessment suitable for analysis of large datasets, the quality assessment was scripted in Python to automatically assign a penalty score and confidence level to each date.

The recommendations for defining and dating basal peat (Chapter 3) are of

relevance for a wide range of peatland studies. To move towards a quantitative definition of *basal peat*, organic matter content is a simple parameter that is easy to measure at low cost, which enables widespread use. Based on the organic matter gradient, a value can be chosen for the organic matter percentage above which the material is called peat. This value may differ between studies, but the described approach ensures that it is reproducible and eases comparison with other studies. The proposed steps for dating basal peat consider the timespan of peat initiation and consequences for sampling size and resolution. Application of the recommendations ensures that dating efforts are embedded in a conceptual framework and helps to obtain accurate dating evidence for peat initiation.

The application of digital soil mapping approaches in reconstructing peat initiation and lateral expansion (Chapter 4) is a field that has so far hardly been explored. A notable exception is the work of Chapman *et al.* (2013), who used second-order polynomial regression to reconstruct peat initiation ages based on empirical relationships with DEM derivatives. However, in their analysis the uncertainty of the prediction is not quantified, which is one of the key advantages of the approach in Chapter 4. The two covariates offer options for reconstructing peat initiation and lateral expansion within and potentially beyond the boundaries of peat remnants. Similar relationships between median peat initiation age and these covariates may be obtained for peat remnants elsewhere, enabling new options for peatland reconstructions in areas where large parts of the peat cover are lost.

## Integration

The case study presented in Chapter 2 illustrated that the benefit of such metaanalyses is first and foremost dependent on the success of data retrieval. Retrieval of legacy data may require a high level of thoroughness, and registration of retrieved information may be a meticulous task. Based on the experiences with the case study in Chapter 2, I therefore emphasise the great importance of FAIR sharing (Wilkinson *et al.*, 2016) information on basic properties of dating samples to prevent further data loss from natural archives at risk of decline. Key properties to share information about include location, elevation and stratigraphy, and details about radiocarbon dating (following Millard, 2014) including the laboratory measurement as a conventional radiocarbon age in <sup>14</sup>C yr BP, the laboratory code, the sample material, and pre-treatments. The conceptual framework and recommendations presented in Chapter 3 may further help to clearly report information on dating samples.

The large age difference between dates of plant macrofossils and humic or humin dates that was found in Chapter 3 (up to  $\sim$ 1700 years between macrofossil and humic ages, and with even larger differences for humins) indicates that

studies reusing legacy bulk dates of basal peat should take great care in data interpretation. Depending on sample details, dates are potentially interpreted more safely as terminus-ante-quem dates for peat initiation. Alternatively, they should be subjected to rigorous quality assessment prior to data analysis as proposed in Chapter 2.

The digital soil mapping approach presented in Chapter 4 requires a radiocarbon dataset for studies on regions with different properties than the Fochteloërveen peatland to obtain the required linear relationships. The effort and resources needed to obtain such a dataset may form an impediment to use the presented approach. However, current methodologies based on basal transects and isochrones require a similar dataset. Potentially the need for a radiocarbon dataset might be covered through reuse of legacy data (Chapter 2). If new data are obtained, a multi-stage dating approach is recommendable (also see staged approaches for dating as suggested by Bayliss, 2009; Piotrowska et al., 2011). The first stage may consist either of an elaborate dating inventory comparable to the approach in Chapter 3 or of a more restricted preceding test (e.g., dating plant macrofossils and humics of three levels dispersed over the mineral-to-peat transition). The former provides detailed information on the peat initiation process and on the accuracy of different carbon fractions for dating, whereas the latter only provides the necessary insights on the duration of peat initiation and may consequently inform choices for sampling size and resolution before performing extensive spatial dating schemes as in Chapter 4.

# 5.3 Advances in understanding initiation and lateral development of peatlands in the Northwest European mainland

In this section, I will zoom out from the microtope (Chapter 3; section 5.3.1) to meso- and macrotope (Chapter 4; section 5.3.2) and beyond the macrotope level (Chapter 1; section 5.3.3). I end with a hypothesis regarding the steering factors for peat formation in coversand landscape of the Northwest European mainland (section 5.3.3).

# 5.3.1 Palaeoenvironment in the Fochteloërveen area during peat initiation

The biostratigraphical analyses presented in Chapter 3 and 4 demonstrate that the initial peat-forming vegetation at Fochteloërveen was mesotrophic and dominated by sedges (*Carex* spp.), with some presence of *Juncus*. For each location where a

peat initiation date was obtained, the source of water during the formation of the first organic deposits could be deduced from the botanical data (Chapter 3 and 4). Of the fifteen peat initiation samples, five can be classified as geogenous, nine as transitional from geogenous to ombrogenous, and only one as truly ombrogenous (for definitions, see Textbox 1.1 in Chapter 1). These findings indicate that the initial formation of peat took place primarily under the influence of groundwater and surface runoff (Charman, 2002c).

As was discussed in Chapter 3, several levels of the three studied cores contained sclerotia of *Cenococcum geophilum*. This mycorrhizal fungus that usually lives in the sandy subsoil, may indicate rather dry conditions when encountered in peat (Van Geel, 1978). The investigated levels contained very little amoebae, but remains of *Difflugia* species were found in a few samples which points to very wet conditions. Interestingly, C. geophilum and Difflugia did not occur simultaneously in any of the investigated levels (see table 3.6 in Chapter 3). This suggests that water tables may have fluctuated and that drier (C. geophilum) and wetter (Difflugia) conditions alternated during the timespan of the peat initiation process. Periods with drier conditions may have caused the limited preservation of uncharred plant macrofossils, as these circumstances probably allowed increased decomposition of organic material. Charred plant macrofossils were encountered in the majority of the investigated levels of the three vertical sequences under study in Chapter 3, and in all but one of the dated levels in Chapter 4. This almost ubiquitous presence of charred organic material demonstrates that (local) fires must have occurred on a regular basis.

The mineral-to-peat transition, which resulted in the formation of mesotrophic fen peat, was followed by a fen-bog transition due to ombrotrophication (for general information on fen-bog transitions see Hughes, 2000; Charman, 2002a; Rydin and Jeglum, 2013d; for examples see e.g. Almquist-Jacobson and Foster, 1995; Hughes and Barber, 2004; Loisel and Bunsen, 2020. As the cores studied in Chapter 3 were directed at obtaining the mineral-to-peat transition, they did not fully contain the higher-positioned fen-bog transition. However, the botanical analyses of the upper layers of the cores indicate that at sites S17 and S18 the peatforming vegetation developed to an oligotrophic bog with Calluna vulgaris, Erica tetralix and Sphagnum. For site S20 the record contained in the core does not clearly contain the fen-bog transition, but it seems that the vegetation became dominated by heathers and mosses (Bryales). At different locations within a peatland, fen-bog transitions may take place at different timings (Väliranta *et al.*, 2017), which is also the case at the three Fochteloërveen sites. At site S17 the transition took place probably at or after ~5000 cal y BP, at site S18 at or after  $\sim$ 1800 cal y BP, and at site S20 after  $\sim$ 2000 cal y BP (compare table 3.6 and fig. 3.7 in Chapter 3). A full description of the palaeoecological development at

Fochteloërveen will be presented in Van Beek et al. [in prep].

# 5.3.2 Peat initiation and lateral expansion at Fochteloërveen and the northern Dutch coversand landscape

An overview of the peat initiation data that were obtained in this thesis is presented in fig. 5.1. As demonstrated in Chapter 4, landscape scale peat initiation (following the conceptual framework in fig. 3.1 in Chapter 3) in the Fochteloërveen area took place at several loci during the Early Holocene (>7000 cal y BP). The reconstructions in fig. 4.8 indicate that these simultaneous peat initiation sites were located in the lower central, west and northwest parts of the area. The rate of lateral expansion can be derived from the distance between isochrones: where the distance between isochrones is small, the rate of lateral growth was low, and vice versa. In the reconstructions of Fochteloërveen, isochrones are mostly lying close to one another in the oldest peat-covered areas, indicating that peat initiation loci initially expanded slowly (see fig. 4.8/4.10 in Chapter 4, and fig. 5.1 below). As time progressed, lateral expansion accelerated, with the most pronounced period of expansion between 5,500 - 3,500 cal y BP. Sustained lateral expansion resulted in a peat cover for half of the Fochteloërveen area by 4,000 -3.500 cal y BP, and for nearly the entire area between 2,500 - 900 cal y BP. The findings from Chapter 4 indicate that Fochteloërveen is a composite peatland that probably formed through coalescence of multiple smaller mires that formed on a topographic plain.

The meta-analysis of the legacy dataset in Chapter 2 shows a bimodal distribution of basal peat dates (fig. 2.8 and fig. 5.1). The legacy basal dates reflect peat initiation at the local scale (following the conceptual framework in fig. 3.1 in Chapter 3). Unfortunately, initiation loci or sites of lateral expansion could not be distinguished from one another, as it often remained unclear whether legacy basal dates originated from the same peatland (this would require a-priori a clear view of their palaeogeography). As a result, the analysis provides information about peat initiation and lateral expansion combined. The first phase of peat initiation (and lateral expansion) started during the Late Pleniglacial from  $\sim 16,000$  cal y BP and lasted until  $\sim$ 12,000 cal y BP (depending on which certainty level is followed in fig. 5.1b). The second phase started  $\sim$ 9,000 cal y BP and continued until  $\sim$ 1,000 cal y BP, with a peak at approximately  $\sim$ 4,500 cal y BP. Peat initiation (and lateral expansion) data were separately analysed for different landform groups, which revealed several trends (fig. 2.9). River valleys were subject to Late Glacial peat growth, which continued during the Holocene. Peat growth on plains and ridges on the other hand started much later at  $\sim$ 6,000 cal y BP and lasted to  $\sim$ 2,000 cal y BP. Topographic depressions such as pingo remnants and deflations in coversand

are characterized by an erratic pattern of peat growth through time. For sites with unclear palaeo-landform underlying the organic deposits, the data fit within the bimodal distribution that was found for the complete dataset. The dating evidence and modelling results for Fochteloërveen fit within the second phase of peat growth (fig. 5.1b/c).

# 5.3.3 Process of peat initiation and steering factors for peat growth

Peat initiation results from terrestrialisation, paludification and/or primary mire formation (see Textbox 1.2 in Chapter 1; Charman, 2002a; Rydin and Jeglum, 2013d). Peat formation in the northern Dutch coversand landscape is not connected with primary mire formation, as this region has been deglaciated and exposed at the surface since the penultimate Glacial (Ter Wee, 1962). Terrestrialisation typically results in gyttja (an organic lacustrine deposit) at the base of the organic deposits. To assess the prevalence of terrestrialisation versus paludification in the northern Dutch coversand landscape, the registered sample material of legacy basal dates (n = 74) was evaluated for presence of gyttja (see section 2.3.2.2) and table 2.5). Gyttja was only present in six samples. For Fochteloërveen, an amorphic organic layer was encountered at the mineral-to-peat transition, but this layer does not meet the requirements of true gyttja (following Bos et al., 2012). Based on these findings, I conclude that paludification was the most prominent process of peat formation in the northern Dutch coversand landscape (and Fochteloërveen), with local occurrence of terrestrialisation sites. In the subsequent discussion of steering factors for peat growth, I will therefore focus on potential drivers of the paludification process.

Peat formation in Northwest Europe is listed as a process with "cause under debate" in the summary chart of environmental and cultural changes during the Holocene by Roberts (2014). The work presented in this thesis points to several environmental factors that may have led to paludification, consequent peat initiation and subsequent lateral expansion. In the subsequent paragraphs I will discuss the potential effects of climate changes during the Last Termination and Holocene, sea level rise during the Holocene, impermeable (sub)surface layers, and human influence. I end with a hypothesis on the steering factors for paludification and peat growth in the Northwest European mainland.

# Climate change

The bimodal distribution in fig. 5.1b shows that the first phase of peat growth started around ~16,000 cal y BP, peaked at ~14,000 cal y BP and lasted until ~12,000 cal y BP, followed by a distinct low at approximately ~9,500 cal y BP. This first phase of peat growth started after the Last Glacial Maximum during the

Late Pleniglacial (fig. 5.1a/b). Its peak coincides with the onset of the Bølling-Allerød interstadial (14.7 – 12.9 cal ky BP; Rasmussen et al., 2006; cal ky BP  $\approx$  ka b2k). The analysis of datasets with different confidence levels indicates unclarity about the duration of this phase. When considering only data with a green confidence level (i.e., most reliable), the phase ends around  $\sim 12,000$  cal y BP, but the datasets that include data with lower confidence levels point towards ending of this phase around  $\sim$ 10,000 cal y BP (fig. 5.1b). It remains uncertain whether growth continued during the Younger Dryas (12.9 - 11.7 cal ky BP)Lowe and Walker, 2015) and Greenlandian Stage of the Holocene (11.7 - 8.2 cal)ky BP; Walker *et al.*, 2019). However, the data subset with green confidence level suggests that peat growth was halted during this time. Peat growth continued during a second phase starting at  $\sim$ 9,000 cal y BP, which peaked at  $\sim$ 4,500 cal y BP and lasted until ~1,000 cal y BP. The rise of this second phase fully overlaps with the Northgrippian Stage (8.2 - 4.2 cal ky BP; Walker et al., 2019). It coincides partly with the hypsithermal or Holocene Thermal Maximum (9,000 - 6,000/5,000 cal y BP; Wanner et al., 2008; Renssen et al., 2009) and with strong Holocene sea level rise (fig. 5.1a/b). The peak of the second phase may have been coeval with the 4.2 cal ky BP event and overlaps with the period of neoglacial cooling (6,000/5,000 cal y BP - pre-industrial time; Wanner et al.,2008). The recession of peat growth falls within the Meghalayan Stage (4.2 cal ky BP – present; Walker et al., 2019) and neoglacial cooling. These findings led to the tentative conclusion in Chapter 2 that the first phase of peat growth was mostly driven by climate, whereas the second was probably the result of climatic conditions favourable to peat growth in combination with Holocene sea level rise and related rise in groundwater levels.

**Figure 5.1:** (Figure on next page). Overview of peat initiation data obtained for the northern Dutch coversand landscape and Fochteloërveen. (a)  $\delta^{18}O$  curve (GICC05 NGRIP  $\delta^{18}O$  data accessed through OxCal), climatic division and events from Walker et al. (2009, 2019). Relative Sea Level (RSL) curve for the Northern Netherlands from Meijles et al. (2018). (b) Results from the kernel density estimation models presented in Chapter 2, showing the bimodal distribution of peat initiation dates (basal peat dates) in the northern coversand landscape of the Netherlands. The dark grey area indicates the sampled KDE estimated distribution. The blue line shows the mean of the KDE distribution, the lighter blue band shows the  $\pm 1\sigma$  range. The calibration curve is indicated for reference (Reimer et al., 2020). (c) Histogram and density plot of predicted peat initiation ages for Fochteloërveen presented in Chapter 4, showing results for the prediction based on the current total thickness of organic deposits [O], and for the prediction based on groundwater-fed wetness [ $z_PH$ ]. Predictions were limited at 900 cal y BP due to onset of peatland reclamations (see Chapter 4). (d) Plot presented in Chapter 4, showing cumulative percentage of peat-covered area within the Fochteloërveen area.



The results of Chapter 4 indicated that lateral expansion at Fochteloërveen accelerated most rapidly between 5,500 - 3,500 cal y BP (fig. 5.1c/d). This overlaps with the peak of the second peat growth phase in fig. 5.1b. Similarly, Korhola *et al.* (2010) found that high-latitude peatlands in Europe expanded most drastically in the period between 5,000 - 3,000 cal y BP. In addition, the meta-analysis of Ruppel *et al.* (2013) demonstrated that lateral expansion accelerated during this time, both for peatlands in Northern Europe and in North America. Ruppel *et al.* (2013) propose that this large-scale trend of accelerated lateral expansion could be related to neoglacial cooling.

Previously, two phases of climate-induced peat initiation for the Netherlands were described by Casparie and Streefkerk (1992). However the evidence for these phases remains implicit and the assumed timing of the phases differs from the results obtained in this thesis. Casparie and Streefkerk (1992) state that the first phase lasted from 7,000 – 6,500 BCE (~9,000 – 8,500 cal y BP) and that the second took place around 5,000 BCE (~7,000 cal y BP). Both of these periods overlap with the rise and peak of the second phase in fig. 5.1b, suggesting that they are only parts of what is in fact one larger phase of peat growth.

Based on a basal peat date of 2920 - 2736 cal y BP (95.4% confidence interval) obtained from a core from Fochteloërveen, and a comparison with data obtained for other regions, Van Geel *et al.* (1998) propose a major effect of the 2.8 ka cal ky BP event leading to peat initiation on previously unsaturated soils. A significant effect of the 2.8 cal ky BP event does not become clear from the bimodal distribution of the legacy data (fig. 5.1b). In addition, the site studied by Van Geel *et al.* (1998) became covered with peat through lateral expansion as indicated by the models of peat initiation age presented in Chapter 4 (fig. 4.6), i.e., the site reflects only local and not landscape scale peat initiation (following the conceptual framework in fig. 3.1 in Chapter 3).

#### Sea level rise

The results from Chapter 2 and 4 suggest that, in addition to a potential climate effect, Holocene sea level rise may be (partly) responsible for the development of non-coastal and non-alluvial peatlands in the coversand landscape of the Northwest European mainland. Peat growth in coastal plains and in tidally influenced floodplains is directly controlled by sea level (e.g. Törnqvist and Hijma, 2012). Peatlands in the Dutch coversand area are considered to be fundamentally different (e.g. Pons, 1992: 7), suggesting that for these peatlands sea level is of lower relevance. In Chapters 3 and 4 however, the source of water during the formation of initial organic deposits could be deduced from the botanical data. This indicated that peat formed primarily under geogenous conditions (also see section 5.3.1, for definitions see Textbox 1.1 in Chapter 1), i.e., under influence

of groundwater and surface runoff (Charman, 2002c). This points towards a strong influence of position within large scale geomorphology (high plains and incised valleys) on peat initiation, related groundwater-fed wetness (Chapter 4) and a potential indirect effect of Holocene sea-level rise (fig. 5.1a).

## Impermeable (sub)surface layers

Some studies emphasise the relevance of impermeable (sub)surface layers or soil horizons in peat formation, which may cause accumulation of rainwater and perched groundwater tables (e.g. Everts *et al.*, 2002; Van der Meij *et al.*, 2018; Jansen *et al.*, 2019; Sevink, 2019). Geology and soils may influence the hydro-logical status during peat initiation (Charman, 2002b). Impermeable geological layers, such as boulder clay (which is present as a discontinuous layer underneath the northern Dutch coversand landscape and Fochteloërveen; Provincie Drenthe, 2022), form a boundary condition for drainage. In contrast, soil layers that are characterized by low permeability may form over the course of time, leading to an alteration in drainage conditions (Charman, 2002b). For instance, Sevink (2019) discusses multiple types of soil forming processes that may result in impermeable horizons, such as accumulation of humus and/or oxides during podzolization which results in a spodic horizon (IUSS Working Group WRB, 2015), that may develop to a degree where the horizon becomes impermeable to water.

Podzols are one of the most frequently occurring soils in the Dutch coversand landscape (Alterra, 2014). However, the results from Chapter 3 and 4 indicated that peat formed primarily under geogenous conditions (discussed above). It therefore seems likely that impermeable (sub)surface layers or soil horizons and related perched groundwater tables only affected paludification and peat growth locally, but are not responsible for large scale formation of peat in the studied region.

## Human influence

In the Northwest European Plain, the transition from hunter-gatherers to early farming communities took place gradually (Raemaekers, 1999). Analysis of the archaeological record of Fochteloërveen and its wider surroundings (Van Beek *et al.* [in prep]), shows that from the second half of the 4<sup>th</sup> millennium BCE onwards (~5,500 cal y BP) agrarian communities had settled in multiple parts of the region. This is supported by palynological evidence, which shows a decline in arboreal pollen and the emergence of cereal pollen from ~5,500 cal y BP onwards (Van Beek *et al.* [in prep]). Also for other areas in the northern Dutch coversand landscape an increase in human landscape impact was recorded around ~5,500 cal y BP. For instance, palynological investigations in the Bargerveen peat remnant showed that human influences can be traced in arboreal pollen data from 5,500 cal years

BP onwards (Dupont, 1986). Also for the Gietsenveentje peatland agricultural activity increased at this time (Bakker, 2003). Arboreal pollen further decreases until the lowest percentage is reached during the Roman Period (~1,700 cal y BP) (Van Beek *et al.* [in prep]).

Zagwijn (1986) and Spek (2004: 116–117) suggested that deforestation by humans may have contributed to a rise in groundwater tables due to a decrease in evapotranspiration following forest clearing. The decline in arboreal pollen (Van Beek *et al.* [in prep]) may reflect a combined effect of deforestation and a natural decrease in forest cover due to groundwater levels that rose in response to non-human factors (see discussion in the section below). Deforestation could be an additional factor that facilitated peat growth. I expect that deforestation could have played a role during the second phase of peat growth (fig. 5.1b).

#### Hypothesis on steering factors for peat growth

In the following section I present a hypothesis of the process and steering factors for peat growth (visualised schematically in fig. 5.2), based on the discussion above and inspired by the outcomes of this thesis.

During the Bølling-Allerød, minerogenous peat began to grow primarily in valleys (alluvial settings) during a first phase of peat growth (fig. 5.2a; Chapter 2). During the Younger Dryas and Greenlandian Stage peat initiation and/or lateral expansion halted (fig. 5.2b; Chapter 2). Peat that formed in valleys potentially degraded during this period as a result of oxidation. In the Northgrippian Stage, peat growth restarted (fig. 5.2c; Chapter 2, 3 and 4). The climatic conditions during the hypsithermal facilitated peat growth. In areas close to the coast, minerogenous peat formation is directly controlled by sea level rise (e.g. Törnqvist and Hijma, 2012). A higher position in the landscape will reduce the relative importance of sea level for peat formation; topographic factors (depth of valley incision and distance between valleys) are then progressively more influential. In a landscape setting where the distance between valleys is relatively small, minerogenous peat will form in valleys. Due to relative proximity of the watershed limit to the valley, drainage is sufficient to prevent formation of watershed mires on the small plateaus between the valleys. In a landscape setting where distance between valleys is relatively large (as in fig. 5.2), peat will form on the plateaus due to poor drainage at the watershed limits. Peat formation in valleys is generally considered as one of the processes leading to poor drainage in the higher areas of the Dutch coversand landscape and ultimately as a facilitating factor in the formation of peat (Vos, 2015a: 58; Douwes and Straathof, 2019). Holocene sea level rise during the Northgrippian, in combination with poor drainage due to growing peat deposits in valleys, and a potential facilitating effect caused by deforestations, led to a rise in regional groundwater levels. This caused a groundwater influence near



**Figure 5.2:** Schematic hypothesis regarding the factors and processes responsible for the development of peatlands in the coversand landscape of the Northwest European mainland (for more information see text).

the surface of the plateaus and resulting minerogenous peat formation (Chapter 3 and 4). Evapotranspiration of a peat-covered landscape is lower than the evapotranspiration of forests (Van der Velde *et al.*, 2021), leading to a positive feedback loop for peat formation and a shift from a forested landscape to a peat landscape. This is supported by descriptions of remnants of a drowned oak forest adjacent to the Fochteloërveen nature reserve (De Telegraaf, 1897; Popping, 1935: 21, 39–43). Initially, conditions on the well-drained plateau slopes remained favourable for forests. In the course of time, and as sea level continued to rise, valley mires grew beyond the valley sides and mires on the plateaus expanded laterally, eventually resulting in an extensive composite peat landscape (fig. 5.2d). Vertical peat growth beyond groundwater level on the plateaus caused ombrotrophication and resulted in (asynchronous) fen-bog transitions (Chapter 3; not shown in fig. 5.2d).

# 5.4 Conclusions

In this thesis I identified two key research deficits (Chapter 1), regarding the need for (I) methodological developments to constrain the spatio-temporal development of peatlands more accurately, and (II) improved understanding of the timing, pace and pattern of the initiation and lateral development of peatlands in the coversand landscape of the Northwest European mainland. The main contributions of this thesis to resolving these research deficits are listed below.

With respect to methodological advances:

- A workflow for reuse of legacy radiocarbon dates in peatland studies was developed (Chapter 2). This workflow includes a rigorous quality assessment, where a penalty is assigned to each date based on criteria that consider taphonomic quality (i.e., sample provenance) and dating quality (i.e., sample material and method used). The workflow and quality assessment can be tailored to specific research questions and study regions.
- A conceptual framework was provided (Chapter 3) that supports the use of the organic matter (OM) gradient for a quantitative and reproducible definition of the mineral-to-peat transition (i.e., the stratigraphical range reflecting the timespan of the peat initiation process) and the layer defined as *basal peat* (i.e., the stratigraphical layer that is defined as the bottom of a peat deposit). A list of recommendations for dating peat initiation was presented based on detailed analyses of three radiocarbon sequences (Chapter 3).

• An approach for modelling peat initiation and lateral expansion through time and space was presented (Chapter 4), where digital soil mapping techniques are applied using one out of two covariate options. Key benefits are that the approach results in a map of peat initiation ages (spatially explicit) and that it provides a quantitative evaluation of the prediction using the standard deviation. The use of total thickness of organic deposits as covariate is only useful where remnant peat survived, whereas the constructed covariate on groundwater-fed wetness may ultimately be applied to reconstruct peat initiation ages and lateral peatland expansion beyond the limits of peat remnants.

With respect to the initiation and lateral development of peatlands in the coversand landscape of the Northwest European mainland:

- Meta-analyses of legacy data show a bimodal distribution of basal peat dates (Chapter 2). The first phase of peat initiation (and lateral expansion) started during the Late Pleniglacial at ~16,000 cal y BP, peaked at ~14,000 cal y BP, and lasted until ~12,000 cal y BP. The second phase started ~9,000 cal y BP and continued until ~1,000 cal y BP, with a peak at approximately ~4,500 cal y BP. River valleys were subject to Late Glacial peat growth, which continued during the Holocene. Peat growth on plains and ridges started much later at ~6,000 cal y BP and lasted to ~2,000 cal y BP. Topographic depressions such as pingo remnants and deflations in coversand are characterized by an erratic pattern of peat growth through time.
- Landscape scale peat initiation in the Fochteloërveen area took place at several loci during the Early Holocene (>7000 cal y BP). As time progressed, the rate of lateral expansion began to accelerate, with the most pronounced period of expansion between 5,500 3,500 cal y BP. Sustained lateral expansion resulted in a peat cover for half of the Fochteloërveen area by 4,000 3,500 cal y BP, and for nearly the entire area between 2,500 900 cal y BP (Chapter 4). The dating evidence and modelling results for Fochteloërveen fit within the second phase of peat growth (Chapter 2).
- Biostratigraphical analyses (Chapter 3 and 4) demonstrate that the initial peat-forming vegetation at Fochteloërveen was mesotrophic and the majority of the peat initiation samples could be classified as geogenous. These findings indicate that the initial formation of peat took place primarily under the influence of groundwater and surface runoff. During the timespan of the peat initiation process, water tables probably fluctuated with alternating drier and wetter conditions. The mineral-to-peat transition was later in time followed by a fen-bog transition (Chapter 3).

Paludification was the most prominent process of peat formation in the northern Dutch coversand landscape (and Fochteloërveen), with local occurrence of terrestrialisation sites. A hypothesis regarding the factors and processes responsible for the development of peatlands in the coversand landscape of the Northwest European mainland was presented to offer direction for future research. Based on combined findings from Chapters 2 – 4 and the synthesis in Chapter 5, I tentatively conclude that the first phase of peat growth was mostly driven by climate, whereas the second was probably the result of climatic conditions favourable to peat growth in combination with Holocene sea level rise and related rise in groundwater levels.

# 5.5 Recommendations for future research

In this section I first discuss recommendations for methodological advances and subsequently directions for future peatland research in the Northwest European mainland.

# 5.5.1 Recommendations for methodological developments

Regarding methodological advances, I see three developments that have great potential for future research.

# Ages of carbon compounds

The first one (associated with Chapter 3) is directed at gaining more knowledge on the ages of the various carbon compounds in the mineral-to-peat transition. The source of the younger carbon in the humin fraction, and the cause for the incoherence that was recorded in the humin dates, remained unclear. A potential explanation is the effect of inclusion of roots in the bulk samples, which were not removed prior to pre-treatment for radiocarbon dating. Downgrowth of roots could be a relevant factor especially where (fen) peat accumulates slowly (Streif, 1972; Törnqvist *et al.*, 1992), as is the case in the studied cores during the mineral-to-peat transition. Additional dating may shed light on the influence of in- or excluding roots in bulk samples, and may help to gain more insight in the mobility of different carbon fractions and their source within the peat profile. Research on this topic has already started in a collaboration by Sanne W.L. Palstra (Centre for Isotope Research, University of Groningen), Marjolein van der Linden (BIAX Consult, Zaandam), and the author of this PhD thesis. So far, the dating procedure has been completed for this study and the results await further analyses.

## Dating the pre-peat landscape

The second recommendation (also related to Chapter 3) proposes an alternative way to constrain the age of peat initiation by making use of Optically Stimulated Luminescence (OSL) dating (see e.g. Preusser *et al.*, 2008 for more background on OSL dating, and Galbraith and Roberts, 2012 for more information on age models such as the Minimum Age Model (MAM), referred to below).

Dating bioturbation processes is a recently developed application of OSL (Reimann et al., 2017). As OSL involves dating of the mineral substrate, it cannot date peat deposits directly as is done in radiocarbon dating. However, it could be used to obtain the age when bioturbation in the mineral soil underlying peat deposits ceased due to paludification. Peat formation started as a result of paludification, and as such OSL ages could provide terminus-post-quem dates for peat initiation. An initial exploration of this concept was performed during MSc thesis research by Koudijs (2022; supervised by Jakob Wallinga and the author of this PhD thesis). For this purpose, equivalent dose distributions were analysed as function of depth for samples obtained from directly underneath the mineral-to-peat transition. Vertical series of MAM ages, and percentages of grains that could be assigned to the MAM over the investigated depths, point towards termination of bioturbation. The OSL results were compared with radiocarbon ages that were obtained in Chapter 3 and 4 of this thesis and correspond quite well. These promising results show the potential of this approach for constraining the timing of peat initiation. As this method does not require preservation of organic material, it could potentially be applied to obtain age information on the timing of paludification in areas where the peat cover is lost or highly degraded, given that the mineral substrate is intact. To develop this method further, dating studies of additional OSL cores of Fochteloërveen (for which complementary radiocarbon dating evidence is already available) would be valuable to gain more insights in the responsible bioturbation processes and potential confounding effects through other processes of soil formation, for instance cessation of bioturbation caused by acidification and podzolization. As a next step, the method could be applied to a mineral soil with a degraded peat cover, to test the potential for obtaining dating evidence beyond peat remnants.

## Model extrapolation

A third recommendation (building on Chapter 4) is to extrapolate the model that builds on groundwater-fed wetness as predictor of peat initiation age beyond peat remnants. The isohypse pattern must resemble the natural (i.e., not artificially drained) groundwater level as closely as possible for this model to reliably predict peat initiation, as was emphasized in Chapter 4 and section 5.2.3. As the current groundwater levels outside Fochteloërveen (and probably also in the vicinity of other peat remnants) are artificially lowered, their pattern cannot be used for extrapolation. Hydrological modelling techniques could offer a solution for this problem, to obtain the natural (not human-modified) groundwater levels based on Pleistocene surface topography, hydraulic conductivity of the mineral subsurface, Holocene climate data and data on Holocene sea-level rise. While modelling the natural hydrological pattern, it is important to keep in mind that peatland development may cause complications, for example where peat growth in valleys changes regional base level or where the formation of peat domes affects drainage divides.

If the natural isohypse pattern of the landscape surrounding a peat remnant could be derived through hydrological modelling techniques, the digital soil mapping approach based on groundwater-fed wetness could provide new insights in peatland initiation and lateral development in regions where this information cannot be collected from the field. Additionally, this may improve process understanding and provide insights in the hydrological and climatological boundary conditions that allowed peat formation to take place. As this involves modelling peat initiation ages based on modelled hydrological data (with related uncertainties), uncertainty propagation techniques (e.g. Heuvelink, 2018) might be useful to constrain the resulting (un)certainty of the reconstructed ages of peat initiation.

In Chapter 1 and Chapter 4 I discussed three approaches for reconstructing spatio-temporal peatland dynamics. Unfortunately, so far no numerical models seem to be available that can model the lateral dimension of peat growth (see e.g. the comprehensive discussion of peat models by Baird *et al.*, 2012). If in the future numerical peatland models would be expanded with functionalities to model lateral expansion, the digital soil mapping approach proposed in Chapter 4 could potentially be used for validating these numerical modeling outputs.

# 5.5.2 Recommendations for peatland research in the Northwest European mainland

I see several opportunities for further study, considering the palaeoenvironment of Fochteloërveen, steering factors for peat growth and the genesis of composite peat landscapes in the Northwest European mainland, and integration of peatland research in interdisciplinary contexts.

#### Palaeoenvironment of Fochteloërveen

As discussed in section 5.3.1, the highly abundant presence of charred organic material in the investigated levels of the Fochteloërveen cores suggests that (local) fires must have occurred regularly. Further study of physical and chemical properties of the organic matter and ash content of the cores may shed more

light on ignition conditions and type of burning (flaming or smouldering). This may be relevant for understanding the time span of peat initiation (to what degree is organic material lost due to fires), for characterizing the interplay with hydrological processes, and could have climatological implications (e.g. Zaccone *et al.*, 2014). In addition, it may be relevant for interdisciplinary integrations that consider human landscape influence (see below). For more information on (palaeo)fires in peatlands see e.g. Nelson *et al.* (2021).

#### Steering factors for peat growth and genesis of composite peat landscapes

A key direction for future research is to improve understanding of the steering factors for peat growth in the coversand landscape of the Northwest European mainland. The hypothesis presented in section 5.3.3 offers guidance for this. Additionally, the recommendation to use hydrological modelling to extrapolate the digital soil mapping approach beyond peat remnants (sections 5.2.3 and 5.5.1) offers a starting point to move towards understanding of processes and feedbacks in the development of peat landscapes in this region.

Another element related to this, is to expand the research scope to include both peatlands that developed on non-coastal and non-alluvial topographic plateaus, and peatlands that developed in valleys (alluvial settings) that intersect these plateaus. Study of the landscape connections between peat on plateaus and valley peat, may yield new knowledge and an integrated understanding of the genesis of the former extensive composite peat landscapes of the Northwest European mainland. As an initial step, the stratigraphy and chronology of one of the upper reaches of the Peizerdiep, named the Slokkert (i.e., draining water from the Fochteloërveen in northern direction), was studied by Jasper Candel and the author of this PhD thesis. The obtained ages along the valley side (unpublished data) fall within the Late Glacial period and mostly within the Bølling-Allerød interstadial. The development of peat during this period fits within the first phase of peat growth (fig. 5.1b). In addition, two dates suggest that a renewed phase of peat growth started approximately synchronous with the 8,2 cal y BP event. Further study of this site, especially a vertical sequence of radiocarbon dates, may shed more light on the timeframe that is captured in the peat deposit in the valley, which may reflect both peat growth phases of fig. 5.1b.

Lastly, data rescue and meta-analysis for other major bog remnants in the Northwest European mainland may help to further substantiate the regional character of the phases of peat growth (fig. 5.1b) and consequently increase understanding of the steering factors for peat growth.

## Interdisciplinary integration

A key interdisciplinary integration is of peatland palaeogeography and palaeoecol-

ogy with archaeological data on human land use and habitation. The interplay between peatland initiation and lateral expansion, human habitation and vegetation dynamics is not well understood (Van Beek *et al.*, 2015; Van Beek, 2015). A study that aims to contribute to this knowledge deficit is currently in preparation (Van Beek *et al.* [in prep]) and uses the Fochteloërveen area (including both the peat remnant and adjacent mineral soils) as a case study (incorporating findings from Chapters 3 and 4). Main aims are to analyse how the initiation and lateral expansion of the Fochteloërveen peatland relates to human habitation patterns through time and space, how the vegetation of the area (locally and regionally) developed over time, and in which ways humans contributed to landscape or vegetation changes. The period between the Late Palaeolithic and the Early High Middle Ages (~15,000 BCE – 1,000 CE) forms the scope of the study.

In addition to the interdisciplinary study mentioned above, further research on levels of the Fochteloërveen peat cores that contain the period from ~8,000 to 6,000 cal y BP (~6,000 to 4,000 BCE; e.g. cores S9, S10, S16, S17, S19) may help to further constrain the Mesolithic to Neolithic transition and agricultural activity of the Swifterbant and Funnel Beaker cultures (Raemaekers, 1999; Bakker, 2003).

The abovementioned suggestions which focus on the Fochteloërveen, may serve as examples for similar interdisciplinary peatland studies in the wider Northwest European mainland. The information that can be obtained from the peat archives offers unique opportunities to analyse the environmental context of (changes in) human-landscape interactions. Many studies on archaeological finds from peatlands have a strong focus on material culture, and attention for environmental and socio-cultural contexts has been limited (cf. Chapman, 2015; Van Beek *et al.*, 2019). Interdisciplinary integration of archaeological studies with research on peatland palaeogeography and palaeoecology may therefore yield new insights in past human behaviour (Van Beek *et al.*, 2019).

# 5.6 Implications

# 5.6.1 Peatlands and ecosystem services

The transition from dryland to wetland and to peatland represents a huge landscape change that has major impacts for landscape functioning and ecosystem services. Ecosystem services provided by peatlands include amongst others carbon storage (Joosten and Couwenberg, 2008), greenhouse gas fluxes (Sirin and Laine, 2008), and biodiversity (Minayeva *et al.*, 2008). Changes in future climate, especially rising temperatures and changes in water availability, are expected to significantly affect the ecosystems services provided by peatlands (Charman *et al.*, 2008; Parish *et al.*, 2008b). Peatlands play an important role in the global carbon cycle (Yu *et al.*, 2011) and form the largest long-term carbon store in the terrestrial biosphere (Joosten and Couwenberg, 2008). Peatland extent forms a fundamental boundary condition in the carbon dynamics of peatlands (Loisel *et al.*, 2013) and in calculations of peatland carbon fluxes the spatial extent of peatlands is a fundamental parameter (Korhola *et al.*, 2010; Yu *et al.*, 2010). Reconstructing the development and size of carbon storage in peatlands throughout the Holocene is an active field of research and debate, as was illustrated recently by the differences in estimating the carbon storage of northern peatlands between the studies by Nichols and Peteet (2019) and Ratcliffe *et al.* (2021). Increased understanding of the age and palaeogeography of peatlands (as in this thesis) may contribute to these ongoing developments and is vital for improving the accuracy of models on past, present and future carbon storage and release.

## 5.6.2 Nature conservation and restoration

The vast losses in peat cover (Chapter 1) and continuous threats to the survival of peatlands in Europe and elsewhere (e.g. Bragazza *et al.*, 2006; Swindles *et al.*, 2019) pose the need to protect the peatland areas that are left. From a scientific point of view, these exceptional archives of the past environment may satisfy future demands for information about the past (cf. Greiser and Joosten, 2018). The study by Greiser and Joosten (2018) offers examples of how to evaluate the value of peat archives based on predefined criteria, as a way to include archive value in decisions on conservation and restoration. The age of the peat archive, as was studied in this thesis, is a key limitation on information that the archive may contain. Therefore knowledge on the timing of peat initiation (and lateral expansion) is of great significance in assessing archive content and value.

Furthermore, evidence-based narratives of palaeoenvironmental and palaeoecological development are needed as dynamic references for nature conservation and restoration goals (Gillson *et al.*, 2021). Such longer-term data is fundamental to understand natural variability and ecosystem resilience, and to contextualize changes that are observed today (Gillson *et al.*, 2021). Knowledge on the age, palaeogeography and palaeoecology of Fochteloërveen may provide this reference for nature conservation and restoration goals in this area. Current assumptions by nature conservation organisations highly underestimate the age of the Fochteloërveen peat remnant, and may be updated based on the outcomes of this thesis. For instance, Altenburg *et al.* (2017) state that Fochteloërveen developed from 800 BCE onwards, and Provincie Drenthe (2016) and Douwes and Straathof (2019) describe that its development started from approximately 1000 BCE onwards. In addition, currently only the southern central area is recognized as the old core of the peatland (Altenburg *et al.*, 2017), whereas the findings of this thesis indicate that several Early Holocene loci of peat initiation are present within the area.

In the 'Assessment on Peatlands, Biodiversity and Climate Change', Parish *et al.* (2008a,b) state that reducing drainage and improving water management in peatlands is the most important step to prevent peatland degradation, to protect biodiversity, to reduce the risk of fire and to stop carbon dioxide emissions from peat. In Fochteloërveen, nature restoration efforts from the 1980s onwards have mainly been directed at improving water retention in the area. In addition, aims were to reduce the occurrence of *Molinia caerulea* (presence of *M. caerulea* strongly increased in the second half on the twentieth century due to nitrogen deposition), and to increase the amount of vascular bog plant species and *Sphagnum* mosses (Provincie Drenthe, 2016; Altenburg *et al.*, 2017). The palaeoecological data that were obtained in this thesis (and in the work of Van Beek *et al.* [in prep]) could assist in re-evaluating the value of areas with a vegetation in which the presence of *Sphagnum* is low. Some of these mesotrophic habitats might be comparable to the palaeoenvironment during and shortly after peat initiation and as such reflect the long-term history of the Fochteloërveen.

# 5.6.3 Cultural history and heritage management

Information on the palaeogeographical and palaeoenvironmental development of peatlands plays a vital role in understanding and contextualising long-term human habitation patterns (Van Beek *et al.*, 2015), and in the analysis of single archaeological sites (either discovered in the distant past or still present in-situ; e.g. Plunkett *et al.*, 2009; Chapman *et al.*, 2019; Van Beek *et al.*, 2019). The knowledge that this thesis provides on the age, palaeogeography and palaeoenvironment of peatlands may serve as direct input for interdisciplinary studies on the changes in human-landscape interactions (section 5.5.2).

In addition, insights in the age and character of peat deposits may be highly informative for cultural heritage management purposes and archaeological prospection, as peat layers may cover or contain well-preserved waterlogged archaeological sites that are unparalleled in dryland environments. The sustainable management of cultural heritage in (former) raised bogs has received relatively little attention in the Netherlands. Several wetland-oriented heritage management projects have been issued by the Dutch Cultural Heritage Agency, but all with a focus on coastal areas (Van Dockum *et al.*, 2001). A key objective of the overarching *Home Turf* project is therefore to develop a proactive strategy for the management of cultural heritage in raised bogs (Paulissen and Van Beek [in prep]). The need to consolidate and advance cross-domain research and pressing lack for joint efforts in natural and cultural heritage management has been recognized for several decades (Buckland, 1993; Gearey and Chapman, 2004) and is still relevant today (Gearey and Everett, 2021). Recently, it has been suggested that cultural heritage should be included within the ecosystem services framework (Tengberg et al., 2012; Gearey et al., 2014; Hølleland et al., 2017), as a way to pose heritage along other potentially competing conservation interests (Gearey et al., 2014). The current management plan of Fochteloërveen (Provincie Drenthe, 2016) does not include heritage management, but restoration practices are adjusted for sites that are marked as archaeological monument. For instance, the central sand ridge ('Bonghaar') in the Fochteloërveen is not subjected to removal of sods (Dutch: plaggen) to eliminate nutrients, but grazed for this purpose. The results of this PhD thesis, along with the findings by Van Beek et al. [in prep], could offer further points of reference for integrating natural and cultural heritage management in the area. The work by Paulissen and Van Beek [in prep] provides a framework for mapping cultural remains in (former) peatlands and for integrated management of natural and cultural heritage.

# 5.6.4 Peatlands in environmental education and science communication

Peatlands are landscapes with great potential to feature in environmental education, both in formal educational contexts and in science communication with the general public. The long-term development of peatlands is relevant for students of physical geography, ecology, archaeology, history, and for applied disciplines that are concerned with landscape management such as forestry, landscape architecture, and environmental policy. In response to this thesis project and the research outcomes on the long-term development of Fochteloërveen, bog landscapes are included in a course of the earth science curriculum at WUR from the academic year 2021 – 2022 onwards, and illustrated through a field excursion to Fochteloërveen (Candel and Makaske, 2022).

Adequate training of peatland researchers is fundamental to ensure accurate and timely decision-making in nature and climate policy (Meleisea, 2022). In addition, in the policy paper by Erwin (2009), which was produced at the request of the Ramsar Convention on Wetlands, the need for education of public and private sectors about the protection and restoration of wetlands is listed as a point of action. One of the obstacles to translate scientific evidence to policy and management practice is that it is not always synthesized and communicated in a way that is useful for decision-makers (Meleisea, 2022). To make research outcomes more accessible, Van Beek *et al.* [in prep] have converted scientific results into two artist impressions, i.e., digital landscape renders created by a renowned artist based directly on research data. The artist impressions show what the surroundings of Fochteloërveen looked like during the Mesolithic and Roman Period. The outcomes of this thesis are also included in both artist impressions. The artist impressions will be presented to the Province of Drenthe and to Natuurmonumenten (the nature conservation organisation that manages the Fochteloërveen area). This approach may serve as inspiration for peatland researchers to communicate their findings to policy makers and management organisations, and also as a way to showcase research to the general public.

Peat is not renewable (IMCG, 2022) and peat landscapes reflect thousands of years of history (this thesis). The ability to tell the long-term development story of a landscape (this thesis) and the way humans interacted with it (Van Beek *et al.* [in prep]) may help to increase public support for conservation and restoration. The research of this thesis project was presented to a non-scientific audience through several outreach activities, including a television broadcast on Fochteloërveen (ROEG!, 2021), a documentary on Fochteloërveen (Story Driven, 2022), and through a series of exchanges between science and art focused on peatlands (Foster, 2022; ISRIC World Soil Museum, 2022a,b). Comparable science communication efforts on peatland research in the Northwest European mainland and elsewhere may help to show the value, history and beauty of these unique landscapes, and the importance of protecting and restoring peat remnants.

#### Acknowledgements

I highly appreciated the constructive discussions and feedback on the Synthesis provided by Jakob Wallinga, Roy van Beek and Ype van der Velde. Regarding fig. 5.2, I thank Jasper Candel and Ype van der Velde for their input. Regarding the information contained in the recommendation about OSL dating in section 5.6.1. I would like to thank Romy Koudijs and Jakob Wallinga. For the opportunity to obtain data about the Slokkert valley, which is discussed in section 5.6.2, I would like to thank Natuurmonumenten Noordenveld, Jasper Candel, Marjolein van der Linden and Lucy Kubiak-Martens (BIAX Consult, Zaandam), Sanne W.L. Palstra (Centre for Isotope Research, University of Groningen), Marcello Novani (Laboratory for Geo-information Science, WUR) and Piet Peters (Soil Hydro Physics Lab, WUR). I thank Maurice Paulissen for sharing information about a manuscript in preparation for section 5.6.3. I highly appreciate the efforts by Jasper Candel and Bart Makaske to include bog landscapes and the example of Fochteloërveen in one of the courses offered by our chair group, the people from ROEG! and Thomas Rovers from Story Driven for the opportunities to join a broadcast and documentary on Fochteloërveen respectively, and Kate Foster for including my work in the Peatland Exchanges series.

# References

- Aaby B (1986). "Palaeoecological studies of mires." In B Berglund (ed.), "Handbook of Holocene Palaeoecology and Palaeohydrology," pp. 145–164. J. Wiley, Chichester; New York.
- AHN (2018). "De details van het Actueel Hoogtebestand Nederland [Details of the Digital Elevation Model of The Netherlands]." http://ahn.maps.arcgis.com/apps/Cascade/index.html?appid= 75245be5e0384d47856d2b912fc1b7ed. Date accessed: 22/02/2018.
- AHN (2021a). "De details van het Actueel Hoogtebestand Nederland [Details of the Digital Elevation Model of The Netherlands]." https://ahn.maps.arcgis.com/apps/Cascade/index.html?appid= 75245be5e0384d47856d2b912fc1b7ed. Date accessed: 11/10/2021.
- AHN (2021b). "Kwaliteitsbeschrijving Actueel Hoogtebestand Nederland [Quality description Digital Elevation Model of The Netherlands]." https://www.ahn.nl/kwaliteitsbeschrijving. Date accessed: 11/10/2021.
- Almquist-Jacobson H, Foster DR (1995). "Toward an integrated model for raised-bog development: Theory and field evidence." *Ecology*, **76**, 2503–2516. doi:10.2307/2265824.
- Altenburg W, Bijkerk W, Douwes R, Straathof N (2017). "Neergang en opkomst van het Fochteloërveen: resultaten van 30 jaar hoogveenherstel." De Levende Natuur, **118**, 79–84.
- Alterra (2014). "Bodemkaart van Nederland 1:50.000 [digital file]." Technical report, Alterra, Wageningen UR.
- Anderberg AL (1994). Atlas of Seeds and Small Fruits of Northwest-European Plant Species; Part 4: Resedaceae-Umbelliferae. Edited by the Swedish Natural Science Research Council. Statens naturvetenskapliga forskningaraåd, Stockholm.
- Anderson RL, Foster DR, Motzkin G (2003). "Integrating lateral expansion into models of peatland development in temperate New England." *Journal of Ecology*, **91**, 68–76.
- Baird A, Morris P, Belyea L (2012). "The DigiBog peatland development model 1: rationale, conceptual model, and hydrological basis." *Ecohydrology*, 5, 242–255. doi:10.1002/eco.230.
- Bakker R (2003). The emergence of agriculture on the Drenthe Plateau. A palaeobotanical study supported by high-resolution 14C dating. Ph.D. thesis, University of Groningen.
- Bauer IE, Gignac LD, Vitt DH (2003). "Development of a peatland complex in boreal western Canada: Lateral site expansion and local variability in vegetation succession and long-term peat accumulation." *Canadian Journal of Botany*, **81**, 833–847. doi:10.1139/b03-076.
- Bayliss A (2009). "Rolling out revolution: using radiocarbon dating in archaeology." *Radiocarbon*, **51**, 123–147.
- Bayliss A, McCormac G, Van der Plicht H (2004). "An illustrated guide to measuring radiocarbon from archaeological samples." *Physics Education*, **39**, 1–8. doi:10.1088/0031-9120/39/2/001.
- Berendsen H, Stouthamer E (2001). Palaeogeographic development of the Rhine-Meuse delta, The Netherlands. Van Gorcum, Assen.
- Berendsen HJ, Makaske B, Van de Plassche O, Van Ree MH, Das S, Van Dongen M, Ploumen S, Schoenmakers W (2007). "New groundwater-level rise data from the Rhine-Meuse delta - Implications for the reconstruction of Holocene relative mean sea-level rise and differential land-level movements." *Geologie en Mijnbouw/Netherlands Journal of Geosciences*, **86**, 333–354. doi:10.1017/s0016774600023568.
- Berendsen HJA, Stouthamer E (2000). "Late Weichselian and Holocene palaeogeography of the Rhine-Meuse delta, the Netherlands." Palaeogeography, Palaeoclimatology, Palaeoecology, 161, 311–335. doi:10.1016/S0031-0182(00)00073-0.
- Berggren G (1969). Atlas of Seeds and Small Fruits of Northwest-European Plant Species; Part 2: Cyperaceae. Edited by the Swedish Natural Science Research Council. Statens naturvetenskapliga

forskningaraåd, Stockholm.

- Berggren G (1981). Atlas of Seeds and Small Fruits of Northwest-European Plant Species; Part 3: Salicaceae-Cruciferae. Edited by the Swedish Natural Science Research Council. Statens naturvetenskapliga forskningaraåd, Stockholm.
- Blaauw M, Van der Plicht J, Van Geel B (2004). "Radiocarbon dating of bulk peat samples from raised bogs: Non-existence of a previously reported 'reservoir effect'?" *Quaternary Science Reviews*, 23, 1537–1542. doi:10.1016/j.quascirev.2004.04.002.
- Booth RK, Lamentowicz M, Charman DJ (2010). "Preparation and analysis of testate amoebae in peatland palaeoenvironmental studies." *Mires and Peat*, **7**, 1–7.
- Bos I (2010). "Architecture and facies distribution of organic-clastic lake fills in the fluviodeltaic Rhine-Meuse system, The Netherlands." *Journal of Sedimentary Research*, **80**, 339–356. doi:10.2110/jsr.2010.035.
- Bos IJ, Busschers FS, Hoek WZ (2012). "Organic-facies determination: A key for understanding facies distribution in the basal peat layer of the Holocene Rhine-Meuse delta, The Netherlands." Sedimentology, 59, 676–703. doi:10.1111/j.1365-3091.2011.01271.x.
- Bosch J (1990). Assen West (12W), Assen Oost (12O). Toelichtingen bij de geologische kaart van Nederland 1:50.000. Rijks Geologische Dienst, Haarlem.
- Bragazza L, Freeman C, Jones T, Rydin H, Limpens J, Fenner N, Ellis T, Gerdol R, Hájek M, Hájek T, Iacumin P, Kutnar L, Tahvanainen T, Toberman H (2006). "Atmospheric nitrogen deposition promotes carbon loss from peat bogs." *Proceedings of the National Academy of Sciences of the United States* of America, 103, 19386–19389. doi:10.1073/pnas.0606629104.
- Breiman L (2001). "Random Forests." Machine Learning, **45**, 5–32. doi:https://doi.org/10.1023/A:1010933404324.
- Breuning-Madsen H, Bird KL, Balstrøm T, Elberling B, Kroon A, Lei EB (2018). "Development of plateau dunes controlled by iron pan formation and changes in land use and climate." *Catena*, **171**, 580–587. doi:10.1016/j.catena.2018.07.011. URL https://doi.org/10.1016/j.catena.2018.07.011.
- Brock F, Lee S, Housley RA, Ramsey CB (2011). "Variation in the radiocarbon age of different fractions of peat: A case study from Ahrenshöft, northern Germany." *Quaternary Geochronology*, 6, 550–555. doi:10.1016/j.quageo.2011.08.003. URL http://dx.doi.org/10.1016/j.quageo.2011.08.003.
- Buckland PC (1993). "Peatland archaeology: a conservation resource on the edge of extinction." Biodiversity and Conservation, 2, 513–527.
- Cambridge Dictionary (2020). "Meaning of quality in English." https://dictionary.cambridge.org/ dictionary/english/quality. Date accessed: 15/09/2020.
- Candel J, Makaske B (2022). "High-lying peat and peat colonial landscape." In B Makaske, I Lubbers, J Candel (eds.), "Field Guide for Geology, Soils and Landscapes of the Northwest European lowlands (SGL 22306)," pp. 31–36. Wageningen University, Wageningen.
- Candel JH, Makaske B, Storms JE, Wallinga J (2017). "Oblique aggradation: a novel explanation for sinuosity of low-energy streams in peat-filled valley systems." *Earth Surface Processes and Landforms*, 42, 2679–2696. doi:10.1002/esp.4100.
- Cappers RTJ, Bekker RM, Jans JEA (2006). *Digitale zadenatlas van Nederland*. Barkhuis Publishing & Groningen University Library.
- Casparie W (1985). "De twee IJzertijd houten veenwegen I(SM) en II(SM) bij de Suermondswijk te Smilde." *Nieuwe Drentse Volksalmanak*, **102**, 145–169.
- Casparie W (1987). "Bog trackways in the Netherlands." Palaeohistoria, 29, 35-65.
- Casparie W, Streefkerk J (1992). "Climatological, stratigraphic and palaeo ecological aspects of mire development." In JTA Verhoeven (ed.), "Fens and Bogs in the Netherlands: Vegetation, History, Nutrient Dynamics and Conservation," pp. 81–130. Kluwer Academic Publishers, Dordrecht. ISBN 9789401579971. doi:10.1007/978-94-015-7997-1.
- Casparie WA (1972). "Bog development in southeastern Drenthe (The Netherlands)." Vegetatio, 24, 1–271. doi:10.1007/BF02675415.
- Casparie WA (1993). "The Bourtanger Moor: Endurance and vulnerability of a raised bog system." *Hydrobiologia*, **265**, 203–215. doi:10.1007/BF00007269.
- Chambers F, Beilman D, Yu Z (2011). "Methods for determining peat humification and for quantifying peat bulk density, organic matter and carbon content for palaeostudies of climate and peatland carbon dynamics." *Mires and Peat*, 7, 1–10. URL http://www.mires-and-peat.net/pages/volumes/map07/ map0707.php.
- Chapman H (2015). "The landscape archaeology of bog bodies." *Journal of Wetland Archaeology*, **15**, 109–121.
- Chapman H, Gearey BR, Bamforth M, Bermingham N, Marshall P, Powlesland I, Taylor M, Whitehouse N (2013). *Modelling archaeology and palaeoenvironments in wetlands: the hidden landscape archaeology of Hatfield and Thorne Moors, eastern England*. Oxbow Books. ISBN 9781782971740. URL https://www.oxbowbooks.com/oxbow/modelling-archaeology-and-palaeoenvironments-in-wetlands.html.
- Chapman H, Van Beek R, Gearey B, Jennings B, Smith D, Nielsen NH, Elabdin ZZ (2019). "Bog bodies in context: Developing a best practice approach." *European Journal of Archaeology*, pp. 1–23. doi:10.1017/eaa.2019.54.
- Charman D (2002a). "Autogenic Change." In D Charman (ed.), "Peatlands and Environmental Change," chapter 7, pp. 143–165. John Wiley & Sons, Ltd, Chichester.
- Charman D (2002b). "Origins and Peat Initiation." In D Charman (ed.), "Peatlands and Environmental Change," chapter 4, pp. 73–91. John Wiley & Sons, Ltd, Chichester.
- Charman D (2002c). "Peat and peatlands." In D Charman (ed.), "Peatlands and Environmental Change," chapter 1, pp. 3–23. John Wiley & Sons, Ltd, Chichester.
- Charman D (2002d). "Peat landforms and structure." In D Charman (ed.), "Peatlands and Environmental Change," chapter 2, pp. 24–38. John Wiley & Sons, Ltd, Chichester.
- Charman D (2002e). Peatlands and Environmental Change. John Wiley & Sons, Ltd, Chichester. ISBN 0471969907.
- Charman D, Laine J, Minayeva T, Sirin A (2008). "Impacts of future climate change on peatlands." In F Parish, A Sirin, D Charman, H Joosten, T Minayeva, M Silvius, L Stringer (eds.), "Assessment on Peatlands, Biodiversity and Climate Change: Main Report," pp. 139–154. Global Environment Centre, Kuala Lumpur and Wetlands International, Wageningen.
- Cohen KM, Gibbard PL, Weerts HJ (2014). "North Sea palaeogeographical reconstructions for the last 1 Ma." *Geologie en Mijnbouw/Netherlands Journal of Geosciences*, **93**, 7–29. doi:10.1017/njg.2014.12.
- Coles J, Lawson A (1987). European Wetlands in Prehistory. Oxford University Press (Clarendon Press), Oxford. ISBN 9780198134060.
- Cook G, Van der Plicht J (2013). "Radiocarbon Dating | Conventional Method." In S Elias, C Mock (eds.), "Encyclopedia of Quaternary Science," pp. 305–315. Elsevier, second edition. ISBN 9780444536426. doi:https://doi.org/10.1016/B978-0-444-53643-3.00048-0.
- Cook GT, Dugmore AJ, Shore JS (1998). "The influence of pretreatment on humic acid yield and 14C age of Carex peat." *Radiocarbon*, **40**, 21–27. doi:10.1017/s0033822200017835.
- Crawford RM, Jeffree CE, Rees WG (2003). "Paludification and forest retreat in northern oceanic environments." Annals of Botany, 91, 213–226. doi:10.1093/aob/mcf185.
- Crushell P, Connolly A, Schouten M, Mitchell FJ (2008). "The changing landscape of Clara Bog: The history of an Irish raised bog." Irish Geography, 41, 89–111. doi:10.1080/00750770801915596.
- Cubizolle H, Bonnel P, Oberlin C, Tourman A, Porteret J (2007). "Advantages and limits of radiocarbon dating applied to peat inception during the end of the Late Glacial and the Holocene: The example of mires in the eastern Massif Central (France)." Quaternaire, 18, 187–208. doi:10.4000/quaternaire.1049.
- De Bakker H, Schelling J (1966). Systeem van bodemclassificatie voor Nederland: De hogere niveaus. Stichting voor Bodemkartering, Pudoc, Wageningen.
- De Gans W (1976). "Dobben op het Drents plateau." K.N.A.G. Geografisch Tijdschrift, (5), 354-364.
- De Gans W, Sohl H (1981). "Weichselian pingo remnants and permafrost on the Drenthian plateau (the Netherlands)." Geologie en Mijnbouw, 60, 447–452.
- De Telegraaf (1897). "Een Oerwoud." Date of newspaper: 07/07/1897.
- De Vries H (1958). "Variation in concentration of radiocarbon with time and location on earth." *Proceedings KNAW*, **B61**, 1–9.
- Dee MW, Palstra SW, Aerts-Bijma AT, Bleeker MO, Bruijn SD, Ghebru F, Jansen HG, Kuitems M, Paul D, Richie RR, Spriensma JJ, Scifo A, Van Zonneveld D, Verstappen-Dumoulin BM, Wietzes-Land P, Meijer HA (2020). "Radiocarbon dating at Groningen: New and updated chemical pretreatment procedures." *Radiocarbon*, **62**, 63–74. doi:10.1017/RDC.2019.101.
- DINOloket TNO (2022). "Ondergrondgegevens, Geologisch Booronderzoek." https://www.dinoloket. nl/. Date accessed: 2022-04-19.
- Douwes R, Straathof N (2019). "Het Fochteloërveen." In A Jansen, A Grootjans (eds.), "Hoogvenen: Landschapsecologie, behoud, beheer, herstel," chapter 14, pp. 133–147. Noordboek Natuur, Gorredijk.

- Dupont L (1986). "Temperature and rainfall variation in the Holocene based on comparative palaeoecology and isotope geology of a hummock and a hollow (Bourtangerveen, The Netherlands)." *Review of Palaeobotany and Palynology*, **48**, 71–159. doi:10.1016/0034-6667(86)90056-4.
- Edvardsson J, Poska A, Van der Putten N, Rundgren M, Linderson H, Hammarlund D (2014). "Late-Holocene expansion of a south Swedish peatland and its impact on marginal ecosystems: Evidence from dendrochronology, peat stratigraphy and palaeobotanical data." *Holocene*, **24**, 466–476. doi:10.1177/0959683613520255.
- Eijkelkamp Agrisearch Equipment (2022). "Percussion drilling equipment." https://en.eijkelkamp.com/ products/augering-soil-sampling-equipment/percussion-drilling-set-gasoline-percussion-hammer. html. Date accessed: 16/02/2022.
- Eijkelkamp Soil & Water (2018). "Peat sampler." *Technical report*, Eijkelkamp Soil & Water, Giesbeek. URL http://www.vanwalt.com/sediment-sampling-gouges-augers.html#PeatSamplerSet.
- Erkens G, Van der Meulen MJ, Middelkoop H (2016). "Double trouble: subsidence and CO2 respiration due to 1,000 years of Dutch coastal peatlands cultivation." *Hydrogeology Journal*, **24**, 551–568. doi:10.1007/s10040-016-1380-4.
- Erwin KL (2009). "Wetlands and global climate change: The role of wetland restoration in a changing world." Wetlands Ecology and Management, **17**, 71–84. doi:10.1007/s11273-008-9119-1.
- ESRI (2022). "Charted Territory Map." https://www.arcgis.com/home/item.html?id= d582a9e953c44c09bb998c7d9b66f8d4.
- Everts FH, Baaijens GJ, Grootjans AP, Verschoor AJ, de Vries NPJ (2002). Hoogveenontwikkeling in veentjes en kleinschalige hoogveencomplexen op het Dwingelerveld: een landschapsbenadering. Deel 2: Landschapsontwikkeling en hydrologie. Expertisecentrum LNV, Ministerie van Landbouw, Natuur en Voedselkwaliteit, Ede.
- Fokkens H (1998). Drowned landscape The occupation of the western part of the Frisian-Drentian Plateau, 4400 BC - AD 500. Van Gorcum & Comp. BV and Rijksdienst voor het Oudheidkundig Bodemonderzoek (ROB), Assen. ISBN 9023233050.
- Foster D, Wright H, Thelaus M, King G (1988). "Bog development and landform dynamics in Central Sweden and South-Eastern Labrador, Canada." *Journal of Ecology*, **76**, 1164–1185.
- Foster K (2022). "Artist's project highlighting diverse cultural values of peatlands." https://peatcultures. wordpress.com/.
- Frenzel B (1983). "Mires Repositories of climatic information or self-perpetuating ecosystems?" In A Gore (ed.), "Mires: Swamp, bog, fen and mor. General Studies. Ecosystems of the World 4A," pp. 35–65. Elsevier.
- Frolking S, Roulet NT, Tuittila E, Bubier JL, Quillet A, Talbot J, Richard PJH (2010). "A new model of Holocene peatland net primary production, decomposition, water balance, and peat accumulation." *Earth System Dynamics*, 1, 1–21. doi:10.5194/esd-1-1-2010.
- Galbraith RF, Roberts RG (2012). "Statistical aspects of equivalent dose and error calculation and display in OSL dating: An overview and some recommendations." *Quaternary Geochronology*, **11**, 1–27. doi:10.1016/j.quageo.2012.04.020.
- Gearey BR, Chapman HP (2004). "Towards realising the full archaeoenvironmental potential of raised (ombrotrophic) mires in the British Isles." Oxford Journal of Archaeology, 23, 199–208. doi:10.1111/j.1468-0092.2004.00209.x.
- Gearey BR, Everett R (2021). "Running out of time? Peatland rehabilitation, archaeology and cultural ecosystem services." *Mires and Peat*, **27**, 1–6. doi:10.19189/MaP.2021.KHR.StA.2195.
- Gearey BR, Fletcher W, Fyfe R (2014). "Managing, valuing, and protecting heritage resources in the twenty-first century: Peatland archaeology, the ecosystem services framework, and the Kyoto protocol." *Conservation and Management of Archaeological Sites*, **16**, 236–244. doi:10.1179/1350503315Z.0000000084.
- Gerding M (1995). Vier eeuwen turfwinning, De verveningen in Groningen, Friesland, Drenthe en Overijssel tussen 1550 en 1950. Ph.D. thesis, Wageningen University, Wageningen.
- Gil Y, David CH, Demir I, Essawy BT, Fulweiler RW, Goodall JL, Karlstrom L, Lee H, Mills HJ, hyun Oh J, Pierce SA (2016). "Toward the Geoscience Paper of the Future: Best practices for documenting and sharing research from data to software to provenance." *Earth and Space Science*, pp. 1–28. doi:10.1002/2015EA000136.Received.
- Gillson L, Dirk C, Gell P (2021). "Using long-term data to inform a decision pathway for restoration of ecosystem resilience." *Anthropocene*, **36**. doi:10.1016/j.ancene.2021.100315.

Givelet N, Roux GL, Cheburkin A, Chen B, Frank J, Goodsite ME, Kempter H, Krachler M, Noernberg T, Rausch N, Rheinberger S, Roos-Barraclough F, Sapkota A, Scholz C, Shotyk W (2004).
"Suggested protocol for collecting, handling and preparing peat cores and peat samples for physical, chemical, mineralogical and isotopic analyses." *Journal of Environmental Monitoring*, 6, 481–492. doi:10.1039/b401601g.

Glob P (1969). The Bog People: Iron Age Man Preserved. Faber & Faber, London.

- Godwin H (1981). The archives of the peat bogs. Cambridge University Press, Cambridge. ISBN 9780521237840.
- Goh K, Molloy B (1978). "Radiocarbon dating of paleosols using soil organic matter components." *Journal of Soil Science*, **29**, 567–573. doi:10.1111/j.1365-2389.1978.tb00805.x.
- Gorham E, Lehman C, Dyke A, Janssens J, Dyke L (2007). "Temporal and spatial aspects of peatland initiation following deglaciation in North America." *Quaternary Science Reviews*, 26, 300–311. doi:10.1016/j.quascirev.2006.08.008.
- Greiser C, Joosten H (2018). "Archive value: measuring the palaeo-information content of peatlands in a conservation and compensation perspective." *International Journal of Biodiversity Science, Ecosystem Services & Management*, **14**, 209–220. doi:10.1080/21513732.2018.1523229.
- Griffin R (2015). "When are Old Data New Data?" *GeoResJ*, **6**, 92–97. doi:10.1016/j.grj.2015.02.004. URL http://dx.doi.org/10.1016/j.grj.2015.02.004.
- Groenendijk H (1997). Op zoek naar de horizon Het landschap van Oost-Groningen en zijn bewoners tussen 8000 voor Chr. en 1000 na Chr. REGIO-PRojekt Uitgevers, Groningen. ISBN 9050280838.
- Groenendijk H (2003). "New archaeological issues in the former Bourtanger Moor (The Netherlands)." In A Bauerochse, H Haßmann (eds.), "Peatlands, archaeological sites, archives of nature, nature conservation, wise use - Proceedings of the Peatland Conference 2002 in Hannover, Germany," pp. 36–47. Leidorf.
- Grondwatertools TNO (2022). "LHM laag 1, berekeningswijze: LHM." https://www.grondwatertools. nl/gwsinbeeld/. Date accessed: 2022-01-24.
- Hammond AP, Goh KM, Tonkin PJ, Manning MR (1991). "Chemical pretreatments for improving the radiocarbon dates of peats and organic silts in a gley podzol environment: Grahams Terrace, North Westland." New Zealand Journal of Geology and Geophysics, 34, 191–194. doi:10.1080/00288306.1991.9514456.
- Heuvelink G (2018). "Uncertainty and uncertainty propagation in soil mapping and modelling." In A McBratney, B Minasny, U Stockmann (eds.), "Pedometrics," Springer. ISBN 9783319634371.
- Hijma M (2009). From river valley to estuary. PhD thesis, Netherlands Geographical Studies 389. Koninklijk Nederlands Aardrijkskundig Genootschap, Faculteit Geowetenschappen Universiteit Utrecht. ISBN 9789068094329.
- Hijma MP, Cohen KM (2019). "Holocene sea-level database for the Rhine-Meuse Delta, The Netherlands: Implications for the pre-8.2 ka sea-level jump." *Quaternary Science Reviews*, **214**, 68–86. doi:https://doi.org/10.1016/j.quascirev.2019.05.001.
- Hilbert D, Roulet N, Moore T (2017). "Modelling and analysis of peatlands as dynamical systems." Journal of Ecology, 84, 617–628.
- Holmquist JR, Finkelstein SA, Garneau M, Massa C, Yu Z, MacDonald GM (2016). "A comparison of radiocarbon ages derived from bulk peat and selected plant macrofossils in basal peat cores from circum-arctic peatlands." *Quaternary Geochronology*, **31**, 53–61. doi:10.1016/j.quageo.2015.10.003. URL http://dx.doi.org/10.1016/j.quageo.2015.10.003.
- Hughes P, Barber K (2004). "Contrasting pathways to ombrotrophy in three raised bogs from Ireland and Cumbria, England." *Holocene*, **14**, 65–77. doi:10.1191/0959683604hl690rp.
- Hughes PD (2000). "A reappraisal of the mechanisms leading to ombrotrophy in British raised mires." *Ecology Letters*, 3, 7–9. doi:10.1046/j.1461-0248.2000.00118.x.
- Hølleland H, Skrede J, Holmgaard SB (2017). "Cultural Heritage and Ecosystem Services: A Literature Review." Conservation and Management of Archaeological Sites, 19, 210–237. doi:10.1080/13505033.2017.1342069.
- IMCG (2021). "Global Peatland Database." https://greifswaldmoor.de/global-peatland-database-en. html. Date accessed: 23/03/2021.
- IMCG (2022). "IMCG Poster: Peat is not renewable." http://www.imcg.net/modules/download\_gallery/ dlc.php?file=26&id=1311190352. Date accessed: 07/12/22.
- IPS (2022). "Types of peatlands." https://peatlands.org/peatlands/types-of-peatlands/

 $\#:\sim: text = Geogenous peat lands \% 2 Ci.e. fens \% 2 Carenutrient-rich \% 28 minerotrophic \% 29 and,$ 

Vegetationisdominatedbygrasses%2Csedges%2Candrushes. Date accessed: 16/06/2022.

- "New ISRIC World Soil Museum (2022a). artist-in-residence at World Soil Museum invites peatland conversations." https://www.isric.org/news/ new-artist-residence-world-soil-museum-invites-peatland-conversations. Date accessed: 07/12/2022.
- ISRIC World Soil Museum (2022b). "Peatland Exchange recordings now available from World Soil Museum." https://www.isric.org/news/peatland-exchange-recordings-now-available-world-soil-museum. Date accessed: 12/12/2022.
- IUSS Working Group WRB (2015). World Reference Base for Soil Resources 2014, update 2015: International soil classification system for naming soils and creating legends for soil maps. World Soil Resources Reports No. 106, FAO, Rome. ISBN 9789251083697.
- Ivanov K (1981). Water movement in mirelands. Academic Press, London, New York, translated edition. ISBN 9780123764607.
- Iversen CM, Sloan VL, Sullivan PF, Euskirchen ES, Mcguire AD, Norby RJ, Walker AP, Warren JM, Wullschleger SD (2015). "The unseen iceberg: Plant roots in arctic tundra." *New Phytologist*, 205, 34–58. doi:10.1111/nph.13003.
- Jansen A, Grootjans A (2019). Hoogvenen: landschapsecologie, behoud, beheer, herstel. Noordboek Natuur, Gorredijk. ISBN 9789056155520.
- Jansen A, Grootjans A, Sevink J (2019). "Hoogvenen en heideveentjes in het Nederlandse landschap: Hun vroegere uiterlijk, ontstaan en ligging in het landschap." In A Jansen, A Grootjans (eds.), "Hoogvenen: Landschapsecologie, behoud, beheer, herstel," pp. 124–131. Noordboek Natuur, Gorredijk.
- Jones MC, Yu Z (2010). "Rapid deglacial and early Holocene expansion of peatlands in Alaska." Proceedings of the National Academy of Sciences of the United States of America, 107, 7347–7352. doi:10.1073/pnas.0911387107.
- Jongmans A, den Berg MV, Sonneveld M, Peek G, den Berg van Saparoea RV (2013). "Deel V: Veen." In AG Jongmans, MW Van den Berg, MPW Sonneveld, GJWC Peek and RM Van den Berg van Saparoea (ed.), "Landschappen van Nederland - Geologie, Bodem en Landgebruik," Wageningen Academic Publishers, Wageningen.
- Joosten H, Clarke D (2002). Wise use of mires and peatlands Background and principles including a framework for decision-making. International Mire Conservation Group (IMCG) & International Peatland Society (IPS). ISBN 9519774483.
- Joosten H, Couwenberg J (2001). "Bilanzen zum Moorverlust. Das Beispiel Europa." In M Succow, H Joosten (eds.), "Landschaftsökologische Moorkunde," pp. 406–408. Stuttgart.
- Joosten H, Couwenberg J (2008). "Peatlands and Carbon." In "Assessment on Peatlands, Biodiversity and Climate Change: Main Report," pp. 99–117. Global Environment Centre, Kuala Lumpur and Wetlands International, Wageningen.
- Joosten H, Grootjans A, Schouten M, Jansen A (2017a). "Netherlands." In H Joosten, F Tanneberger, A Moen (eds.), "Mires and peatlands in Europe: Status, distribution and conservation," pp. 523–535. Schweizerbart Science Publishers, Stuttgart. ISBN 9783510653836. doi:10.1127/mireseurope/2017/0001-0044.
- Joosten H, Tanneberger F, Moen A (2017b). Mires and peatlands in Europe: Status, distribution and conservation. Schweizerbart Science Publishers, Stuttgart. ISBN 9783510653836. doi:10.1127/mireseurope/2017/0001-0044.
- Jull A, Burr G (2015). "Accelerator Mass Spectrometry." In "Encyclopedia of Scientific Dating Methods," pp. 3–6. ISBN 9789400763036. doi:10.1007/978-94-007-6304-3. URL http://link.springer.com/10. 1007/978-94-007-6326-5%0Ahttp://link.springer.com/10.1007/978-94-007-6304-3.
- Kennedy DM, Woods JL (2013). Determining Organic and Carbonate Content in Sediments, volume 14. Elsevier Ltd. ISBN 9780080885223. doi:10.1016/B978-0-12-374739-6.00389-4. URL http://dx.doi. org/10.1016/B978-0-12-374739-6.00389-4.
- Kilian M, Van der Plicht J, Van Geel B (1995). "Dating raised bogs: new aspects of AMS 14C wiggle matching, a reservoir effect and climatic change." Quaternary Science Reviews, 14, 959–966.
- Kimmel K, Mander U (2010). "Ecosystem services of peatlands: Implications for restoration." Progress in Physical Geography, 34, 491–514. doi:10.1177/0309133310365595.
- Klaver E (1981). Een Holocene vegetatie successie in het Fochtelooerveen. Master's thesis, Hugo de Vries Laboratorium, Universiteit van Amsterdam, Amsterdam.
- Kleinen T, Brovkin V, Schuldt RJ (2012). "A dynamic model of wetland extent and peat accumulation:

Results for the Holocene." Biogeosciences, 9, 235–248. doi:10.5194/bg-9-235-2012.

- Klijnstra B (1979). "1.2.1 Geologisch-geografische beschrijving." In M Ter Wee (ed.), "Emmen West (17W), Emmen Oost (17O). Toelichtingen bij de geologische kaart van Nederland 1:50.000," pp. 10–16. Rijks Geologische Dienst, Haarlem.
- KNMI (2021). "Klimaattabel Station Eelde, periode 1991-2020." https://www.knmi.nl/klimaat-viewer.
- Kohzu A, Matsui K, Yamada T, Sugimoto A, Fujita N (2003). "Significance of rooting depth in mire plants: Evidence from natural 15N abundance." *Ecological Research*, 18, 257–266. doi:10.1046/j.1440-1703.2003.00552.x.
- Körber-Grohne U (1964). "Bestimmungsschlüssel für subfossile Juncus-Samen und Gramineen-Früchte." Probleme der Küstenforschung im südlichen Nordseegebiet, 7.
- Körber-Grohne U (1991). "Bestimmungsschlüssel für subfossile Gramineen-Früchte." Probleme der Küstenforschung im südlichen Nordseegebiet, **18**, 169–234.
- Korhola A (1994). "Radiocarbon evidence for rates of lateral expansion in raised mires in southern Finland." *Quaternary Research*, **42**, 299–307.
- Korhola A (1996). "Initiation of a sloping mire complex in southwestern Finland: Autogenic versus allogenic controls." *Ecoscience*, **3**, 216–222.
- Korhola A, Ruppel M, Seppä H, Väliranta M, Virtanen T, Weckström J (2010). "The importance of northern peatland expansion to the Late-Holocene rise of atmospheric methane." *Quaternary Science Reviews*, **29**, 611–617. doi:10.1016/j.quascirev.2009.12.010.
- Koster E (1988). "Ancient and modern cold-climate aeolian sand deposition: a review." *Journal of Quaternary Science*, **3**, 69–83.
- Koster E (2005). "Aeolian Environments." In E Koster (ed.), "Physical Geography of Western Europe," pp. 139–160. Oxford University Press, Oxford.
- Koster K (2017). 3D characterization of Holocene peat in the Netherlands: Implications for coastal-deltaic subsidence. Ph.D. thesis, Utrecht University, Utrecht.
- Koudijs R (2022). Constraining peat initiation by OSL-dating the pre-peat landscape. Master's thesis, Soil Geography and Landscape Group, Wageningen University & Research, Wageningen.
- Kuijer P, Rosing H (1994). Zwolle Oost (210). Toelichtingen bij de bodemkaart van Nederland schaal 1:50.000. Staring Centrum, Wageningen.
- Lanting JN, Van der Plicht J (1994). "14C-AMS: Pros and cons for archaeology." *Palaeohistoria*, **35/36**, 1–12. ISSN 0552-9344.
- Libby W, Anderson E, Arnold J (1949). "Age determination by radiocarbon content: World-wide assay of natural radiocarbon." *Science*, **109**, 227–228.
- Limpens J (2003). Prospects for Sphagnum bogs subject to high nitrogen deposition. Ph.D. thesis, Wageningen University, Wageningen.
- Loisel J, Bunsen M (2020). "Abrupt fen-bog transition across southern Patagonia: Timing, causes, and impacts on carbon sequestration." Frontiers in Ecology and Evolution, 8, 1–19. doi:10.3389/fevo.2020.00273.
- Loisel J, Yu Z, Parsekian A, Nolan J, Slater L (2013). "Quantifying landscape morphology influence on peatland lateral expansion using ground-penetrating radar (GPR) and peat core analysis." *Journal of Geophysical Research: Biogeosciences*, **118**, 373–384. doi:10.1002/jgrg.20029.
- Lowe J, Walker M (2015). "Definition of the Last Termination." In J Lowe, M Walker (eds.), "Reconstructing Quaternary Environments," pp. 413–415. Routledge, London & New York, third edition.
- Macdonald GM, Beilman DW, Kremenetski KV, Sheng Y, Smith LC, Velichko AA (2006). "Rapid early development of circumarctic peatlands and atmospheric CH4 and CO2 variations." *Science*, **314**, 285–288. doi:10.1126/science.1131722.
- Martin R (1999). "Introduction: the science of taphonomy." In R Martin (ed.), "Taphonomy: A Process Approach," pp. 1–26. Cambridge Paleobiology Series 4, Cambridge University Press, Cambridge. ISBN 0521598338.
- Meijles EW, Kiden P, Streurman HJ, van der Plicht J, Vos PC, Gehrels WR, Kopp RE (2018). "Holocene relative mean sea-level changes in the Wadden Sea area, northern Netherlands." *Journal of Quaternary Science*, **33**, 905–923. doi:10.1002/jgs.3068.
- Meleisea E (2022). "Why training peatlands researchers is fundamental to good decision-making for the climate and nature." https://app-gpi.tbf.dev/resource-library/ why-training-peatlands-researchers-fundamental-good-decision-making-climate-and. Date accessed: 07/12/22.

- Millard AR (2014). "Conventions for reporting radiocarbon determinations." *Radiocarbon*, **56**, 555–559. doi:10.2458/56.17455.
- Minayeva T, Bragg O, Cherednichenko O, Duinen GV, Giesen W, Grootjans A, Grundling P, Nikolaev V, der Schaaf SV (2008). "Peatlands and Biodiversity." In F Parish, A Sirin, D Charman, H Joosten, T Minayeva, M Silvius, L Stringer (eds.), "Assessment on Peatlands, Biodiversity and Climate Change: Main Report," pp. 60–98. Global Environment Centre, Kuala Lumpur and Wetlands International, Wageningen.
- Ministerie van Economische Zaken Directie Natuur & Biodiversiteit (2018). "Natura 2000-gebieden peildatum 27 augustus 2018 [Data File]."
- Ministerie van Verkeer en Waterstaat (2007). "Landelijke CONCEPT dataset lijnvormige waterlichamen status mrt 2007 [Data file]."
- Mitsch W, Gosselink J, Anderson C, Zhang L (2009). "Peatlands." In "Wetland Ecosystems," pp. 149–187. John Wiley & Sons, Ltd. ISBN 9780470286302.
- Moen A (1985). "Classification of mires for conservation purposes in Norway." *Aquilo Seriales Botanica*, **21**, 95–100.
- Mook W (2006). Introduction to Isotope Hydrology. CRC Press, Taylor and Francis Group, London. ISBN 9780415398053.
- Mook W, Streurman H (1983). "Physical and chemical aspects of radiocarbon dating." *PACT Publication*, **8**, 31–55.
- Moore P (1993). "The origin of blanket mire, revisited." In F Chambers (ed.), "Climate change and human impact on the landscape," pp. 217–224. Chapman and Hal, London.
- Moore PD (1975). "Origin of blanket mires." *Nature*, **256**, 267–269. ISSN 00280836. doi:10.1038/256267a0.
- Morris P, Baird A, Belyea L (2012). "The DigiBog peatland development model 2: ecohydrological simulations in 2D." *Ecohydrology*, 5, 256–268. doi:10.1002/eco.229.
- Morris PJ, Swindles GT, Valdes PJ, Ivanovic RF, Gregoire LJ, Smith MW, Tarasov L, Haywood AM, Bacon KL (2018). "Global peatland initiation driven by regionally asynchronous warming." Proceedings of the National Academy of Sciences of the United States of America, 115, 4851–4856. doi:10.1073/pnas.1717838115.
- Mäkilä M (1997). "Holocene lateral expansion, peat growth and carbon accumulation on Haukkasuo, a raised bog in southeastern Finland." *Boreas*, **26**, 1–14. doi:10.1111/j.1502-3885.1997.tb00647.x.
- Mäkilä M, Moisanen M (2007). "Holocene lateral expansion and carbon accumulation of Luovuoma, a northern fen in finnish Lapland." *Boreas*, **36**, 198–210. doi:10.1111/j.1502-3885.2007.tb01192.x.
- Nelson K, Thompson D, Hopkinson C, Petrone R, Chasmer L (2021). "Peatland-fire interactions: A review of wildland fire feedbacks and interactions in Canadian boreal peatlands." *Science of the Total Environment*, **769**, 1–14. doi:10.1016/j.scitotenv.2021.145212.
- Nichols JE, Peteet DM (2019). "Rapid expansion of northern peatlands and doubled estimate of carbon storage." *Nature Geoscience*, **12**, 917–921. doi:10.1038/s41561-019-0454-z.
- Nilsson M, Klarqvist M, Bohlin E, Possnert G (2001). "Variation in 14C age of macrofossils and different fractions of minute peat samples dated by AMS." *Holocene*, **11**, 579–586. ISSN 09596836. doi:10.1191/095968301680223521.
- Oliver M, Webster R (2014). "A tutorial guide to geostatistics: Computing and modelling variograms and kriging." *Catena*, **113**, 56–69. doi:10.1016/j.catena.2013.09.006.
- Olsson IU (2009). "Radiocarbon dating history: Early days, questions, and problems met." *Radiocarbon*, **51**, 1–43.
- Palstra SWL, Wallinga J, Viveen W, Schoorl JM, Van den Berg M, Van der Plicht J (2021). "Crosscomparison of last glacial radiocarbon and OSL ages using periglacial fan deposits." *Quaternary Geochronology*, **61**. doi:10.1016/j.quageo.2020.101128.
- Pantzer E (1986). Landschaftsentwicklung und Besiedlungsgeschichte im Stader Raum. Ein interdisciplinäres Forschungsproject. Die Untersuchungen der Jahre 1983-1984 in Hammah und Groß Sterneberg. Hansa-Druckerei Stelzer GmbH + Co, Stade.
- Paris F, Cleveringa P, Gans WD (1979). "The Stokersdobbe: Geology and palynology of a deep pingo remnant in Friesland (The Netherlands)." *Geologie en Mijnbouw*, **58**, 33–38.
- Parish F, Silvius M, Lin CK, Joosten H, Reed M, Suryadiputra N, Stringer L (2008a). "Management of Peatlands for Biodiversity and Climate Change." In F Parish, A Sirin, D Charman, H Joosten, T Minayeva, M Silvius, L Stringer (eds.), "Assessment on Peatlands, Biodiversity and Climate Change:

Main Report," pp. 155–179. Global Environment Centre, Kuala Lumpur and Wetlands International, Wageningen.

- Parish F, Sirin A, Charman D, Joosten H, Minayeva T, Silvius M, Stringer L (2008b). Assessment on Peatlands, Biodiversity and Climate Change: Main Report. Global Environment Centre, Kuala Lumpur and Wetlands International, Wageningen. ISBN 9789834375102.
- Paulissen M, Van Beek R (In prep.). "A legacy with potential: mapping cultural remains in present and former raised bogs and considerations for management."
- Peregon A, Uchida M, Yamagata Y (2009). "Lateral extension in Sphagnum mires along the southern margin of the boreal region, Western Siberia." *Environmental Research Letters*, 4. doi:10.1088/1748-9326/4/4/045028.
- Philippsen B (2013). "The freshwater reservoir effect in radiocarbon dating." Heritage Science, 1, 1–19. doi:10.1186/2050-7445-1-24.
- Pierik HJ, Cohen KM (2020). "The use of geological, geomorphological and soil mapping products in palaeolandscape reconstructions for the Netherlands." *Netherlands Journal of Geosciences*, **99**, 1–20. doi:https://doi.org/10.1017/njg.2020.8.
- Pierik HJ, Cohen KM, Vos PC, Van der Spek AJF, Stouthamer E (2017). "Late Holocene coastal-plain evolution of the Netherlands: the role of natural preconditions in human-induced sea ingressions." *Proceedings of the Geologists' Association*, **128**, 180–197. doi:10.1016/j.pgeola.2016.12.002.
- Piotrowska N, Blaauw M, Mauquoy D, Chambers F (2011). "Constructing deposition chronologies for peat deposits using radiocarbon dating." *Mires and Peat*, 7, 1–14.
- Plunkett G, Whitehouse NJ, Hall VA, Charman DJ, Blaauw M, Kelly E, Mulhall I (2009). "A multi-proxy palaeoenvironmental investigation of the findspot of an Iron Age bog body from Oldcroghan, Co. Offaly, Ireland." *Journal of Archaeological Science*, **36**, 265–277. doi:10.1016/j.jas.2008.09.014.
- Pons L (1992). "Holocene peat formation in the lower parts of the Netherlands." In J Verhoeven (ed.), "Fens and Bogs in the Netherlands: Vegetation, History, Nutrient Dynamics and Conservation," pp. 7–79. Kluwer Academic Publishers, Dordrecht. ISBN 9789401579971. doi:10.1007/978-94-015-7997-1.
- Popping H (1935). Praehistorie in een veengebied.
- Preusser F, Degering D, Fuchs M, Hilgers A, Kadereit A, Klasen N, Krbetschek M, Richter D, Spencer JQG (2008). "Luminescence dating: basics, methods and applications." *E&G Quaternary Science Journal*, 57, 95–149. doi:https://doi.org/10.3285/eg.57.1-2.5.
- Provincie Drenthe (2016). "Beheerplan Fochteloërveen Op weg naar een levend hoogveen." Technical report, Provincie Drenthe, Assen.
- Provincie Drenthe (2022). "Dikte van het keileempakket in Drenthe, in meters." https:// kaartportaal.drenthe.nl/portal/home/item.html?id=109e34c7141147178fbd523060170c42. Date accessed: 16/06/22.
- Prummel W, Niekus MJ (2011). "Late Mesolithic hunting of a small female aurochs in the valley of the River Tjonger (the Netherlands) in the light of Mesolithic aurochs hunting in NW Europe." *Journal* of Archaeological Science, **38**, 1456–1467. doi:10.1016/j.jas.2011.02.009.
- Pynacker C (1664). "Historical map: Drenthe Drentia Comitatus Transiselaniae Tabula II (Cornelio Pynacker, 1664)." https://commons.wikimedia.org/wiki/File:Drenthe\_-\_Drentia\_Comitatus\_-\_ Transiselaniae\_Tabula\_II\_(Cornelio\_Pynacker, 1664).jpg. Date accessed: 24/07/22.
- Quik C, Palstra SW, Van Beek R, Van der Velde Y, Candel JH, Van der Linden M, Kubiak-Martens L, Swindles GT, Makaske B, Wallinga J (2022a). "Dating basal peat: The geochronology of peat initiation revisited." *Quaternary Geochronology*, **72**, 1–22. doi:10.1016/j.quageo.2022.101278.
- Quik C, Van der Velde Y, Candel J, Steinbuch L, Van Beek R, Wallinga J (2022b). "Data from: Faded landscape: unravelling peatland initiation and lateral expansion at one of NW-Europe's largest bog remnants." doi:10.4121/20237958. 4TU.Centre for Research Data.
- Quik C, Van der Velde Y, Harkema T, Van der Plicht H, Quik J, Van Beek R, Wallinga J (2021). "Using legacy data to reconstruct the past? Rescue, rigor and reuse in peatland geochronology." *Earth Surface Processes and Landforms*, **46**, 2607–2631. doi:10.1002/esp.5196.
- Raemaekers D (1999). The articulation of a 'New Neolithic'. The meaning of the Swifterbant Culture for the process of neolithisation in the western part of the North European Plain (4900-3400 BC). Ph.D. thesis, Leiden University, Leiden.
- Ramsey CB (1995). "Radiocarbon calibration and analysis of stratigraphy: The OxCal program." Radiocarbon, 37, 425–430. doi:10.1017/s0033822200030903.
- Ramsey CB (2008a). "Deposition models for chronological records." Quaternary Science Reviews, 27,

42-60. doi:10.1016/j.quascirev.2007.01.019.

- Ramsey CB (2008b). "Radiocarbon dating: Revolutions in understanding." Archaeometry, 50, 249–275. doi:10.1111/j.1475-4754.2008.00394.x.
- Ramsey CB (2009). "Dealing with outliers and offsets in radiocarbon dating." Radiocarbon, 51, 1023– 1045. doi:10.1017/s0033822200034093.
- Ramsey CB (2017). "Methods for summarizing radiocarbon datasets." Radiocarbon, 59, 1809–1833. doi:10.1017/RDC.2017.108.
- Rappol M (1987). "Saalian till in the Netherlands: a review." In J Van der Meer (ed.), "INQUA Symposium on the Genesis and Lithology of glacial deposits Amsterdam, 1986," pp. 3–21. Balkema, Rotterdam, Boston.
- Rappol M, Haldorsen S, Jørgensen P, Van der Meer J, Stoltenberg H (1989). "Composition and origin of petrographically stratified thick till in the Northern Netherlands and a Saalian glaciation model for the North Sea basin." *Mededelingen van de Werkgroep voor Tertiair en Kwartair Geologie*, 26, 31–64.
- Rasmussen SO, Andersen KK, Svensson AM, Steffensen JP, Vinther BM, Clausen HB, Siggaard-Andersen ML, Johnsen SJ, Larsen LB, Dahl-Jensen D, Bigler M, Röthlisberger R, Fischer H, Goto-Azuma K, Hansson ME, Ruth U (2006). "A new Greenland ice core chronology for the last glacial termination." *Journal of Geophysical Research*, **111**, 1–16. doi:10.1029/2005JD006079.
- Ratcliffe JL, Peng H, Nijp JJ, Nilsson MB (2021). "Lateral expansion of northern peatlands calls into question a 1,055 GtC estimate of carbon storage." *Nature Geoscience*, pp. 3–5. doi:10.1038/s41561-021-00770-9.
- RCE (2019). "Archeologisch Informatiesysteem (Archis)." https://archis.cultureelerfgoed.nl/. Date accessed: 08/11/2019.
- RCE (2022). "Paleogeografische kaarten." https://www.cultureelerfgoed.nl/onderwerpen/ bronnen-en-kaarten/overzicht/paleografische-kaarten. Date accessed: 05/05/22.
- Reimann T, Román-Sánchez A, Vanwalleghem T, Wallinga J (2017). "Getting a grip on soil reworking Single-grain feldspar luminescence as a novel tool to quantify soil reworking rates." *Quaternary Geochronology*, 42, 1–14. doi:10.1016/j.quageo.2017.07.002.
- Reimer P, Austin W, Bard E, Bayliss A, Blackwell P, Ramsey CB, Butzin M, Cheng H, Edwards R, Friedrich M, Grootes P, Guilderson T, Hajdas I, Heaton T, Hogg A, Hughen K, Kromer B, Manning S, Muscheler R, Palmer J, Pearson C, Van der Plicht J, Reimer R, Richards D, Scott E, Southon J, Turney C, Wacker L, Adolphi F, Büntgen U, Capano M, Fahrni S, Fogtmann-Schulz A, Friedrich R, Köhler P, Kudsk S, Miyake F, Olsen J, Reinig F, Sakamoto M, Sookdeo A, Talamo S (2020). "The IntCal20 Northern Hemisphere radiocarbon age calibration curve (0–55 cal kBP)." *Radiocarbon*, 62, 725–757. doi:10.1017/RDC.2020.41.
- Rein G, Huang X (2021). "Smouldering Wildfires in Peatlands, Forests and the Arctic: Challenges and Perspectives." Current Opinion in Environmental Science & Health. doi:10.1016/j.coesh.2021.100296.
- Renssen H, Seppä H, Heiri O, Roche DM, Goosse H, Fichefet T (2009). "The spatial and temporal complexity of the Holocene thermal maximum." *Nature Geoscience*, 2, 411–414. doi:10.1038/ngeo513.
- Roberts N (2014). "The environmental future: A Holocene perspective." In N Roberts (ed.), "The Holocene An Environmental History," pp. 336–351. Wiley-Blackwell, Chichester, third edition.
- Robichaud A, Bégin Y (2009). "Development of a raised bog over 9000 years in Atlantic Canada." *Mires and Peat*, **5**, 1–19.
- Roe HM, Charman DJ, Gehrels WR (2002). "Fossil testate amoebae in coastal deposits in the UK: Implications for studies of sea-level change." Journal of Quaternary Science, 17, 411–429. doi:10.1002/jqs.704.
- "Veenboring ROEG! (2021).brengt eeuwenoude geschiedenis aan de oppervlakte in het Fochteloërveen." https://www.roeg.tv/roeg-nieuws/ veenboring-brengt-eeuwenoude-geschiedenis-aan-de-oppervlakte-in-het-fochteloerveen/. Television broadcast. Date accessed: 07/12/22.
- Rotherham I (2020). "Peatland ecosystem services." In I Rotherham (ed.), "Peatlands Ecology, conservation and heritage," pp. 61–68. Routledge (Taylor & Francis Group), Abingdon & New York. ISBN 9781138343207.
- Ruppel M, Väliranta M, Virtanen T, Korhola A (2013). "Postglacial spatiotemporal peatland initiation and lateral expansion dynamics in North America and northern Europe." *Holocene*, 23, 1596–1606. doi:10.1177/0959683613499053.
- Rydin H, Jeglum J (2013a). The Biology of Peatlands. Oxford University Press, Oxford, second edition. ISBN 9780199603008.

- Rydin H, Jeglum J (2013b). "The Peat Archives." In H Rydin, J Jeglum (eds.), "The Biology of Peatlands," pp. 109–126. Oxford University Press, Oxford, second edition.
- Rydin H, Jeglum J (2013c). "Peatland habitats." In H Rydin, J Jeglum (eds.), "The Biology of Peatlands," pp. 1–20. Oxford University Press, Oxford, second edition.
- Rydin H, Jeglum J (2013d). "Peatland succession and development." In H Rydin, J Jeglum (eds.), "The Biology of Peatlands," pp. 127–147. Oxford University Press, Oxford, second edition.
- Rydin H, Jeglum J (2013e). "Productivity and peat accumulation." In H Rydin, J Jeglum (eds.), "The Biology of Peatlands," pp. 254–273. Oxford University Press, Oxford, second edition.
- Savage C, Vickers A (2009). "Empirical Study of Data Sharing by Authors Publishing in PLoS Journals." PLoS ONE, 4. doi:10.1371/journal.pone.0007078.
- Sevink J (2019). "Bodemvorming en hoogveen." In A Jansen, A Grootjans (eds.), "Hoogvenen: Landschapsecologie, behoud, beheer, herstel," pp. 48–53. Noordboek Natuur, Gorredijk.
- Shore JS, Bartley DD, Harkness DD (1995). "Problems encountered with the 14C dating of peat." Quaternary Science Reviews, 14, 373–383. doi:10.1016/0277-3791(95)00031-3.
- Sirin A, Laine J (2008). "Peatlands and greenhouse gases." In F Parish, A Sirin, D Charman, H Joosten, T Minayeva, M Silvius, L Stringer (eds.), "Assessment on Peatlands, Biodiversity and Climate Change: Main Report," pp. 118–138. Global Environment Centre, Kuala Lumpur and Wetlands International, Wageningen.
- Small D, Clark CD, Chiverrell RC, Smedley RK, Bateman MD, Duller GA, Ely JC, Fabel D, Medialdea A, Moreton SG (2017). "Devising quality assurance procedures for assessment of legacy geochronological data relating to deglaciation of the last British-Irish Ice Sheet." *Earth-Science Reviews*, **164**, 232–250. doi:10.1016/j.earscirev.2016.11.007.
- Smith A (2004). The Moss Flora of Britain and Ireland. Cambridge University Press, Cambridge, second edition.
- Smith MJ, Keesstra S, Rose J (2015). "Use of legacy data in geomorphological research." *GeoResJ*, **6**, 74–80. doi:10.1016/j.grj.2015.02.008.
- Smolders AJ, Tomassen HB, Pijnappel HW, Lamers LP, Roelofs JG (2001). "Substrate-derived CO2 is important in the development of Sphagnum spp." New Phytologist, 152, 325–332. doi:10.1046/j.0028-646X.2001.00261.x.
- Soortenbanknl (2020). "Soortenbank.nl." https://web.archive.org/web/20220420040928/http:// soortenbank.nl/.
- Spek (2004). Het Drentse Esdorpenlandschap Een historisch-geografische studie. Uitgeverij Matrijs, Utrecht, in samenwerking met Stichting Het Drentse Landschap. ISBN 9789053452547.
- Staring W (1983). De wording van kienhout. Dekker & Huisman, Wildervank. ISBN 9789070377199.
- Story Driven (2022). "Et Fochtelervene Van lestied Naor Wieshied (een documentaire over het Fochteloërveen)." https://www.youtube.com/watch?v=TXekDsSrYFc. Television broadcast. Date accessed: 07/12/22.
- Streif H (1972). "The results of stratigraphical and facial investigations in the coastal Holocene of Woltzeten/Ostfriesland, Germany." Geologiska Föreningen i Stockholm Förhandlingar, 94, 281–299. doi:10.1080/11035897209454203.
- Stuiver M, Polach H (1977). "Discussion: Reporting of 14C Data." Radiocarbon, 19, 355-363.
- Sullivan ME, Booth RK (2011). "The potential influence of short-term environmental variability on the composition of testate amoeba communities in Sphagnum peatlands." *Microbial Ecology*, **62**, 80–93.
- Swindles GT, Morris PJ, Mullan DJ, Payne RJ, Roland TP, Amesbury MJ, Lamentowicz M, Turner TE, Gallego-Sala A, Sim T, Barr ID, Blaauw M, Blundell A, Chambers FM, Charman DJ, Feurdean A, Galloway JM, Gałka M, Green SM, Kajukało K, Karofeld E, Korhola A, Łukasz Lamentowicz, Langdon P, Marcisz K, Mauquoy D, Mazei YA, McKeown MM, Mitchell EA, Novenko E, Plunkett G, Roe HM, Schoning K, Ülle Sillasoo, Tsyganov AN, Van der Linden M, Väliranta M, Warner B (2019).
  "Widespread drying of European peatlands in recent centuries." Nature Geoscience, 12, 922–928. doi:10.1038/s41561-019-0462-z.
- Swindles GT, Roe HM (2007). "Examining the dissolution characteristics of testate amoebae (Protozoa: Rhizopoda) in low pH conditions: Implications for peatland palaeoclimate studies." *Palaeogeography, Palaeoclimatology, Palaeoecology*, **252**, 486–496. doi:10.1016/j.palaeo.2007.05.004.
- Swindles GT, Roland TP, Amesbury MJ, Lamentowicz M, McKeown MM, Sim TG, Fewster RE, Mitchell EA (2020). "Quantifying the effect of testate amoeba decomposition on peat-based water-table reconstructions." *European Journal of Protistology*, **74**, 125693. doi:10.1016/j.ejop.2020.125693.

- Swinnen W, Broothaerts N, Verstraeten G (2021). "Modelling long-term alluvial-peatland dynamics in temperate river floodplains." *Biogeosciences*, **18**, 6181–6212. doi:10.5194/bg-18-6181-2021.
- Synal H, Stocker M, Suter M (2007). "MICADAS: a new compact radiocarbon AMS system." Nuclear Instruments and Methods, B259, 7–13.
- Tanneberger F, Tegetmeyer C, Busse S, Barthelmes A, Shumka S, Mariné AM, Jenderedjian K, Steiner GM, Essl F, Etzold J, Mendes C, Kozulin A, Frankard P, Milanović, Ganeva A, Apostolova I, Alegro A, Delipetrou P, Navrátilová J, Risager M, Leivits A, Fosaa AM, Tuominen S, Muller F, Bakuradze T, Sommer M, Christanis K, Szurdoki E, Oskarsson H, Brink SH, Connolly J, Bragazza L, Martinelli G, Aleksāns O, Priede A, Sungaila D, Melovski L, Belous T, Saveljić D, Vries FD, Moen A, Dembek W, Mateus J, Hanganu J, Sirin A, Markina A, Napreenko M, Lazarević P, Šefferová Stanová V, Skoberne P, Pérez PH, Pontevedra-Pombal X, Lonnstad J, Küchler M, Wüst-Galley C, Kirca S, Mykytiuk O, Lindsay R, Joosten H (2017). "The peatland map of Europe." *Mires and Peat*, 19, 1–17. doi:10.19189/MaP.2016.OMB.264.
- Taylor RE (2000). "Fifty years of radiocarbon dating This widely applied technique has made major strides since its introduction a half-century ago at the University of Chicago." American Scientist, 88, 60–67. doi:10.1511/2000.15.765.
- Tengberg A, Fredholm S, Eliasson I, Knez I, Saltzman K, Wetterberg O (2012). "Cultural ecosystem services provided by landscapes: Assessment of heritage values and identity." *Ecosystem Services*, 2, 14–26. doi:10.1016/j.ecoser.2012.07.006.
- Ter Wee M (1962). "The Saalian glaciation in the Netherlands." Mededelingen van de Geologische Stichting, Nieuwe Serie, 15, 57–76.
- Ter Wee M (1966). Steenwijk Oost (160). Toelichtingen bij de geologische kaart van Nederland 1:50.000. Geologische Stichting, Afdeling Geologische Dienst, Haarlem.
- Ter Wee M (1972). "Geologische opbouw van Drenthe." Technical Report nr. 877, Rijks Geologische Dienst, Haarlem.
- Ter Wee M (1979). Emmen West (17W), Emmen Oost (17O). Toelichtingen bij de geologische kaart van Nederland 1:50.000. Rijks Geologische Dienst, Haarlem.
- TNO Geological Survey of the Netherlands (2021a). "Basisveen Bed." https://www.dinoloket.nl/en/ stratigraphic-nomenclature/basisveen-bed. Date accessed: 16/03/2021.
- TNO Geological Survey of the Netherlands (2021b). "Gieten Member. In: Stratigraphic Nomenclature of the Netherlands." http://www.dinoloket.nl/en/stratigraphic-nomenclature/gieten-member. Date accessed: 01/04/2021.
- TNO Geological Survey of the Netherlands (2021c). "Laagpakket van Wierden. In: Stratigrafische Nomenclator van Nederland." https://www.dinoloket.nl/en/stratigraphic-nomenclature/ wierden-member. Date accessed: 16/03/2021.
- Tolonen K, Turunen J (1996). "Accumulation rates of carbon in mires in Finland and implications for climate change." Holocene, 6, 171–178. doi:10.1177/095968369600600204.
- TOPCON (2017). "Hiper V Versatile Function GNSS Receiver." Technical report, TOPCON Corporation.
- Tuittila ES, Juutinen S, Frolking S, Väliranta M, Laine AM, Miettinen A, Seväkivi ML, Quillet A, Merilä P (2013). "Wetland chronosequence as a model of peatland development: Vegetation succession, peat and carbon accumulation." *Holocene*, 23, 25–35. doi:10.1177/0959683612450197.
- Tuniz C, Bird J, Fink D, Herzog G (1998). Accelerator Mass Spectrometry: ultrasensitive analysis for global science. CRC Press, Boca Raton. ISBN 9780849345388.
- Turunen C, Turunen J (2003). "Development history and carbon accumulation of a slope bog in oceanic British Columbia, Canada." *Holocene*, **13**, 225–238. doi:10.1191/0959683603hl609rp.
- Turunen J, Räty A, Kuznetsov O, Maksimov A, Shevelin P, Grabovik S, Tolonen K, Pitkänen A, Turunen C, Miriläinen J, Jungner H (2002a). "Development history of Patvinsuo Mire, Eastern Finland." pp. 1–72. Nature Protection Publications of the Finnish Forest and Park Service, Series A, No 38.
- Turunen J, Tolonen K, Tomppo E, Reinikainen A (2002b). "Estimating carbon accumulation rates of undrained mires in Finland - Application to boreal and subarctic regions." *Holocene*, **12**, 69–80. doi:10.1191/0959683602hl522rp.
- Törnqvist TE, de Jong AF, Oosterbaan WA, Van der Borg K (1992). "Accurate dating of organic deposits by AMS 14C measurement of macrofossils." *Radiocarbon*, **34**, 566–577. doi:10.1017/S0033822200063840.
- Törnqvist TE, Hijma MP (2012). "Links between early Holocene ice-sheet decay, sea-level rise and abrupt climate change." Nature Geoscience, 5, 601–606. doi:10.1038/ngeo1536.

- Törnqvist TE, Van Ree MH, Van 't Veer R, Van Geel B (1998). "Improving methodology for highresolution reconstruction of sea-level rise and neotectonics by paleoecological analysis and AMS 14C dating of basal peats." *Quaternary Research*, **49**, 72–85. doi:10.1006/qres.1997.1938.
- United Nations Environment Programme (2021). "The Global Peatland Map 2.0." https://wedocs.unep. org/20.500.11822/37571. Date accessed: 31/10/2022.
- Van Aalst J (2021). "OpenSimpleTopo, 3200 pixels per km (scale 1:3,125 as hardcopy), current release: 2021-R02, feb. 2021, map sheets used: 3200-11HNb, 3200-11HNd, 3200-11HZb, 3200-12CNa, 3200-12CNb, 3200-12CNc, 3200-12CNd, 3200-12CZa, 3200-12CZb." www.opentopo.nl. Date accessed: 19/05/2021.
- Van Aalst J (2022). "OpenTopo, 200 pixels per km, current release: 2021-R11, nov. 2021, map sheet used: 09." https://www.opentopo.nl. Date accessed: 04/04/2022.
- Van Asselen S, Cohen KM, Stouthamer E (2017). "The impact of avulsion on groundwater level and peat formation in delta floodbasins during the middle-Holocene transgression in the Rhine-Meuse delta, The Netherlands." *Holocene*, **27**, 1694–1706. doi:10.1177/0959683617702224.
- Van Beek R (2009). Reliëf in tijd en ruimte Interdisciplinair onderzoek naar bewoning en landschap van Oost-Nederland tussen vroege prehistorie en middeleeuwen. PhD thesis. Wageningen University, Wageningen. ISBN 9789085854609.
- Van Beek R (2015). "An interdisciplinary approach to the long-term history of raised bogs: A case study at Vriezenveen (the Netherlands)." Journal of Wetland Archaeology, 15, 1–33. doi:10.1080/14732971.2015.1112591.
- Van Beek R, Candel JHJ, Quik C, Bos JAA, Gouw-Bouman MTIJ, Makaske B, Maas GJ (2019). "The landscape setting of bog bodies: Interdisciplinary research into the site location of Yde Girl, The Netherlands." *The Holocene*, **29**, 1206–1222. doi:10.1177/0959683619838048.
- Van Beek R, Maas GJ, Van den Berg E (2015). "Home Turf: An interdisciplinary exploration of the long-term development, use and reclamation of raised bogs in the Netherlands." *Landscape History*, 36, 5–34. doi:10.1080/01433768.2015.1108024.
- Van Beek R, Quik C, Van der Linden M (In prep.). "Drowning landscapes revisited. Correlating peatland expansion, human habitation trends and vegetation dynamics in the Northwest European mainland."
- Van den Berg M, Beets D (1987). "Saalian glacial deposits and morphology in The Netherlands." In J Van der Meer (ed.), "INQUA Symposium on the Genesis and Lithology of glacial deposits – Amsterdam, 1986," pp. 235–251. Balkema, Rotterdam, Boston.
- Van der Meij WM, Temme A, Lin HS, Gerke HH, Sommer M (2018). "On the role of hydrologic processes in soil and landscape evolution modeling: concepts, complications and partial solutions." *Earth-Science Reviews*, **185**, 1088–1106. doi:10.1016/j.earscirev.2018.09.001.
- Van der Plicht J (1992). "The new Groningen 14C data base." Radiocarbon, 34, 493-499.
- Van der Plicht J, Hogg A (2006). "A note on reporting radiocarbon." *Quaternary Geochronology*, **1**, 237–240. doi:10.1016/j.quageo.2006.07.001.
- Van der Plicht J, Streurman H, Van Mourik J (2019). "Radiocarbon dating of soil archives." In J Van Mourik, J Van der Meer (eds.), "Reading the Soil Archives - Unravelling the geoecological code of palaeosols and sediment cores (Developments in Quaternary Science 18)," pp. 81–113. Elsevier B.V.
- Van der Plicht J, Streurman HJ (2018). "De (pre)historie van 14C-dateringen." In A Nieuwhof, E Knol, J Schokker (eds.), "Fragmenten uit de rijke wereld van de archeologie," pp. 205–210. Vereniging voor Terpenonderzoek, Groningen. ISBN 9789081171496.
- Van der Sanden W (1996). Through Nature to Eternity: The Bog Bodies of Northwest Europe. Batavian Lion International, Amsterdam. ISBN 9789067074186.
- Van der Velde Y, Temme AJ, Nijp JJ, Braakhekke MC, Van Voorn GA, Dekker SC, Dolman AJ, Wallinga J, Devito KJ, Kettridge N, Mendoza CA, Kooistra L, Soons MB, Teuling AJ (2021). "Emerging forest-peatland bistability and resilience of European peatland carbon stores." *Proceedings of the National Academy of Sciences of the United States of America*, **118**, 1–9. doi:10.1073/pnas.2101742118.
- Van Dockum S, Hallewas D, Van Heeringen R, Jungerius E (2001). "The Netherlands." In B Coles, A Olivier (eds.), "The Heritage Management of Wetlands in Europe," pp. 35–45. Europae Archaeologiae Consilium, WARP, Short Run Press, Exeter.
- Van Geel B (1978). "A palaeoecological study of Holocene peat bog sections in Germany and The Netherlands, based on the analysis of pollen, spores and macro- and microscopic remains of fungi, algae, cormophytes and animals." *Review of Palaeobotany and Palynology*, 25, 1–120. doi:10.1016/0034-6667(78)90040-4.

- Van Geel B, Buurman J, Waterbolk HT (1996). "Archaeological and palaeoecological indications of an abrupt climate change in the Netherlands, and evidence for climatological teleconnections around 2650 BP." *Journal of Quaternary Science*, **11**, 451–460. doi:10.1002/(SICI)1099-1417(199611/12)11:6<451::AID-JQS275>3.0.CO;2-9.
- Van Geel B, Van der Plicht J, Kilian M, Klaver E, Kouwenberg J, Renssen H, Reynaud-Farrera I, Waterbolk H (1998). "The sharp rise of delta 14C ca. 800 cal BC: Possible causes, related climatic teleconnections and the impact on human environments." *Radiocarbon*, **40**, 535–550.
- Van Giffen A (1958). "Prähistorische Hausformen auf den Sandböden in den Niederlanden." *Germania*, **36**, 35–71.
- Van Heerd RM, Kuijlaars EAC, Zeeuw MP, Van 't Zand RJ (2000). "Productspecificatie AHN 2000."
- Van Mourik J, Wartenbergh P, Mook W, Streurman H (1995). "Radiocarbon dating of palaeosols in aeolian sands." *Mededelingen Rijks Geologische Dienst*, **52**, 425–439.
- Vleeschouwer FD, Chambers FM, Swindles GT (2010). "Coring and sub-sampling of peatlands for palaeoenvironmental research." *Mires and Peat*, **7**, 1–10.
- Vos P (2015a). "Compilation of the Holocene palaeogeographical maps of the Netherlands." In "Origin of the Dutch coastal landscape: Long-term landscape evolution of the Netherlands during the Holocene, described and visualized in national, regional and local palaeogeographical map series.", pp. 50–79. Barkhuis, Groningen (also published as the author's PhD thesis, Utrecht University, 2015). ISBN 9789491431821.
- Vos P (2015b). Origin of the Dutch coastal landscape: Long-term landscape evolution of the Netherlands during the Holocene, described and visualized in national, regional and local palaeogeographical map series. Barkhuis, Groningen (also published as the author's PhD thesis, Utrecht University, 2015). ISBN 9789491431821.
- Vos P, Van der Meulen M, Weerts H, Bazelmans J (2020). Atlas of the Holocene Netherlands, landscape and habitation since the last ice age. Amsterdam University Press, Amsterdam. ISBN 9789463724432.
- Vos P, Vries SD (2013). "2e generatie palaeogeografische kaarten van Nederland (versie 2.0)." www. cultureelerfgoed.nl. Date accessed: 22/01/2020.
- Väliranta M, Salojärvi N, Vuorsalo A, Juutinen S, Korhola A, Luoto M, Tuittila ES (2017). "Holocene fen-bog transitions, current status in Finland and future perspectives." *Holocene*, 27, 752–764. doi:10.1177/0959683616670471.
- Waddington JM, Roulet NT (1997). "Groundwater flow and dissolved carbon movement in a boreal peatland." *Journal of Hydrology*, **191**, 122–138. doi:10.1016/S0022-1694(96)03075-2.
- Walker M (2005). "Radiometric Dating 1: Radiocarbon Dating." In "Quaternary Dating Methods," pp. 17–56. John Wiley & Sons, Ltd, Chichester. ISBN 9780470869260.
- Walker M, Gibbard P, Head MJ, Berkelhammer M, Björck S, Cheng H, Cwynar LC, Fisher D, Gkinis V, Long A, Lowe J, Newnham R, Rasmussen SO, Weiss H (2019). "Formal subdivision of the Holocene Series/Epoch: A summary." *Journal of the Geological Society of India*, **93**, 135–141. doi:10.1007/s12594-019-1141-9.
- Walker M, Johnsen S, Rasmussen SO, Popp T, Steffensen JP, Gibbard P, Hoek W, Lowe J, Andrews J, Björck S, c Cwynar L, Hughen K, Kershaw P, Kromer B, Litt T, Lowe DJ, Nakagawa T, Newnham R, Schwander J (2009). "Formal definition and dating of the GSSP (Global Stratotype Section and Point) for the base of the Holocene using the Greenland NGRIP ice core, and selected auxiliary records." *Journal of Quaternary Science*, 24, 3–17. doi:10.1002/jqs.1227.
- Waller MP, Long AJ, Schofield JE (2006). "Interpretation of radiocarbon dates from the upper surface of late-Holocene peat layers in coastal lowlands." *Holocene*, **16**, 51–61. doi:10.1191/0959683606hl895ra.
- Wallinga J, Van der Staay J (1999). "Sampling in waterlogged sands with a simple hand-operated corer." Ancient TL, 17, 59–61.
- Wanner H, Beer J, Bütikofer J, Crowley TJ, Cubasch U, Flückiger J, Goosse H, Grosjean M, Joos F, Kaplan JO, Küttel M, Müller SA, Prentice IC, Solomina O, Stocker TF, Tarasov P, Wagner M, Widmann M (2008). "Mid- to Late Holocene climate change: an overview." *Quaternary Science Reviews*, 27, 1791–1828. doi:10.1016/j.quascirev.2008.06.013.
- Waterbolk H (2007). "Zwervend tussen de venen. Een poging tot reconstructie van het woongebied van de hunebedbouwers op het centrale deel van het Fries-Drents plateau." In J Bloemers (ed.), "Tussen D26 en P14: Jan Albert Bakker 65 jaar," pp. 181–208. Amsterdams Archeologisch Centrum, Universiteit van Amsterdam, Amsterdam.
- Weckström J, Seppä H, Korhola A (2010). "Climatic influence on peatland formation and lateral expansion

in sub-arctic Fennoscandia." Boreas, 39, 761–769. doi:10.1111/j.1502-3885.2010.00168.x.

- Westerhoff W, Geluk M, Mulder FD (2003). "Deel 2. Geschiedenis van de ondergrond; Hoofdstuk 4.5 Holoceen [Part 2. History of the subsoil; Chapter 4.5 Holocene]." In FD Mulder, M Geluk, I Ritsema, W Westerhoff, T Wong (eds.), "De Ondergrond van Nederland," pp. 211 – 239. Wolters-Noordhoff BV, Groningen, Houten. ISBN 9001605141.
- Wieder R, Vitt D (2006). Boreal Peatland Ecosystems. Springer, Berlin, Heidelberg. ISBN 9783540319122. doi:10.1007/978-3-540-31913-9.
- Wilkinson MD, Dumontier M, Aalbersberg IJ, Appleton G, Axton M, Baak A, Blomberg N, Boiten JW, Santos LBDS, Bourne PE, Bouwman J, Brookes AJ, Clark T, Crosas M, Dillo I, Dumon O, Edmunds S, Evelo CT, Finkers R, Gonzalez-Beltran A, Gray AJ, Groth P, Goble C, Grethe JS, Heringa J, t Hoen PA, Hooft R, Kuhn T, Kok R, Kok J, Lusher SJ, Martone ME, Mons A, Packer AL, Persson B, Rocca-Serra P, Roos M, Van Schaik R, Sansone SA, Schultes E, Sengstag T, Slater T, Strawn G, Swertz MA, Thompson M, Van der Lei J, Van Mulligen E, Velterop J, Waagmeester A, Wittenburg P, Wolstencroft K, Zhao J, Mons B (2016). "Comment: The FAIR Guiding Principles for scientific data management and stewardship." *Scientific Data*, 3, 1–9. doi:10.1038/sdata.2016.18.
- Wohlfarth B, Skog G, Possnert G, Holmquist B (1998). "Pitfalls in the AMS radiocarbon-dating of terrestrial macrofossils." *Journal of Quaternary Science*, **13**, 137–145. doi:10.1002/(SICI)1099-1417(199803/04)13:2<137::AID-JQS352>3.0.CO;2-6.
- Wood R (2015). "From revolution to convention: The past, present and future of radiocarbon dating." Journal of Archaeological Science, 56, 61–72. doi:10.1016/j.jas.2015.02.019.
- Wyborn L, Hsu L, Lehnert K, Parsons MA (2015). "Guest Editorial: Special issue Rescuing Legacy data for Future Science." GeoResJ, 6, 106–107. doi:10.1016/j.grj.2015.02.017.
- Wüst RA, Jacobsen GE, van der Gaast H, Smith AM (2008). "Comparison of radiocarbon ages from different organic fractions in tropical peat cores: Insights from Kalimantan, Indonesia." *Radiocarbon*, 50, 359–372. doi:10.1017/S0033822200053492.
- Yu Z, Beilman DW, Frolking S, MacDonald GM, Roulet NT, Camill P, Charman DJ (2011). "Peatlands and their role in the global carbon cycle." *Eos*, **92**, 1–3. doi:10.1029/2011EO120001.
- Yu Z, Loisel J, Brosseau DP, Beilman DW, Hunt SJ (2010). "Global peatland dynamics since the Last Glacial Maximum." *Geophysical Research Letters*, **37**, 1–5. doi:10.1029/2010GL043584.
- Zaccone C, Rein G, Orazio VD, Hadden RM, Belcher CM, Miano TM (2014). "Smouldering fire signatures in peat and their implications for palaeoenvironmental reconstructions." *Geochimica et Cosmochimica Acta*, 137, 134–146. doi:10.1016/j.gca.2014.04.018.
- Zagwijn W (1986). Nederland in het Holoceen Geologie van Nederland Deel 1. Rijks Geologische Dienst Haarlem, Staatsuitgeverij, 's Gravenhage.
- Zhao Y, Tang Y, Yu Z, Li H, Yang B, Zhao W, Li F, Li Q (2014). "Holocene peatland initiation, lateral expansion, and carbon dynamics in the Zoige Basin of the eastern Tibetan Plateau." *Holocene*, 24, 1137–1145. doi:10.1177/0959683614538077.

## Summary

#### Peatland initiation through time and space

Peatlands comprise 50 – 70% of all global wetlands, making them the most prevalent wetland type. Peat deposits consist of organic material that accumulated under more or less water-saturated conditions, and form a natural archive of past environmental changes. For Europe, it is estimated that over 50% of the peatland area was lost during the twentieth century. For the Netherlands, only 1% of the former peatland area is left today. These tremendous losses in peatland surface area, and ongoing threats to their existence in Europe and elsewhere, pose the urgency to accumulate the knowledge that is contained in the peat archives, and the need to protect the areas that are left. One of the biggest changes recorded in the peat archives, the transition to peat growth itself, is underexposed in scientific research. The transition from dryland to wetland (and to peatland) represents however a huge landscape change, with major impacts for landscape functioning, ecosystem services and human-landscape interactions.

Reconstructing the period, pace and pattern of peat initiation and lateral expansion requires dating the bottom of peat deposits overlying mineral sediment, often called the basal peat. To obtain dating information for reconstructions, one can either look back in the scientific record to build on and integrate existing information, or obtain new data from the field. The huge losses of peat landscapes in Europe call for reuse of existing data, which may contain information that can no longer be obtained from the field. However, for geochronological peat research no overviews exist of factors that need to be taken into consideration for reuse of radiocarbon dates, and standardized workflows or designs for quality assessments of peat dates are lacking. When obtaining new information from the field, accurately dating basal peat is key. However, the lack of a universally applicable and quantitative definition for basal peat, combined with multiple concerns that have been raised previously regarding the radiocarbon dating of peat, may result in apparent ages that are either too old or too young for the timing of peat initiation. Furthermore, in areas where large areas of peatlands are lost, an adapted strategy is required to collect field data based on peat remnants. Alternative methods of analysis are needed to reconstruct peat initiation and lateral expansion in a way that uncertainty is quantified, which is especially relevant when reconstructing a former landscape of which large parts are lost.

In this thesis I focus on mires and peatlands that formed on non-coastal (located above mean sea level) and non-alluvial topographic plains in the coversand landscape of the temperate Northwest European mainland. The spatio-temporal development of these peatlands remains uncertain, probably as a result of their large-scale disappearance following reclamation activities in the past few centuries and consequent limited amount of data for these areas. In addition, existing information on the age of the former peatlands in the coversand landscape of the Northwest European mainland is not yet fully synthesized. I identified two key research deficits, namely (I) methodological developments are needed to constrain the spatio-temporal development of peatlands more accurately; and (II) understanding of the timing, pace and pattern of the initiation and lateral development of peatlands in the coversand landscape of the Northwest European mainland is limited, and responsible steering factors are not well understood. In this thesis I aim to reconstruct peat initiation and lateral expansion in the coversand landscape of the Northwest European mainland, and to develop the required methodological tools, which can be applied irrespective of the case study region.

Each thesis chapter (2 - 4) focuses on elements of research deficits (I) and (II). The period after the Last Glacial Maximum to the present (Late Pleniglacial and Holocene) forms the temporal scope of the thesis. The northern part of the Dutch coversand area was selected as case study region, which contains one of the largest bog remnants of the Northwest European mainland, the Fochteloërveen. This PhD-study is part of the NWO-Vidi project 'Home Turf. An integrated approach to the long-term development, cultural connections and heritage management of Dutch raised bogs', led by Roy van Beek. This project has an interdisciplinary design and consists of several interlinked project elements: landscape archaeology (by Roy van Beek), historical geography (by Maurice Paulissen), ore geology (by Aukjen Nauta), and palaeogeography (this thesis).

In Chapter 2, I developed a workflow for reuse of legacy radiocarbon dates in peatland studies, including a rigorous quality assessment that can be tailored to specific research questions and study regions. In the quality assessment, a penalty is assigned to legacy dates based on criteria that consider taphonomic quality (i.e., sample provenance) and dating quality (i.e., sample material and method used). The weights of the quality criteria may be adjusted based on the research focus, and resulting confidence levels may be used in further analyses to ensure robustness of conclusions. I applied the proposed approach to the northern Dutch coversand landscape, to synthesize existing data for this area and to gain insight in regional peat growth trends. The data search for this area yielded 313 radiocarbon dates from the 1950s to 2019. Based on the quality assessment, the dates -of highly diverse quality- were assigned to four confidence levels. Results indicate a bimodal distribution of peat initiation and lateral expansion, with a first phase of peat growth that peaked at  $\sim 14,000$  cal years BP, followed by a distinct low at approximately  $\sim$ 9,500 cal years BP, and a second phase that peaked at ~4,500 cal years BP. This chapter highlights the potential of legacy data

for palaeogeographic reconstructions, as it is cost-efficient and provides access to information that is (partly) no longer available in the field. However, data retrieval may be challenging, and reuse of data requires that basic information on location, elevation, stratigraphy, sample and laboratory analysis are documented irrespective of the original research aims.

In Chapter 3, I provide a conceptual framework that supports the use of the organic matter (OM) gradient for a quantitative and reproducible definition of the mineral-to-peat transition (i.e., the stratigraphical range reflecting the timespan of the peat initiation process) and the layer defined as basal peat (i.e., the stratigraphical layer that is defined as the bottom of a peat deposit). I analysed the mineral-to-peat transition based on three highly detailed sequences of radiocarbon dates from cores obtained at Fochteloërveen, including dates of plant macrofossils and the humic and humin fractions obtained from bulk samples. Fochteloërveen currently harbours a bog vegetation, but biostratigraphical analvses show that during peat initiation the vegetation was mesotrophic. Results show that plant macrofossils provide the most accurate age in the mineral-to-peat transition and are therefore recommendable to use for <sup>14</sup>C dating basal peat. If these are unattainable, the humic fraction provides the best alternative and is interpreted as a terminus-ante-quem for peat initiation. The potential large age difference between dates of plant macrofossils and humic or humin dates (up to  $\sim$ 1700 years between macrofossil and humic ages, and with even larger differences for humins) suggests that studies reusing existing bulk dates of basal peat should take great care in data interpretation. The potentially long timespan of the peat initiation process (with medians of  $\sim 1000$ ,  $\sim 1300$  and  $\sim 1500$  years at the three studied sites) demonstrates that choices regarding sampling size and resolution need to be well substantiated. I summarise all findings as a set of recommendations for dating basal peats, and advocate the widespread use of OM determination to obtain a low-cost, quantitative and reproducible definition of basal peat that eases intercomparison of studies.

In Chapter 4, I searched for explanatory variables within a digital soil mapping approach that enables reconstructions of the pattern of peat initiation and lateral expansion within (and potentially beyond) peat remnants, with quantified uncertainty. Basal radiocarbon dates were obtained from the Fochteloërveen peat remnant, which formed the basis for subsequent analyses. I investigated the relationship between peat initiation age and three potential covariates: (1) total thickness of organic deposits, (2) elevation of the Pleistocene mineral surface that underlies the organic deposits, and (3) a constructed variable representing groundwater-fed wetness based on elevation of the mineral surface and current hydraulic head. Significant relationships were found with covariate (1) and (3), which were hence used for subsequent modelling. Results indicate simultaneous peat initiation at several loci in the Fochteloërveen during the Early Holocene, and continuous lateral expansion until 900 cal y BP. Lateral expansion accelerated between 5,500 - 3,500 cal y BP. The presented approach is spatially explicit (i.e., results in a map of peat initiation ages), and allows for a quantitative evaluation of the prediction using the standard deviation and comparison of predictions with validation points. The applied method based on covariate (1) is only useful where remnant peat survived, whereas covariate (3) may ultimately be applied to reconstruct peat initiation ages and lateral peatland expansion beyond the limits of peat remnants.

The methodological tools presented in this thesis may help to advance science on peatland reconstructions. Depending on research objectives, they can be applied separately, or concurrently in a complementary manner.

Peat initiation results from terrestrialisation, paludification and/or primary mire formation. Based on the findings of this thesis, I conclude that paludification was the most prominent process of peat formation in the northern Dutch coversand landscape (and Fochteloërveen), with local occurrence of terrestrialisation sites. The work presented in this thesis points to several environmental factors that may have led to paludification, consequent peat initiation and subsequent lateral expansion. I present a hypothesis on the steering factors for paludification and peat growth in the Northwest European mainland, that offers starting points for further research.

Increased understanding of the age and palaeogeography of peatlands is highly relevant for research on ecosystem services provided by peatlands (such as carbon dynamics), for nature conservation and restoration, and for cultural history and heritage management. Carbon dynamics form a key ecosystem service provided by peatlands. The age and spatial extent of peat deposits are crucial parameters for modelling past and future carbon storage. In addition, knowledge on the timing of peat initiation is of great significance in assessing archive content and value, and may provide long-term palaeoenvironmental and palaeoecological references for nature conservation and restoration goals. Information on the palaeogeographical development of peatlands plays a vital role in contextualising long-term human habitation patterns, and in the analysis of single archaeological sites. In addition, insights in the age and character of peat deposits may be highly informative for cultural heritage management purposes and archaeological prospection, as peat layers may cover or contain well-preserved waterlogged archaeological sites that are unparalleled in dryland environments. The knowledge on the palaeogeographical and palaeoenvironmental development of peatlands in the Northwest European mainland that was gained in this thesis is directly incorporated in interdisciplinary studies within the overarching *Home Turf* project. Peatlands are landscapes with a strong potential to feature in environmental education, both in formal educational contexts and in science communication with the general public. Science communication efforts on peatland research in the Northwest European mainland and elsewhere may help to show the value, history and beauty of these unique landscapes, and the importance of protecting and restoring peat remnants.

# **PE&RC Training and Education Statement**

With the training and education activities listed below the PhD candidate has complied with the requirements set by the C.T. de Wit Graduate School for Production Ecology and Resource Conservation (PE&RC) which comprises of a minimum total of 32 ECTS (= 22 weeks of activities).



## Review/project proposal (8.5 ECTS)

• Reconstructing Holocene spatio-temporal bog developments

## Post-graduate courses (5.4 ECTS)

- Pre-conference workshop: Spatial sampling for mapping; Pedometrics Conference, Wageningen (2017)
- Bayesian statistics; Wageningen Graduate Schools (2017)
- Radiocarbon dating and Bayesian chronological analysis; Oxford Department for Continuing Education (2018)
- Groundwater processes; VU Amsterdam (2018)
- Resilience of living systems From fundamental concepts to interdisciplinary applications; Graduate Schools PE&RC and WIAS (2018)

## Deficiency, refresh, brush-up courses (3.4 ECTS)

- Spatial modelling and statistics; WUR (2017)
- The fourth dimension; WUR (2018)

## Laboratory training and working visits (1 ECTS)

 Peatland workshop; Hyytiälä Forestry Field Station, University of Helsinki, Finland (2018)

## Competence strengthening/skills courses (4.71 ECTS)

- Effective behaviour in your professional surroundings; Wageningen Graduate Schools (2017)
- Orientation on teaching for PhD candidates; Wageningen Graduate Schools (2017)
- IT Workshop online samenwerken; WUR (2018)
- Scientific artwork Vector graphics and images; Wageningen Graduate Schools (2020)
- Career orientation; Wageningen Graduate Schools (2020)

• Visual communication; WUR (2021)

### Scientific integrity/ethics in science activities (0.6 ECTS)

• Research integrity; Wageningen Graduate Schools (2017)

### PE&RC Annual meetings, seminars and the PE&RC weekend (1.1 ECTS)

- PE&RC First years weekend (2017)
- PE&RC Last year afternoon; online (2020)

#### Discussion groups/local seminars or scientific meetings (5.23 ECTS)

- Field visit to a peatland site in the Binnenveld area with RCE (2019)
- Excursion to Schokland with Home Turf and Wetfutures teams and guest researchers (2019)
- Excursion to Bargerveen and Drents Museum with Home Turf team and guest researchers (2018)
- Peatland exchanges (2022)
- Other activities related to peatland science (2017-2022)
- Young WUR seminars (2017-2022)
- Land Dynamics Discussion Group (2017-2022)
- R Users meeting (2017-2022)
- Other activities related to the wider scientific field (2017-2022)
- Member hiring committee for a postdoc position at WUR and for a PhD position at WUR (2018, 2020)
- Participation as PhD-representative in the Soil Science Cluster review (2021)

#### International symposia, workshops and conferences (5.8 ECTS)

- Symposium of the Netherlands Centre for Luminescence dating; oral presentation; Utrecht, the Netherlands (2017)
- 23rd Annual meeting of the European Association of Archaeologists; oral presentation; Maastricht, the Netherlands (2017)
- Symposium RCE/Deltares/TNO; oral presentation; Amersfoort, the Netherlands (2020)
- Reuvens-digi-dag door Stichting Reuvens; oral presentation; online (2020)

#### Societally relevant exposure (3.44)

- Creating a website for our project: www.boglandscapes.eu (2017)
- Co-writing an article for Nature Today (2018)

- NBV Themadag: Bodem & Archeologie (presentatie voor breder publiek) (2019)
- Filmed interview about my research for a documentary on Fochteloërveen ('Et Fochtelervene - Van lestied Naor Wieshied') (2020)
- Filmed interview about my research for the tv-series 'ROEG!' (2021)
- Oral presentation during Session 1 of the seminar series 'Peatland Exchanges' (about science and art) (2022)

## Committee work (0.75 ECTS)

 Member of the General Board of the Dutch Soil Society (Nederlandse Bodemkundige Vereniging, NBV) (2017-2022)

### Lecturing/supervision of practicals/tutorials (11.83 ECTS)

- Integration course soil, water and atmosphere (2017)
- Environmental data collection and analysis (2017)
- Soils and landscapes of the Netherlands (2017, 2018, 2019)
- Introduction soil, water, atmosphere (2017, 2018)
- Landscape geography (2017, 2018, 2019)
- Introduction to Soil Geography (2018, 2019)
- Coastal oceanography and delta geology (2019, 2020, 2021)
- Evaluator at BSc symposium (2020)

#### BSc/MSc thesis supervision (5.61 ECTS)

- Best practices for creating high-resolution 3D pre-peat landscapes
- Sacrificing with wet feet? An environmental reconstruction of the Bolleveen area (Drenthe, the Netherlands) in the Roman period (12 BC – AD 400)
- The spatial and temporal distribution of bog bodies in Northwestern Europe
- Constraining peat initiation by OSL-dating the pre-peat landscape
- Problems and solutions of conventional peat meadow farmers in the Netherlands (internship at Stichting Boerengroep)

# List of publications

#### Manuscripts in preparation

 Van Beek R, Quik C, Van der Linden M. Drowning landscapes revisited. Correlating peatland expansion, human habitation trends and vegetation dynamics in the Northwest European mainland. In preparation.

#### **Peer-reviewed** articles

- Quik C, Van der Velde Y, Candel JHJ, Steinbuch L, Van Beek R, Wallinga J, 2023. Faded landscape: unravelling peat initiation and lateral expansion at one of northwest Europe's largest bog remnants. *Biogesciences* 20, 695–718, https://doi.org/10.5194/bg-20-695-2023.
- Van Beek R, Quik C, Bergerbrandt S, Huisman F, Kama P, 2023. Bogs, bones and bodies. The deposition of human remains in European mires (9000 BC – AD 1900). Antiquity 1-21, https://doi.org/10.15184/aqy.2022.163.
- Quik C, Palstra SWL, Van Beek R, Van der Velde Y, Candel JHJ, Van der Linden M, Kubiak-Martens L, Swindles GT, Makaske B, Wallinga J, 2022. Dating basal peat: The geochronology of peat initiation revisited. *Quaternary Geochronology* 72, 1-22, https://doi.org/10.1016/j.quageo.2022.101278.
- Quik C, Harkema T, Van der Plicht J, Quik J, Van der Velde Y, Van Beek R, Wallinga J, 2021. Using legacy data to reconstruct the past? Rescue, rigour and reuse in peatland geochronology. *Earth Surface Processes and Landforms* 46, 2607–2631, https://doi.org/10.1002/esp.5196.
- Quik C, Candel JHJ, Makaske B, Van Beek R, Paulissen M, Maas GJ, Verplak M, Spek T, Wallinga J, 2020. Anthropogenic drivers for exceptionally large meander formation during the Late Holocene. *Anthropocene* 32, 1–15, https://doi.org/10.1016/j.ancene.2020.100263.
- Van Beek R, Candel JHJ, Quik C, Bos JAA, Gouw-Bouman MTIJ, Makaske B, Maas GJ, 2019. The landscape setting of bog bodies: Interdisciplinary research into the site location of Yde Girl, The Netherlands. *The Holocene* 29(7), 1206–1222, https://doi.org/10.1177/0959683619838048.
- Van Dobben HF, Quik C, Wamelink GWW, Lantinga EA, 2019. Vegetation composition of *Lolium perenne*-dominated grasslands under organic and conventional farming. *Basic and Applied Ecology* 36, 45-53, https://doi.org/10.1016/j.baae.2019.03.002.
- Candel JHJ, Kleinhans MG, Makaske B, Hoek WZ, Quik C, Wallinga J, 2018. Late Holocene channel pattern change from laterally stable to meandering – a palaeohydrological reconstruction. *Earth Surface Dynamics* 6, 723-741, https://doi.org/10.5194/esurf-6-723-2018.

 Quik C, Wallinga J, 2018. Reconstructing lateral migration rates in meandering systems - a novel Bayesian approach combining optically stimulated luminescence (OSL) dating and historical maps. *Earth Surface Dynamics* 6, 705-721, https://doi.org/10.5194/esurf-6-705-2018.

#### Datasets

- Quik C, Van der Velde Y, Candel JHJ, Steinbuch L, Van Beek R, Wallinga J, 2023. Data from: Faded landscape: unravelling peat initiation and lateral expansion at one of northwest Europe's largest bog remnants. *4TU.Centre for Research Data*: https://doi.org/10.4121/20237958.
- Quik C, Steinbuch L, 2022. Workshop FAIR Data and Data Reuse for Environmental Science Group Researchers. 4TU.Centre for Research Data: https://doi.org/10.4121/21399975.v1.
- Quik C, Palstra SWL, Van Beek R, Van der Velde Y, Candel JHJ, Van der Linden M, Kubiak-Martens L, Swindles G, Makaske B, Koudijs R, Wallinga J. 2022. Data from: Dating basal peat: The geochronology of peat initiation revisited. *4TU.Centre for Research Data*: https://doi.org/10.4121/16923358.
- Quik C, Van der Velde Y, Harkema T, Van der Plicht H, Quik J, Van Beek R, Wallinga J, 2021. Data from: Using legacy data to reconstruct the past? Rescue, rigor and reuse in peatland geochronology. *4TU.Centre for Research Data*: https://doi.org/10.4121/14406347.
- Quik C, Candel JHJ, Makaske B, Van Beek R, Paulissen M, Maas GJ, Verplak M, Spek T, Wallinga J, 2020. Data from: Anthropogenic drivers for the rapid formation of extremely large meanders during the Late Holocene. *4TU.Centre for Research Data*: https://doi.org/10.4121/uuid:c9c892de-4f3f-4c1b-b684-17c700b02f31.
- Van Leeuwen MMWJ, Schols AM, Quik C. 2019. Data from: Quantitative visual soil observation for visual soil evaluation on dairy farms. *4TU.Centre for Research Data*: https://doi.org/10.4121/uuid:237f0826-d04b-43e0-82d3-d2f4a6bf4ce4.
- Quik C, Wallinga J, 2018. Data from: Reconstructing lateral migration rates in meandering systems - a novel Bayesian approach combining OSL dating and historical maps. *4TU.Centre for Research Data*: https://doi.org/10.4121/uuid:1ca25393-aa99-48dc-b382-0506322bc449.

# Acknowledgements

### Dankwoord

Peatlands are sometimes perceived as swamps with murky skies and low-hanging fog, where the way is difficult to find. Doing a PhD sometimes feels a bit like trying to find your way in a swamp. As in a peatland, the route may sometimes be unclear or appear troublesome, until you see the beautiful *Sphagnum* mosses, the dancing ringlets, and the foraging cranes. I would like to thank all who contributed in any way to this thesis, for helping me find my way through the peatland. I would like to express my gratitude to the following people in particular.

Venen worden wel eens gezien als moerassen met grauwe luchten en laaghangende mist, waar de weg moeilijk te vinden is. Het doen van een PhD voelt af en toe ook een beetje als het zoeken van je weg in een moeras. Net als in een veengebied is de route soms onduidelijk of moeilijk, tot je de prachtige veenmossen ziet, de dansende veenhooibeestjes, en de foeragerende kraanvogels. Ik wil graag allen bedanken die op welke wijze dan ook een bijdrage hebben geleverd aan dit proefschrift, voor de hulp bij het vinden van mijn weg door het veen. In het bijzonder gaat mijn dank uit naar de volgende personen.

Many thanks to my promoter Jakob and co-promoters Roy and Ype for their trust, kind guidance and constructive feedback over the years. Jakob, thank you for your shrewdness, your skill in alternating between detail and overview, and sympathetic approach. Roy, thank you for sharing your passion for archaeology, for your ability to maneuver with poise through interdisciplinary research, and strong eye for detail. Ype, thank you for your down-to-earth mentality, patience and skill in explaining hydrology, and enthusiasm to brainstorm about everything that has to do with peat.

Veel dank aan mijn promotor Jakob Wallinga en co-promotoren Roy van Beek en Ype van der Velde voor hun vertrouwen, vriendelijke begeleiding en constructieve feedback tijdens de afgelopen jaren. Jakob, dank voor je scherpe blik, je vaardigheid om te schakelen tussen detail en overzicht, en sympathieke benadering. Roy, dank voor het delen van je passie voor archeologie, voor je vaardigheid om op verfijnde wijze te manouvreren door interdisciplinair onderzoek, en sterke oog voor detail. Ype, dank voor je down-to-earth mentaliteit, geduld en vaardigheid in het uitleggen van hydrologie, en enthousiasme om te brainstormen over alles dat met veen te maken heeft. Many thanks to my paranimfs Luc Steinbuch and Jasper Candel. Luc, thank you for all your help with statistics and code, your efforts for the thesis lay-out, and the pleasant cooperation during the project for the Data Science & Artificial Intelligence Fellowship Program 2022. Jasper, thank you for all the fruitful discussions, the nice joint field days, and for your great help to find the red thread again during the spring of 2019, when the peatland did seem rather hazy.

Veel dank aan mijn paranimfen Luc Steinbuch en Jasper Candel. Luc, bedankt voor al je hulp met statistiek en code, je inzet voor de thesis lay-out, en de goede samenwerking tijdens het project voor het Data Science & Artificial Intelligence Fellowship Program 2022. Jasper, dank voor al het sparren, de leuke gezamenlijke velddagen, en je goede hulp om de rode draad weer terug te vinden tijdens het voorjaar van 2019, toen het veen toch wel mistig leek.

In addition to the people mentioned above, I would like to express my gratitude and appreciation to all other co-authors that contributed to this PhD thesis: Tom Harkema, Hans van der Plicht, Jim Quik, Sanne Palstra, Marjolein van der Linden, Lucy Kubiak-Martens, Graeme Swindles, and Bart Makaske. Thank you for your efforts and analyses, which were indispensable for my research.

Naast de bovengenoemde personen, wil ik graag mijn dankbaarheid en waardering uitspreken aan alle andere co-auteurs die hebben bijgedragen aan dit proefschrift: Tom Harkema, Hans van der Plicht, Jim Quik, Sanne Palstra, Marjolein van der Linden, Lucy Kubiak-Martens, Graeme Swindles, and Bart Makaske. Dank voor jullie inzet en analyses, die onmisbaar waren voor mijn onderzoek.

Thanks to Natuurmonumenten Fochteloërveen and Staatsbosbeheer Kop van Drenthe for the opportunity to do field research in the Fochteloërveen area. Tom Harkema, Marte Stoorvogel, Teun Fiers and Klais Blaauw, thanks for your work in the field, without your help the fieldwork would not have been possible! Mieke, thank you for all the administrative support, especially during the field work period. Romy Koudijs, thanks for your sharing your MSc thesis data on organic matter contents of a selection of cores. Thanks to the WUR Library, the Wageningen Data Competence Center and 4TU.Centre for Research Data for the support in publishing the research data.

Dank aan Natuurmonumenten Fochteloërveen en Staatsbosbeheer Kop van Drenthe voor de geboden mogelijkheden om onderzoek te doen in het Fochteloërveen. Tom Harkema, Marte Stoorvogel, Teun Fiers en Klais Blaauw, dank voor jullie werk in het veld, zonder jullie hulp was het veldwerk niet mogelijk geweest! Mieke, bedankt voor alle administratieve ondersteuning, in het bijzonder tijdens de veldwerkperiode. Romy Koudijs, bedankt voor het delen van je MSc thesis data over organische stofgehalten van een selectie van kernen. Dank aan de WUR Library, het Wageningen Data Competence Center en 4TU.Centre for Research Data voor de ondersteuning bij het publiceren van de onderzoeksdata.

Finally, I would like to thank my family and my partner. Many thanks for your support and belief in me.

Als laatste wil ik graag mijn familie en partner bedanken. Veel dank voor jullie steun en geloof in mij.

# Acknowledgements of financial support

This research is part of the research programme *Home Turf.* An integrated approach to the long-term development, cultural connections and heritage management of Dutch raised bogs, funded by the Netherlands Organization for Scientific Research (NWO) under grant no. 276-60-003 (granted to Dr Roy van Beek).

Financial support from Wageningen University for printing this thesis is gratefully acknowledged.

Cover design by: Cindy Quik Lay-out by: Luc Steinbuch & Cindy Quik Printed by: ProefschriftMaken || www.proefschriftmaken.nl

Printed on FSC-certified 100% recycled paper, with a plastic-free cover.

